Western University Scholarship@Western

Electronic Thesis and Dissertation Repository

8-7-2024 2:00 PM

The Continuous Learning Needs of Personal Support Workers Who Care For People Living with Dementia in Long-Term Care

Grace I. Norris, Western University

Supervisor: Savundranayagam, Marie Y., *The University of Western Ontario* A thesis submitted in partial fulfillment of the requirements for the Master of Science degree in Health and Rehabilitation Sciences © Grace I. Norris 2024

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Medicine and Health Sciences Commons

Recommended Citation

Norris, Grace I., "The Continuous Learning Needs of Personal Support Workers Who Care For People Living with Dementia in Long-Term Care" (2024). *Electronic Thesis and Dissertation Repository*. 10251. https://ir.lib.uwo.ca/etd/10251

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

Abstract

Personal support workers (PSWs) comprise over half of the long-term care (LTC) workforce who care for people living with dementia yet are considered less skilled compared to other healthcare professionals. Improvements to dementia education from the perspectives of PSWs are under-explored. To address this gap, this study investigated PSW perspectives of their dementia-specific learning needs in LTC. Guided by Interpretive Description (Thorne, 2016) and adult learning theory (Knowles, 1990), four major learning needs were discerned through a secondary qualitative analysis: understanding dementia, addressing responsive behaviours, person-centered communication and attitudes, and delirium. Learning needs are best met in supportive environments with experiential methods that involve peer learning, feedback, and evaluation. Successful learning is mediated through an openness to learning and a good teamwork culture. The findings underscore the importance of ongoing dementia education tailored to the needs of PSWs, with implications for future training programs aimed at improving dementia care.

Keywords: Dementia, personal support workers, formal caregivers, learning need assessment, education, training, long-term care, qualitative secondary analysis.

Summary for Lay Audience

When it comes to providing care to people living with dementia in long-term care homes, personal support workers (PSWs) provide the most out of any healthcare professional. Despite this, PSWs are considered the least skilled due to the lack of comprehensiveness in their formal education and few opportunities for continuous education throughout their career. As a result, the quality of dementia care is compromised. Improvements to dementia education from the perspectives and needs of PSWs have been under-explored. To address this knowledge gap, this study investigated PSW perspectives of their dementia-specific learning needs while working in long-term care homes. Interpretive Description (Thorne, 2016) and adult learning theory (Knowles, 1990) were used to guide a secondary qualitative analysis of 22 one-hour focus groups with 39 PSWs working in long-term care settings across London, Ontario. Three major findings were discerned from the data: 1) gaps in dementia-specific competencies (i.e., the learning needs), 2) how the gaps should be addressed (i.e., learning methods), and 3) contextual factors that can influence the application of knowledge into practice (i.e., mediators). The gaps in dementia-specific competencies were attributed to limited preparation during formal PSW education and a lack of continuous training opportunities. PSWs wanted additional education concerning dementia, how to address responsive behaviours, how to use person-centered communication and attitudes, and understanding and recognize delirium among persons living with dementia. These topics are best met within a supportive peer environment with opportunities to learn from coworkers through feedback, groups discussions, observations, and experiential learning activities. For education to translate into practice, PSWs needed to be open to learning from both educational programs and their coworkers. The findings underscore the importance of ongoing dementia education tailored to the needs of PSWs, with implications for future training programs aimed at improving dementia care.

Co-Authorship Statement

This thesis is composed of original work conducted and written by Grace Irene Norris in fulfillment of a Master of Science under the supervision of Dr. Marie Y. Savundranayagam. The thesis includes an introductory chapter (chapter one), an integrated manuscript (chapter two), a secondary qualitative data analysis (chapters three and four), and a concluding chapter (chapter five). Chapter two was published in the Educational Gerontology, a peer-reviewed journal, on May 21st, 2024 (citation below). A manuscript for the remaining chapters of this thesis is in progress for publication. The data analyzed in chapter four was collected by Dr. Savundranayagam and members of the Sam Katz Community Health and Aging Research Unit at the University of Western Ontario.

Dr. Savundranayagam provided invaluable guidance in the design, analysis, interpretation, and written conceptualization of all chapters within this thesis. Additionally, the advisory committee, Dr. Ashfin Vafaei and Dr. Gail Teachman, provided significant intellectual guidance and input regarding the design, interpretations, and written conceptualizations of all chapters within this thesis. In addition, all mentioned collaborators provided revisions and feedback on all chapters.

Citation (chapter two):

Norris, G., Savundranayagam, M.Y., Vafaei, A., & Teachman, G. (2024). The representation of personal support workers in dementia-specific learning need assessments: a scoping review. *Educational Gerontology*. https://doi.org/10.1080/03601277.2024.2356925

Acknowledgments

I am extremely thankful for the support I received throughout the past two years while completing this thesis. My deepest gratitude goes to my supervisor, Dr. Marie Y. Savundranayagam. Your unwavering support, encouragement, and dedicated mentorship have been invaluable. Thank you for being a role model for what a successful woman in academia looks like.

I would also like to thank the members of my advisory committee, Dr. Ashfin Vafaei and Dr. Gail Teachman for their assistance and guidance throughout my research process. Your feedback, support, and expertise have been critical in building my confidence and success as a graduate student.

To my family and friends, I truly could not have completed this thesis without your unconditional support and love. To all of my friends and family, thank you for providing me with encouragement, reassurance, and guidance through many phone calls, text messages, emails, and face-to-face conversations. I would also like to extend my gratitude to my colleagues and friends in the Care Lab, whose daily support, companionship, and many laughs kept me moving forward.

Researching about care provision is one thing, but being a frontline caregiver providing that care is an entirely different experience. I would like to thank my colleagues from Community Living in Picton, Ontario, who I had the privilege to work closely with during the COVID-19 pandemic. Your dedication, passion, and love for caring for older adults living with disabilities and chronic illnesses within an imperfect system inspired me to pursue this degree. I am deeply grateful for your assistance and patience as you consistently guided me through the physical, emotional, and mental complexities of caregiving. Thank you for igniting a flame within me to advocate for equitable improvements in educational support and care provision for older adults and frontline caregivers.

Table of Contents

Abstrac	ct	ü
Summa	ary for	Lay Audienceiii
Co-Au	thorshi	ip Statement iv
Acknow	wledgr	nentsv
Table of	of Con	tentsvi
List of	Tables	six
List of	Figure	28X
List of	Apper	ndices xi
1 Intro	duction	n1
1.1 H	Backgr	ound1
1	1.1.1	Dementia1
1	1.1.2	Long-Term Care
1	1.1.3	PSW In Practice
1	1.1.4	Ontario PSW Education
1	1.1.5	Dementia-Specific Continuing Education Programs5
1.2 A	Adult I	Learning Theory7
1.3 \$	Signifi	cance and Outline of the Research9
2 The H Nee	Repres d Asse	entation of Personal Support Workers in Dementia-Specific Learning essments: A Scoping Review11
2.1 H	Backgr	round11
2.2 N	Method	ds14
2	2.2.1	Inclusion and Exclusion Criteria14
2	2.2.2	Evidence Selection

	2.2.3	Data Extraction and Charting	16
2.3	Result	s	17
	2.3.1	Sources of Evidence Selection	17
	2.3.2	Characteristics of Sources of Evidence	18
	2.3.3	PSW Representation in Sample Population	27
	2.3.4	Data Collection Methods	28
	2.3.5	Dementia-Specific Learning Needs Assessed in Included Studies	29
2.4	Discus	sion	31
	2.4.1	Strengths and Limitations	34
	2.4.2	Implications for Future Research	35
2.5	Conclu	usion	35
3 Met	thod		37
3.1	Paradi	gm	37
3.2	Metho	dology: Interpretive Description	37
3.3	Resear	ch Methods	39
	3.3.1	Ethics	39
	3.3.2	Sampling and Recruitment	40
	3.3.3	Be EPIC	40
	3.3.4	Data Collection	40
	3.3.5	Data Analysis and Interpretation	42
3.4	Rigor.		44
3.5	Positic	onality of the Researcher	45
4 Res	ults		47
4.1	Partici	pants and Demographic Data	47
4.2	Demar	nd for Dementia-Specific Education	50

	4.2.1	Formal and Continuing Education	50
	4.2.2	Learning Needs	53
	4.2.3	Evolution of Learning Needs	63
4.3	Learni	ng Methods	64
	4.3.1	Experiential Learning	68
4.4	Media	tors	70
5 Dis	cussion		73
5.1	Demai	nd for Dementia-Specific Education	73
	5.1.1	Understanding Dementia	77
	5.1.2	Person-Centered Communication and Addressing Responsive Behaviours	80
	5.1.3	Delirium	84
5.2	Learni	ng Methods	85
5.3	Media	tors	87
5.4	Streng	ths and Limitations	88
5.5	Implic	ations and Future Directions	89
5.6	Conclu	usion	92
Refe	ences		94
Appe	ndices		111
Curri	culum '	Vitae	115

List of Tables

Table 1: Search strategy keywords and terms	15
Table 2: Summary of included articles	19
Table 3: Methods of data collection	29
Table 4: Study objectives	30
Table 5: Focus group sessions	41
Table 6: Participant demographics	48

List of Figures

Figure 1: PRISMA diagram	18
Figure 2: Identification of learning needs before, during, and after participating in Be EPIC	64
Figure 3: Identification of learning methods before, during, and after participating in B	e 65

List of Appendices

Appendix A: Ethics Approval	104
Appendix B: Semi-Structure Focus Group Questions	105

Chapter 1

1 Introduction

1.1 Background

1.1.1 Dementia

A dementia diagnosis occurs every three seconds (Alzheimer's Disease International, n.d.; World Health Organization, 2023). In 2020, over 500,000 Canadians were living with dementia (Alzheimer Society of Canada, 2022). Within ten years, this number will reach one million (Alzheimer Society of Canada, 2022). Dementia is classified as a chronic neurodegenerative syndrome causing significant deterioration in cognitive functioning (Alzheimer's Disease International, n.d.; World Health Organization, 2017). The deterioration commonly targets areas of the brain responsible for memory, executive functioning, comprehension, verbal communication, navigational abilities, judgement, and motor skills (World Health Organization, 2023). As a result, the ability to verbally communicate about physical, social, spiritual, occupational, and emotional needs can become increasingly difficult (American Psychiatric Association, 2022; Eggenberger et al., 2013; Kitwood, 1997). A person living with dementia may communicate their unmet needs through physical behaviours (e.g., resisting care, pushing, kicking) and/or vocalizations (e.g., yelling, crying, repeated questioning), referred to as responsive behaviours (Clifford & Doody, 2018; Whall & Kolanowski, 2004). It can be equally distressing to both persons living with dementia and their caregivers when the latter is unable to understand, communicate, and provide care in ways that meet these needs, and can result in physical harm (Holst & Skär, 2017; Whall & Kolanowski, 2004). Due to the severity of symptom progression, dementia is considered a major cause of disability and dependency among older adults (World Health Organization, 2017), requiring specialized and high quality care provision.

Person-centered care constitutes the philosophical foundation for optimal dementiaspecific services and support standards (Alzheimer Society of Canada, 2014; Breen et al., 2022). Person-centered care involves recognizing a person living with dementia as a unique individual with feelings, experiences, values, and preferences, which are incorporated and considered into care practices (Kitwood, 1997; Savundranayagam et al., 2016). It promotes a high quality of care by equipping caregivers with the knowledge, skills, and attitudes (collectively referred to as 'competencies') to provide care that focuses on understanding and responding to the unique needs, preferences, and experiences of each person (Kitwood, 1997). Person-centered care principles promote the development of meaningful relationships with persons with dementia, personalized care provision, prioritization of emotional and psychological well-being, creating supportive environments, and empathetically understanding the experiences and feelings of a person living with dementia (Breen et al., 2022; Kary et al., 2020; Kitwood, 1997). For instance, Cohen and colleagues (2022) found that caregivers with person-centered attitudes were able to positively and effectively interact and communicate with persons with dementia, interpret meanings behind responsive behaviours, and address needs more easily than staff without person-centered attitudes. The provision of person-centered care is significantly important to persons living with dementia, as they are often viewed through a perspective of loss (e.g., loss of memory and cognitive abilities) or behavioural symptoms (e.g., aggressive, difficult to care for, unable to socially engage), placing them at risk for depersonalized care and no longer being viewed as a person (Holst & Skär, 2017; Kitwood, 1997). Therefore, ensuring caregivers are equipped adequately with the competencies to provide person-centered care, is essential for high quality dementia care.

1.1.2 Long-Term Care

Many individuals living with dementia rely on long-term care services (Sethna, 2013). As of 2023, there are 626 long-term care homes in Ontario with approximately 63% of residents living with dementia (Office of the Auditor General of Ontario, 2023). The purpose of long-term care homes is to provide 24/7 essential care to individuals who are not able to access care in the community or care for themselves (Office of the Auditor General of Ontario, 2023). Long-term care services include nursing (e.g., medication administration) and personal care (e.g., assistance with activities of daily living), therapeutic services (e.g., physiotherapy), dietary services, and recreational programs

2

(Office of the Auditor General of Ontario, 2023). The fundamental principle of long-term care homes is defined in Section One of the *Fixing Long-Term Care Homes Act, 2021*, as a place where residents "may live with dignity and in security, safety and comfort and have their physical, psychological, social, spiritual and cultural needs adequately met" (Government of Ontario, 2023), underscoring a person-centered model of care.

Yet, long-term care homes have consistently reported inadequate working conditions which fail to meet the needs of persons living with dementia and foster a culture of dignity and respect among both residents and staff (Baines & Armstrong, 2018; Office of the Auditor General of Ontario, 2023; Ostaszkiewicz et al., 2020). The prioritization of profits over the quality of care has resulted in under-valuing and under-supporting longterm care staff and perpetuating a "task-oriented" culture of care rather than a "personcentered" one (Lightman, 2022; Savundranayagam et al., 2021). PSWs represent over half of the long-term care workforce and provide 80% of daily direct care to persons living with dementia (Chamberlain et al., 2019; Ministry of Long-Term Care, 2020; Public Health Agency of Canada, 2021). However, PSWs consistently report feeling unprepared, under-supported, under-pressure, and lacking confidence which contributes to work-related stress and burnout, low occupational satisfaction, and high turnover rates (Boamah et al., 2023; Kane et al., 2023; Li et al., 2021; Lightman, 2022; Rivett et al., 2019; Savundranayagam et al., 2021; Scales, 2022). The quality of care provided to residents is also compromised as care provision becomes rushed, depersonalized, and limits autonomy, dignity, and respect (Holst & Skär, 2017; Mialkowski, 2020; Rasmussen et al., 2023; Savundranayagam et al., 2016; Scott et al., 2011; Swinton et al., 2023). Persons living with dementia in long-term care homes rely on and trust PSWs to provide safe, effective, timely, and person-centered care, underscoring the importance of ensuring PSWs feel prepared, supported, and confident in their abilities to quality dementia care.

National strategies have recognized the importance of improving and building the capacity of long-term care organizations and staff to provide optimal dementia care (Public Health Agency of Canada [PHAC], 2021). This includes initiatives to improve

3

the quality of dementia care provided by PSWs, such as Bill 121 *Improving Dementia Care in Ontario Act, 2023*, which, upon approval, would require the Ontario Ministry of Colleges and Universities to review the PSW Training Standard (2022) to include more in-depth learning about person-centered dementia care (Kusendova-Bashta & Smith, 2023) (see below for further information regarding the PSW Training Standard). Placing a greater emphasis on ensuring PSWs receive adequate education in dementia care can significantly improve the quality of care provided to persons living with dementia (PHAC, 2021; Ministry of Long-Term Care, 2020).

1.1.3 PSW In Practice

The goal of PSW practice is to provide quality daily personal care and support to vulnerable individuals (Grant, 2016). PSWs provide a wide range of care activities based on the needs, desires, and well-being of the individuals they support. These include both basic (e.g., eating, toileting, etc.) and instrumental (e.g., shopping, cooking, etc.) activities of daily living, clinical care services (e.g., measuring temperature and blood pressure), and carrying out controlled acts by the Regulated Health Professions Act (e.g., administering medication) under the supervision of a regulated healthcare professional (e.g., nurse) (Government of Canada, 2021; Grant, 2016). In addition, PSWs provide mental and emotional assistance by encouraging independence, dignity, and comfort (Grant, 2016). Unlike other healthcare professionals, PSWs are not regulated or licensed by government bodies or professional associations. This means that PSWs do not have a governing body, a certification process, and standardized code of conduct (Rossiter & Godderis, 2020; Sethna, 2013).

1.1.4 Ontario PSW Education

An education standard for Ontario PSWs was first developed and implemented within Ontario colleges in 2014 (Grant, 2016; Kelly & Bourgeault, 2015; Ministry of Training, Colleges and Universities, 2014). Formal PSW education is offered through three distinct types of institutions: district school board continuing education programs, community colleges, and private career colleges (Grant, 2016; Ministry of Training, Colleges and Universities, 2022). While each program in Ontario is required to comply with the Personal Support Worker Training Standard by the Ministry of Training, Colleges and Universities (2022), separate government branches regulate each institution (Grant, 2016). Thus, variability exists between program costs, duration, hours of clinical experience, delivery methods, degree of theoretical competencies included, and evaluation process (Brooks et al., 2008; Grant, 2016; Kelly & Bourgeault, 2015; Sethna, 2013). The training standard includes 14 'vocational standards', 'essential employability skills', and general education requirements which students must reliably demonstrate to receive a PSW certificate and begin caring for persons living with dementia (Kelly & Bourgeault, 2015; Ministry of Training, Colleges and Universities, 2014, 2022). Out of the 14 vocational learning outcomes, one requires students to "provide client-centered and client-directed care to individuals experiencing various mental health illnesses and challenges, cognitive and intellectual impairments, and/or responsive behaviours by using supportive approaches and evidence-based practices to promote positive and safe behaviours in clients" (Ministry of Training, Colleges and Universities, 2022, p. 40). However, PSW education programs have been criticized for the content largely focusing on vocational and task-based activities with many lacking thorough dementia-related content and person-centered language (Foster et al., 2019; Grant, 2016). PSWs have reported a need for education to include more dementia-specific and person-centered content to adequately prepare them to provide optimal care within long-term care homes (Rasmussen et al., 2023). The 2014 standard was updated in 2022 to include more person-centered language and as well as three additional competencies related to dementia, namely identifying changes in an individual's behaviour by understanding signs and symptoms of cognitive impairments (Ministry of Colleges and Universities, 2022). However, it is not well understood how these changes have addressed gaps in dementia-specific competencies. Furthermore, when PSWs enter the workforce, they rely on continuing education programs to address gaps in knowledge, skills, and attitudes left by formal education.

1.1.5 Dementia-Specific Continuing Education Programs

There are several dementia-specific continuing education programs (also referred to as professional development opportunities or training programs) available across Ontario,

including Gentle Persuasive Approaches (GPA), U-FIRST, DementiAbility, and various education modules provided by the Alzheimer's Society of Canada. These programs are delivered predominantly online, with some in-person opportunities, and designed based on adult learning theory (Knowles, 1990), but vary in terms of target audience (e.g., healthcare professionals, informal care partners), cost (\$0 - 450/participant), and learning objectives (e.g., addressing responsive behaviours, self-care, understanding dementia) (Geriatric Certificate Program, n.d.-a, -b, -c). GPA was developed to teach healthcare professionals how to address, prevent, and understand responsive behaviours from a person-centered perspective, and has since become the most widely distributed dementiaspecific training program in Ontario. (Advanced Gerontological Education, n.d.; Geriatric Certificate Program, n.d.-b). As of 2022, GPA is recommended in the PSW Training Standard for best practices in addressing responsive behaviours (Ministry of Training, Colleges and Universities, 2022). U-First focuses on building healthcare professionals' self-efficacy in addressing responsive behaviours and encouraging quality interactions with persons living with dementia (Geriatric Certificate Program, n.d.-c; McAiney & Service, 2005). DementiAbility builds the capacity of healthcare professionals to limit excess disability, promote independence, and enhance meaningful engagement in occupational interests among persons living with dementia (DementiAbility, n.d.). Finally, the Alzheimer's Society of Canada offers over thirteen online training modules for healthcare professionals, in addition to webinars and workshops, most of which are offered for free online and on-demand (Alzheimer Society of Ontario, n.d.). These programs cover a large range of topics including foundational information on dementia (i.e., signs and symptoms, treatments, progression, risks, etc.), palliative care, self-care, person-centered care, oral care, and transitions into long-term care (Alzheimer Society of Ontario, n.d.).

Despite these opportunities for dementia-specific continuing education, PSWs consistently report the lowest levels of confidence and competence in providing dementia care compared to other healthcare professionals (Hapsari et al., 2022; Morgan, Kosteniuk, O'Connell, Dal Bello-Haas, et al., 2016; Nguyen et al., 2022; K. Zagrodney & Saks, 2017; Zeytinoglu et al., 2009). Kosteniuk and colleagues (2016) found that

compared to nurses, PSWs are more likely to report that dementia-specific continuing education programs do not address their learning needs, acting as a barrier to participation (Kosteniuk et al., 2016). Furthermore, recent shortages in the number of incoming PSW students have drastically undermined the growing available positions and vacancies within long-term care and is not sustainable to meet the needs of the growing dementia population (Grant, 2016; Ministry of Long-Term Care, 2020). Therefore, it is critical to equip the current PSW workforce with the competencies for optimal dementia care. This can be done by developing and implementing dementia-specific continuing education programs tailored and designed to the specific learning needs of PSWs in practice.

1.2 Adult Learning Theory

Educational theories provide conceptual frameworks for understanding how an individual acquires knowledge, skills, and attitudes that influence their behaviour (Mukhalalati & Taylor, 2019). Educational theories for children and adolescents, referred to as pedagogy, is different compared to learning theories in adulthood, termed andragogy (Merriam & Bierema, 2013). The pedagogical model of education arose in the seventh and twelfth centuries and has since been continuously adopted worldwide as the major model of institutional education from preschool to higher education levels (Knowles, 1990). This model outlines a set of beliefs where teachers and external stakeholders are solely responsible for determining when, why, where, and what should be learned (Knowles, 1990). Pedagogy is, therefore, a teacher-oriented model of learning, where learners are passive recipients of education, with little input or control over educational decisions. Andragogy is the opposite of pedagogy. First coined by Alexander Kapp (1799-1869) and refined by Malcolm Knowles throughout the nineteenth century to be known as 'adult learning theory', the andragogy model is learner-oriented and places adult learners in an active, self-directed role to control educational decisions (Knowles, 1990; Sorin-Peters, 2003). This means the responsibility of deciding when, why, how, and what is learned is placed on the adult learner, promoting curriculums that are developed around the needs and interests of the students (Ahmed et al., 2021; Knowles, 1990).

Adult learning theory encompasses six principles of adult learning that constitute a foundation for understanding how and why adults make decisions about their learning. The six principles are: 1) learning must be self-directed, 2) individual experiences provide the richest resource for learning, 3) an individual's readiness to learning is intimately linked with their social roles (e.g., parent, community member, child, working citizen, etc.), 4) learning must be problem-centred, 5) internal motivation drives an adult's openness to learning rather than external motivation, and 6) the learner must understand why learning is necessary and apply it to themselves (Knowles, 1990; Merriam & Bierema, 2013). Merriam & Bierema (2013) describe adult learning theory as a "mostly humanist philosophy wherein the individual is central, internally motivated and self-directed, and engages in learning for self-fulfillment, problem solving, and greater competency in life roles" (p. 56).

Self-direction in learning assumes the adult learner to be the most knowledgeable in their own learning capacities and deficiencies, with their experiences providing the strongest foundation for learning (Knowles, 1990). Experiences provide a basis to draw on to enhance relevancy and openness to learning (e.g., relating educational content to personal experiences) and stimulate a need for learning (Knowles, 1990). An individual's motivation and readiness to learn is driven by understanding or recognizing the need to learn something to cope with real-life situations within their social roles (e.g., caring for a person living with dementia as a PSW) (Knowles, 1990). Based on these principles, education programs tailored to the learners ensure that the content is relevant and often directly applicable to the needs and contexts of its intended audience (Knowles, 1990; Laxdal, 1982). The relevancy and applicability of topics covered in education programs to PSWs' caring for people living with dementia are, therefore, best perceived by PSWs themselves, necessitating their involvement in determining what, how, and when something needs to be learned. Adult learning theory therefore provides a theoretical foundation that positions PSWs as unique healthcare professionals with a rich composite of experience and knowledge in determining learning needs in dementia care. Furthermore, the theory promotes the importance of having PSWs play an active role in

designing, disseminating, and ensuring continuing education programs are relevant to their needs.

1.3 Significance and Outline of the Research

The objectives of this study are two-fold: 1) identify the dementia-specific learning needs of working PSWs in long-term care homes, and 2) examine how learning needs evolve as PSWs engage in a dementia-specific training program (Be EPIC). The following chapters aimed to describe the dementia-specific learning needs of PSWs in long-term care homes across London, Ontario.

PSWs are uniquely positioned to provide optimal, frequent, and highly personalized dementia care to residents because of their consistent and prominent role in daily personal care. This also positions PSWs as individuals with a large composite of knowledge concerning the daily care needs of persons living with dementia within the long-term care context. Yet, the perspectives of PSWs regarding their needs in dementiaspecific education remains under explored due their minimal representation in learning need assessments (Norris et al., 2024). Therefore, there is a significant knowledge gap regarding the learning needs of PSWs in long-term care homes who care for persons living with dementia. PSWs are provided the least comprehensive education and fewest continuing education opportunities compared to all other healthcare workers (Afzal et al., 2018; Hewko et al., 2015; Knopp-Sihota et al., 2023). As a result, PSWs consistently report the lowest levels of confidence and competence in dementia care (Hapsari et al., 2022; Morgan, Kosteniuk, O'Connell, Dal Bello-Haas, et al., 2016; Nguyen et al., 2022; K. Zagrodney & Saks, 2017; Zeytinoglu et al., 2009). Thus, education should be provided to PSWs in ways that are meaningful to their roles and address their unique learning needs.

The introductory chapter outlined relevant background information regarding my research study, including a brief description of dementia, quality of dementia care within long-term care homes, PSW education and roles, current dementia-specific continuing education programs and the study's theoretical background. Chapter two includes a

scoping review examining how PSWs are represented within dementia-specific learning need assessments and how their needs are assessed. The review presents a rationale for the current study, highlighting that PSWs' dementia-specific learning needs are largely under-explored due to their inadequate representation in academic literature. Chapter three details the qualitative methods employed in the current study, including the paradigm, methodology, research methods, rigor, and my positionality as a researcher. Chapter four presents the analytic findings of PSWs' dementia-specific learning needs, including learning methods and mediators for sustainable learning. These findings advocate for the improvement of PSWs' formal and continuing education by tailoring content and delivery to the needs of PSWs, as well as providing more equitable and ongoing opportunities for PSWs to engage in continuing education. Chapter five discusses the findings in relation to the current literature, my interpretations in relation to adult learning theory, the study's strengths and limitations, and implications for future research.

Chapter 2

2 The Representation of Personal Support Workers in Dementia-Specific Learning Need Assessments: A Scoping Review

2.1 Background

There is a growing pressure to equip healthcare workers with adequate competencies to care for the projected global increase in the incidence of people living with dementia (Marx et al., 2014; Public Health Agency of Canada, 2021; Rasmussen et al., 2023). Globally, over 55 million people are living with dementia (Alzheimer's Disease International, n.d.). This number is estimated to exceed 78 million by 2030 and 139 million by 2050 (Alzheimer's Disease International, n.d.; World Health Organization, 2017, 2023). Unregulated healthcare workers, such as PSWs, represent over half (65.5%) of the healthcare workers in formal care settings and provide the most daily direct care to people living with dementia in Canada (Grant, 2016; Ministry of Long-Term Care, 2020; Morgan, Kosteniuk, O'Connell, Dal Bello-Haas, et al., 2016; Rossiter & Godderis, 2020; Statistics Canada, 2021).

Despite their critical and essential role, PSWs in Canada are often overlooked, undervalued, and provided insufficient support compared to other healthcare workers (Hewko et al., 2015; Rossiter & Godderis, 2020). PSWs are the lowest-paid group of healthcare workers and are frequently considered the least skilled in the healthcare workforce (Hapsari et al., 2022; Zagrodney & Saks, 2017; Zeytinoglu et al., 2009). Relatively low wages are a common characteristic of PSW work, with the hourly range in Ontario being \$16.78-\$27.23, almost half of what nurses receive (Ministry of Long-Term Care, 2020). The ability to increase wages remains limited within government funding envelopes and therefore has caused wages to plateau over the past decade (Zagrodney et al., 2023). PSWs experience significant work-related stress and burnout rates, often due to their exposure to physically demanding activities, physical and verbal abuse, and inadequate support (Hignett et al., 2016). Moreover, they are often not provided with the opportunities to engage in training or continuing education programs to improve or maintain their skills, especially those that are specific to PSW practice and challenges (Afzal et al., 2018; Hewko et al., 2015; Knopp-Sihota et al., 2023). The insufficient support provided to PSWs also highlights the presence of sexism and discrimination faced by these essential workers, given that the role is predominantly occupied by women and historically marginalized communities (Block & Galabuzi, 2011; Sethna, 2013). In Ontario long-term care homes, 90% of PSWs are female and 41% are historically marginalized persons (Ministry of Long-Term Care, 2020). One explanation for the discrepancy in support is that PSW practice is not licensed or regulated by professional associations or government bodies unlike nurses and physicians (Grant, 2016; Kosteniuk et al., 2016). For example, in Canada, PSW education, certification standards, and titles are provided on a province-to-province basis, making it difficult to establish standardized practice regulations and ensure competency standards (Grant, 2016; Sethna, 2013).

Continuing education and professional development opportunities for healthcare workers are essential for maintaining and improving their clinical competencies including knowledge, skills, attitudes, and clinical behaviours (Al-Ismail et al., 2023; Laxdal, 1982). Competency in clinical contexts is most commonly defined as consistently demonstrating professional responsibility through practice and providing exceptional care by combining knowledge, skills, and professional attitudes (Fukada, 2018; Laxdal, 1982). Most healthcare workers are mandated to participate in continuing education programs, illuminating competency as a lifelong endeavour and demonstrating a commitment to excellence and adaptability to local care needs (Al-Ismail et al., 2023; Leach, 2002). The foundation for successful continuing education programs is ensuring they are relevant and directly applicable to the target audience through the identification and assessment of the learning needs and priorities of the learners (Laxdal, 1982; Rasmussen et al., 2023).

12

A learning need is defined as "the gap between current and optimal competence or performance" (Laxdal, 1982, p. 828). Learning need assessments can be used to establish a baseline of competency gaps to shape the development of continuing education programs and evaluate their effectiveness (Hauer & Quill, 2011). The assessments themselves are personal and specific to an individual or group and utilize a range of methods such as surveys, practice audits, peer reviews, observations, and interviews. Learning need assessments are useful in eliciting thoughtful reflections on gaps in personal abilities and educational topics of interest (Laxdal, 1982; Norman et al., 2004). For example, a 2020 systematic review found that training and education that are responsive to staff needs is a facilitator of effective dementia training (Cunningham et al., 2020). Moreover, evidence has demonstrated that programs developed based on welldesigned learning need assessments are the most successful in changing healthcare provider behaviours, practices, and competencies (Davis et al., 1995; Davis et al., 2008; Fox & Bennett, 1998; Grant, 2002; Laxdal, 1982). This has also been demonstrated in dementia-specific contexts (Kang et al., 2017; Karlin et al., 2017; Lawler et al., 2021; Mastel-Smith et al., 2019), however, the applicability of these programs to PSWs is unknown.

Currently, PSWs are not mandated to complete a registration exam to demonstrate their competence to enter the workforce, nor are they required to participate in continuing education to maintain competencies throughout their careers (Sethna, 2013). This may explain why previous evidence has reported that continuing education programs do not address the learning needs of PSWs (Kosteniuk et al., 2016). Yet, the extent to which PSWs' dementia-specific learning needs are included in learning need assessments remains unclear. Understanding the representation of PSWs' learning needs in academic literature can play a crucial role in advocating for their perspectives and ensuring their needs are considered and incorporated into future educational programs. Accordingly, the purpose of this scoping review is to examine the ways in which PSWs are included in dementia-specific learning needs assessments and how their needs are assessed.

13

2.2 Methods

The scoping review was conducted according to the Johanna Briggs Institute (JBI) Manual for Evidence Synthesis for Scoping Reviews (Peters et al., 2020) and reported according to the Preferred Reporting Items for Systematic Reviews and Meta-Analyses Extension for Scoping Reviews (PRISMA-ScR) checklist (Tricco et al., 2018). The search was conducted in July 2023. The databases PubMed, Scopus, CINAHL, PsychINFO (Ovid), Embase, and Medline (Ovid) were searched due to their focus on nursing and health sciences literature. The search terms used are stated in Table 1. The initial search strategy was produced in consultation with a faculty librarian at the University of Western Ontario, who assisted the first author (GN) in developing an initial search strategy using CINAHL. This strategy was then applied and modified to each remaining database when necessary. All final search strategies were reviewed and refined in consultation with the faculty librarian.

2.2.1 Inclusion and Exclusion Criteria

Articles were eligible for the title and abstract inclusion if they were primary research articles, written or translated into English and focused on identifying dementia-specific learning needs of groups or individuals containing unregulated healthcare workers. Learning needs were understood according to Laxdal's (1982) definition. Therefore, learning needs not resulting from a lack of education, skills, attitudes, personal qualities and/or opportunities to participate in continuing education were excluded. Review reference lists were searched manually to identify relevant articles.

To be eligible for full-text screening, the sample population needed to include unregulated healthcare workers or a broad description of the sample population such as "long-term care staff", "home care staff", "nursing home personnel", or "nursing staff". This eliminated the risk of excluding papers that grouped PSWs with other health and social care workers in the sample populations. See Table 1 for the complete list of occupational titles included in the search strategy. During the full-text screening, studies were eligible for inclusion if the sample population included PSWs, nursing assistants (NAs), care aides, and healthcare assistants (HCAs), or defined the sample population in accordance with PSW roles and responsibilities. Definitions from PSW competency frameworks or practice guides were used to determine eligibility. These documents commonly described PSWs as unregulated healthcare professionals providing and assisting vulnerable individuals across the lifespan with activities of daily living (Government of Alberta, 2018; Government of British Columbia, 2014; Ministry of Training, Colleges and Universities, 2022; Nova Scotia Department of Health and Wellness, 2019). Studies that did not provide sufficient definitions of the study population, or included a definition that did not align with the PSW definition were excluded. Studies containing heterogeneous sample populations were included in the review to ensure that the examination of PSWs and their learning needs within the learning need assessment literature was comprehensive in scope. Studies were further excluded if they did not focus on identifying dementia-specific learning needs, conducting a learning need assessment related to dementia care, or including diseases other than dementia (e.g., cancer, schizophrenia, etc.).

Table 1: Search strategy keywords and terms

Keywords	Search Terms ¹
Learning needs	Learning need or learning need assessment or learning need analysis or needs assessment or education need or training need or information need
AND	
Dementia	Dementia ¹ or Alzheimer's disease ¹ or Dementia patient ¹ or Lewy Body Disease ¹ or Frontotemporal dementia ¹ or Vascular dementia ¹ or Delirium, Dementia, Amnestic, Cognitive Disorders ¹ or Multi-Infarct dementia ¹ or AIDs dementia complex ¹ or Senile dementia ¹ or Presenile dementia ¹ or Lewy Body dementia or Parkinson's or frontotemporal or Creutfeldt-Jakob or Wernicke-Korsakoff or Mixed dementia or Cognitive impairment or mild cognitive impairment
AND	

15

Personal Support Personal support worker or PSW or nursing assistant¹ or nurse aide¹ or healthcare assistant or healthcare aide or HCA or long-term care staff or home care staff or unlicensed healthcare worker or unlicensed health personnel or home health aides or nursing home personnel or nursing staff ¹

Note. ¹MeSH headings applied where applicable.

2.2.2 Evidence Selection

The first author (GN) and two undergraduate research assistants used the platform Covidence to conduct all screenings and organize articles. Covidence automatically removed all duplicate studies once the research strategy results from each database were imported. Title and abstract screening were conducted based on the inclusion and exclusion criteria. The screening was an iterative and collaborative process, as many studies included various terms for PSWs (e.g., certified and uncertified NAs, HCAs, etc.), used broad terms to describe the sample population (e.g., care team) and conducted learning need assessments in conjunction with other research aims (i.e., assessments of continuous training programs). The inclusion and exclusion criteria were refined based on the literature during the title and abstract screening and collaborative discussions between research team members to ensure relevant studies were not missed. Articles flagged as "maybe" included reviews and those without an abstract. GN completed manual searching of review reference lists independently. All final screening decisions were made collaboratively.

2.2.3 Data Extraction and Charting

Data was charted by GN and an undergraduate research assistant. The chart was based on the review's objectives (i.e., sample population, reporting of learning needs from PSWs, methods and study design, objectives, etc.) and iteratively refined as evidence was extracted. A descriptive summary table was developed in Excel with the following headings: first author and publication year, objective(s), country, study type, sample population, setting, method, learning needs assessment tool(s), and if PSW learning needs were analyzed and/or reported.

2.3 Results

2.3.1 Sources of Evidence Selection

In April 2023, a total of 729 articles were derived from a search of six electronic databases. Once duplicates were removed (n = 341) automatically by Covidence, 338 studies underwent title and abstract screening by two research team members. In July 2023, the search strategy was updated in consultation with MYS to include additional dementia keywords and MeSH terms and reran through all six databases. An additional 39 studies were identified, with five being automatically removed as duplicates. Thus, as seen in Figure 1, a total of 763 articles were identified and 346 were removed as duplicates. In total, 417 studies underwent title and abstract screening. Based on the inclusion and exclusion criteria, 362 articles were excluded, with 55 full-text articles to be assessed for eligibility. A total of 38 articles were identified through manual reference list searches. The remaining 17 studies were considered eligible for this review.



Figure 1: PRISMA diagram

2.3.2 Characteristics of Sources of Evidence

The 17 studies included in this review spanned 19 years (2003 - 2022), with 13 (81.2%) published within the last decade. The selected articles were conducted in the United States of America (n = 4), United Kingdom (n = 4), Netherlands (n = 2), Australia (n = 2), Malta (n = 1), Ireland (n = 1), New Zealand (n = 1), Canada (n = 1), and Japan (n = 1). Six (35.3%) studies were conducted in both long-term care and home care settings while the remaining included five (29.4%) in long-term care exclusively (including nursing homes and assisted living homes), three (17.6%) in home care, two (11.7%) in hospital wards and one (5%) unspecified. Table 2 summarizes the research objectives, sample population, inclusion of PSW learning needs, and study design/methods of the 17 studies.

Table 2: Summary of included articles

	Author, Year	Aim of Study	Sample Population	Study Design/Methods	PSW Learning Needs Reported
1	Adler <i>et al.</i> , 2015	Investigated the dementia- specific learning needs of staff at community-based outpatient clinics who care for veterans living with dementia.	n = 105 healthcare workers (62 nurses, 23 support staff, and 12 physicians)	Quantitative. A survey explored interests and perspectives on dementia-specific education using the Approaches to Dementia Questionnaire. Participants ranked education topics related to medical and psychosocial management.	No
2	Annear <i>et al.</i> , 2018	Investigated the self- reported learning needs of a randomly selected national sample of geriatric healthcare workers.	n = 117 geriatric healthcare workers (74 care worker and 22 other (e.g., managerial or service staff), 13 other geriatric healthcare staff (e.g., physical therapists), 6 nurses, a clinical educator, and a medical student)	Quantitative. A survey explored the self-identified perceptions of learning needs and confidence in dementia care, preferences concerning education, and demographic questions.	No
3	Attard, Sammut, and Scerri, 2020	Investigated the knowledge, attitudes, and learning needs of healthcare professionals caring for people living with dementia. Explored the	n = 207 healthcare professionals (103 NAs and 104 nurses)	Quantitative. The survey included three measurement tools: (1) the Alzheimer's Disease Knowledge Scale to assess carers' knowledge about the	Yes

	Author, Year	Aim of Study relationships between knowledge, attitudes, and demographic characteristics.	Sample Population	Study Design/Methods disease; true/false statements; (2) the Dementia Attitude Scale to identify carers' attitudes towards people with dementia, and; (3) the Dementia Learning Needs Assessment tool (rating 20-items to indicate learning needs).	PSW Learning Needs Reported
4	Bolt <i>et al.</i> , 2020	Explored the palliative care learning needs of staff working with people living with dementia, the types of support needed, and compare the educational levels of staff between home care and long-term care settings.	n = 416 staff members (164 RNs, 218 certified NAs, and 34 uncertified NAs)	Quantitative. The survey focused on palliative caregiving and end-of-life communication, and interprofessional collaboration and transitions of care, self-perceived competence in providing palliative care for people living with dementia, learning needs, and preferred forms of support.	Yes
5	Chang <i>et</i> <i>al.</i> , 2009	Investigated the challenges of providing care for people living with advanced dementia in residential aged-care facilities from the perspectives of multiple key healthcare professionals.	Focus groups n = 29 (4 general practitioners and 4 palliative care specialty staff, 5 palliative care RNs, 4 volunteers, 3 senior RNs, 2 palliative care volunteer managers, 2 RNs from a community psychogeriatrics team, a social worker, an RN from a community aged care	Qualitative. Focus group discussion centered on the needs and deficits in service delivery. Interviews focused on building from focus group information.	No

	Author, Year	Aim of Study	Sample Population assessment team, an RN from acute care, a recreational therapist, and an NA) Interviews n = 20 (4 NAs, 3 RNs, 2 directors of nursing, 2 enrolled nurses, and 2 therapists, 1 specialist medical officer, 1 psycho- geriatric nurse specialist, a geriatric nurse specialist, an	Study Design/Methods	PSW Learning Needs Reported
			OT, and a bereavement counsellor)		
6	Foster <i>et al.</i> , 2019	Uncovered and critically examined healthcare assistants' experiences of providing person-centered care to people living with dementia during end-of-life stages and perspectives on formal education preparedness for palliative and dementia care.	n = 34 HCAs	Qualitative. Semi-structured interviews were conducted to explore experiences of end-of-life care for people living with dementia, communication with staff, residents, and families relating to end- of-life, and education experiences.	Yes

	Author, Year	Aim of Study	Sample Population	Study Design/Methods	PSW Learning Needs Reported
7	Gandesha <i>et al.</i> , 2012	Assessing how hospital staff view the sufficiency of formal training preparation for caring for people living with dementia	n= 1779 (270 physicians, 968 nurses, and 541 HCAs)	Qualitative. Two audits were performed at the hospital level (open to all) and ward- level (nominated ward staff only). Questionnaires were shared with staff during the ward-level audit, inquiring about the sufficiency of training in 13 areas related to dementia care.	Yes
8	Gurnik and Hollid- Sawyer, 2003	Uncovered the learning needs of frontline care staff to develop a training manual based tailored to their needs. Also evaluated the effectiveness and helpfulness of the training manual.	n = 24 care staff. Did not specify the titles of care staff but defined them as "staff who conduct the day-to- day full range of care needed by residents".	Mixed Methods. A survey included open-ended items on four topics areas: (1) role needs; (2) social interaction needs; (3) personal growth needs, and (4) knowledge and understanding of dementia and the aging process. A second survey and focus group were conducted to collect post-training feedback.	No

	Author, Year	Aim of Study	Sample Population	Study Design/Methods	PSW Learning Needs Reported
9	Leverton <i>et al.</i> , 2021	To identify the skills, training and support needs of homecare workers providing care for people living with dementia from the perspectives of multiple key stakeholders involved in homecare provision.	n = 82 participants. (11 people living with dementia, 22 family caregivers, 19 health and social care professionals, and 30 homecare staff including 7 managers, 4 office staff, and 19 homecare workers)	Qualitative. Observations of homecare workers (n = 16) and people living with dementia and family caregiver care interactions. Semi-structured interviews with all stakeholders focused on perspectives on preferences for content and delivery method of a training program for homecare workers caring for people living with dementia.	Yes
10	Marx <i>et al.</i> , 2014	Identified dementia knowledge, learning needs, and experiences within the care environment.	n = 37 staff members of a specialized geropsychiatry unit (15 nursing staff (patient care technicians, nurses, and NAs), 8 physiotherapists, 7 occupational therapists (OT), 6 recreation therapists, and a speech-language pathologist)	Quantitative. A survey assessed dementia knowledge (using a modified Alzheimer's Disease Knowledge Scale), attitudes toward dementia- specific education needs, and the care environment.	No

	Author, Year	Aim of Study	Sample Population	Study Design/Methods	PSW Learning Needs Reported
11	Mayrhofer, Goodman and Smeeton, 2016	Understand the perspectives of dementia champions regarding their roles, remit, and training needs.	n = 34 dementia champions. (13 nurses, therapists and emergency staff, 11 senior practitioners, 6 managers, and 2 support staff)	Quantitative. A survey explored perceptions of the dementia champion role, its remit, and perceived learning needs.	No
12	Morgan <i>et</i> <i>al.</i> , 2016	Investigated perceived competency in dementia- related homecare activities, frequency of activities, and interest in dementia-specific continuing education topics.	n = 82 homecare staff (41 NAs, 20 home care nurses, 10 case managers (5 nurses and 5 social workers) and 11 dual nurse/case manager roles (all nurses))	Quantitative. A survey identified dementia-related work activities based on the frequency of their occurrence, perceived competence, and ranking as a preferred continuing education topic.	Yes
13	Nguyen <i>et</i> <i>al.</i> , 2022	Investigated the communication learning needs, challenges, coping strategies, and influencing factors of healthcare workers caring for people living with dementia.	n = 258 carers (141 formal caregivers, 57 informal or family caregivers, 11 volunteer caregivers, and 49 other (e.g. registered nurses (RNs)))	Quantitative. A survey identified communication challenges in dementia care and related coping strategies, perceived factors that influence communication, and communication learning needs.	No
14	Szymcynska & Innes, 2011	Evaluated a training workshop developed for rural health and social-care staff providing services to	 n = 18 health and social care staff, 10 of whom provided dementia care. Staff demographics were not provided but the sample 	Qualitative. Participants attended a workshop developed to address dementia- specific learning needs, raise awareness of dementia, identify	No

	Author, Year	Aim of Study people living with dementia, regarding its success in achieving learning outcomes and addressing learning needs.	Sample Population population was defined as: "all staff who may offer services to persons with dementia were encouraged to participate."	Study Design/Methods service and learning needs, and understand staff's roles in providing care to people living with dementia. Informal discussions were conducted to collect information about staff experiences with the workshop. A post-workshop questionnaire was sent to participants to identify satisfaction with the information and content, how they thought the course would influence their practice, and identify future learning needs	PSW Learning Needs Reported
15	Timmons <i>et</i> <i>al.</i> , 2021	Investigated staff training in dementia and palliative care, learning needs, readiness for change, and the care environment.	n = 60 staff members (2 nurses, 27 HCAs, and 10 health care practitioners (only used demographic data))	Quantitative. A survey explored barriers and facilitators to implementing changes in long-term care facilities, experiences of powerlessness, confidence, and de-motivation, and asked participants to rank their top three learning needs relating to medication, pain, hydration and nutrition. It also allowed for free text responses as well for learning needs that were not included under the three guidance topics.	Yes
	Author, Year	Aim of Study	Sample Population	Study Design/Methods	PSW Learning Needs Reported
----	---	---	---	---	--------------------------------------
16	Toteh Osakwe et al., 2022	Investigated the dementia- specific learning needs of home HCAs regarding dementia care plans	n = 25 HCAs	Qualitative. Interview questions were specific to processes of care in the homecare setting and related outcomes such as interactions with homecare nurses and desired information from nurses regarding care plans.	Yes
17	Verkaik <i>et</i> <i>al.</i> , 2017	Investigated the learning needs and preparedness of nursing staff to provide self- management support to people living with dementia.	n = 206 (questionnaire) (46 RNs and 160 NAs) n = 12 (interviews) (8 RNs and 4 NAs)	Mixed Methods. Quantitative data were derived from a secondary analysis of a questionnaire on nursing staff's experiences, opinions and perceived knowledge on self-management support (SMS). Semi-structured interviews were conducted to further inquire and deepen the data from the questionnaires. The interviews also covered topics on essential SMS, practice and experiences with SMS, interest in self-management support, practices and experiences of SMS with people living with dementia and learning needs for SMS in dementia care.	Yes

2.3.3 PSW Representation in Sample Population

Of the 17 included studies, two (11.7%) reported the dementia-specific learning needs of PSWs exclusively (Foster et al., 2019; Toteh Osakwe et al., 2022). Twelve (70.5%) studies involved multi-professional sample populations of at least two healthcare professions. Of these, eight (47%) studies included at least three distinct professions, such as physicians, nurses, and PSWs (Adler et al., 2015; Chang et al., 2009; Gandesha et al., 2012; Leverton et al., 2021; Marx et al., 2014; Mayrhofer et al., 2016; Morgan, Kosteniuk, O'Connell, Dal Bello-Haas, et al., 2016; Nguyen et al., 2022). The remaining four (23.5%) studies grouped PSWs with nurses (Attard et al., 2020; Bolt et al., 2020; Timmons et al., 2021; Verkaik et al., 2017). Three (17.6%) studies did not specify the official occupational titles of participants but rather described them broadly such as "care staff" or "support staff", and a respective definition (Annear et al., 2017; Gurnik & Hollis-Sawyer, 2003; Szymczynska & Innes, 2011).

PSWs represented an equal proportion of the sample population with other healthcare workers in two (11.7%) studies (Attard et al., 2020; Morgan et al., 2016) and below half of the sample population in seven (41.1%) (Adler et al., 2015; Chang et al., 2009; Gandesha et al., 2012; Leverton et al., 2021; Marx et al., 2014; Mayrhofer et al., 2016; Timmons et al., 2021). PSWs represented greater than half of the sample population in five (29.4%) studies (Bolt et al., 2020; Foster et al., 2019; Nguyen et al., 2022; Toteh Osakwe et al., 2022; Verkaik et al., 2017). Two of these studies contained sample populations of nursing assistants exclusively while the remaining three used a multiprofessional sample population. Notably, the ratio of PSWs to other healthcare workers fluctuated depending on the type of data collection method being used. In Verkaik et al., (2017), data was initially collected through a survey and followed by interviews. Nursing assistants represented three times the number of nurses (160 nursing assistants to 46 nurses) in the survey responses, which inquired about staff experiences caring for people living with dementia and opinions and perceived expertise regarding self-management support (Verkaik et al., 2017). In contrast, the number of nurses was twice that of nursing assistants (4 nursing assistants to 8 nurses) in follow-up interviews to obtain a deeper

insight and context into the survey results (Verkaik et al., 2017). Furthermore, in Leverton and colleagues (2021) a larger proportion of PSWs (referred to as homecare workers) consented to be observed (n=16) compared to being interviewed (n = 4). The proportion of PSWs was unidentifiable in three (17.6%) studies due to the use of broad occupation terms such as "care staff" and "support staff" (Annear et al., 2017; Gurnik & Hollis-Sawyer, 2003; Szymczynska & Innes, 2011).

Eight (47%) of the studies did not analyze and/or report learning need data based on occupational titles of the sample population (Adler et al., 2015; Annear et al., 2017; Chang et al., 2009; Gurnik & Hollis-Sawyer, 2003; Marx et al., 2014; Mayrhofer et al., 2016; Nguyen et al., 2022; Szymczynska & Innes, 2011), while nine (52.9%) did (Attard et al., 2020; Bolt et al., 2020; Foster et al., 2019; Gandesha et al., 2012; Leverton et al., 2021; Morgan et al., 2016; Timmons et al., 2021; Toteh Osakwe et al., 2022; Verkaik et al., 2017). Of those that reported PSW-specific learning needs, four were quantitative studies, four were qualitative, and one was mixed methods. Of the four studies that included nurses and PSWs in the sample population, all studies disaggregated the data based on occupational title, however, only three reported the findings based on disaggregation while the remaining article did not report differences in learning needs since there was an abundance of similarities in the results (Bolt et al., 2020).

2.3.4 Data Collection Methods

The studies included ten (58.8%) quantitative, five (29.4%) qualitative and two (11.7%) mixed methods. Surveys were used in 13 (76.4%) of the studies, with nine (52.9%) using them as the sole method of data collection (Adler et al., 2015; Annear et al., 2017; Attard et al., 2020; Bolt et al., 2020; Gurnik & Hollis-Sawyer, 2003; Marx et al., 2014; Mayrhofer et al., 2016; Morgan et al., 2016, 2016; Szymczynska & Innes, 2011; Timmons et al., 2021; Verkaik et al., 2017). Three (17.6%) studies used surveys in conjunction with qualitative methods such as interviews (Verkaik et al., 2017), focus groups (Gurnik & Hollis-Sawyer, 2003), or opportunistic discussions (Szymczynska & Innes, 2011). Interviews and/or focus groups alone were less frequent, as only three (17.6%) studies adopted these as the sole method for data collection (Chang et al., 2009;

Foster et al., 2019; Toteh Osakwe et al., 2022). Leverton and colleagues (2021) combined the use of interviews with direct participant observation. The mixed method studies used interviews and informal discussions to support and deepen the understanding of quantitative insights from surveys (Gurnik & Hollis-Sawyer, 2003; Verkaik et al., 2017). See Table 3 for a summary of the reported methods adopted in the included studies.

All 17 studies collected data based on the participants' self-assessment of their own knowledge, skills, training sufficiency, and/or learning needs. Among the 13 studies that used surveys, four used standardized survey tools, such as the Dementia Attitude Scale (Attard et al., 2020), the Dementia Learning Needs Assessment tool (Attard et al., 2020), Approaches to Dementia Questionnaire (Adler et al., 2015), and the Alzheimer's Disease Knowledge Scale (Attard et al., 2020) (with some modification in Marx et al., (2014)). The remaining studies used surveys developed by either the primary researchers or secondary data analysis of previous survey responses.

Data Collection Method	<i>n /</i> 17		
	(%)		
Survey/Questionnaire	10 (58.8)		
Interview	2 (11.7)		
Survey and Focus Groups	1 (5)		
Survey and Informal Discussions	1 (5)		
Survey and Interview	1 (5)		
Interview and Focus Group	1 (5)		
Survey and Audit	1 (5)		

Table 3: Methods of data collection

2.3.5 Dementia-Specific Learning Needs Assessed in Included Studies

Although all 17 reviewed studies conducted a dementia-specific learning needs assessment, 11 (64.7%) had a specific focus on an aspect of dementia care, while six (35.3%) investigated dementia-specific learning needs as a broad, all-inclusive topic (Adler et al., 2015; Annear et al., 2017; Attard et al., 2020; Gandesha et al., 2012;

Leverton et al., 2021; Marx et al., 2014). See Table 4 for the specific dementia-care topics. All six studies focusing on dementia care as a broad topic for learning need assessments were quantitative and used surveys and questionnaires as the main data collection method.

Table 4:	Study	objectives
----------	-------	------------

Learning Need Assessment Focus			
	(%)		
Broad			
Dementia Care	6 (35.3)		
Specific			
Palliative Care	3 (17.6)		
Evaluation of an education program developed from a learning needs	2 (11.7)		
assessment			
Advanced Dementia	1 (5)		
Dementia Care Plans	1 (5)		
Self-Management	1 (5)		
Dementia Champion Roles	1 (5)		

Six of the nine articles that aggregated and reported PSW-specific learning needs had a narrow focus on an aspect of dementia care. These included learning needs related to palliative and end-of-life care for people living with dementia (Bolt et al., 2020; Foster et al., 2019; Timmons et al., 2021), self-management support (Verkaik et al., 2017), dementia home care activities (Morgan et al., 2016) and information for care plans (Toteh Oskawe et al., 2022). Attard et al., (2020), Leverton et al., (2021), and Gandesha et al., (2012) disaggregated and reported PSW's specific learning needs related to dementia as a broad topic of investigation relative to other healthcare professionals within the sample populations. The two (11.7%) articles that used PSWs exclusively in the sample population also had a specific focus on dementia care learning needs. Foster et al., (2019) investigated the learning needs of home care NAs related to palliative care for persons living with dementia and Toteh Oskawe et al., (2022) explored the information needs of NAs regarding dementia care plans.

2.4 Discussion

This review demonstrated a critical gap in the validity of data which involves but does not report on PSWs' needs. PSWs' distinct perspectives, skills, and knowledge are overlooked and oversaturated with those of other healthcare workers. This gap makes it difficult for key stakeholders to understand PSW challenges, experiences, and areas of knowledge and skill needed to provide optimal dementia care. Compared to other healthcare workers, PSWs are underrepresented in academic literature, lending them few opportunities to be involved in evidence-based research, practices, or contribute their perspectives in meaningful and impactful ways.

Learning needs will vary based on educational qualifications and experience levels, highlighting the importance of conducting assessments with PSWs exclusively in mind (Bing-Jonsson et al., 2013). Only two studies (Foster et al., 2019; Toteh Osakwe et al., 2022) in this review included PSWs exclusively, while the remaining 15 used a multiprofessional sample. Furthermore, nearly half reported PSW-specific learning needs amongst other healthcare workers within the sample populations. These results are consistent with other learning need assessments, which have included PSWs in sample populations with other healthcare professionals, but report needs in a summative nature rather than based on occupational titles (Gillham et al., 2018). The grouping of PSWs with other, similar occupations during data collection and analysis is a common and recognized issue among researchers and organizations (Grant, 2016). Historically, the voices of PSWs have been ignored and devalued by organizations, as many have reported feeling their needs and concerns frequently being dismissed, minimized, or slow to resolve (Hapsari et al., 2022; Rossiter & Godderis, 2020). This was also recently demonstrated during the COVID-19 pandemic, where Ontario PSWs did not receive the same concern, treatment, or resources for their risk of contracting the disease, nor were they included in social configurations of infection risk assessments (Rossiter & Godderis, 2020). The invisibility and minimization of the importance of PSW knowledge, perspectives, and voices perpetuate the narrative of PSWs being unskilled, poorly compensated, and invisible/illegitimate as a professional healthcare occupation

(Zagrodney & Saks, 2017; Zeytinoglu et al., 2009). Consequently, PSWs being overlooked and undervalued is reflected in the inaccuracy of learning needs unique to their profession being represented in dementia-specific learning need assessments. This therefore creates difficulties in understanding where challenges in delivering optimal dementia care exist and identifying opportunities to address specific gaps in PSW competencies.

In terms of data collection methods, this review demonstrates that researchers most commonly use quantitative approaches, namely surveys, as the predominant method for assessing dementia-specific learning needs of PSWs. This finding aligns with a recent scoping review on learning need assessments, which identified the widespread use of quantitative approaches (Al-Ismail et al., 2023). Quantitative approaches to learning need assessments are often attractive due to their cost-effectiveness, ability to objectively measure outcomes (i.e., dementia knowledge), and use of collecting data from large groups (Grant, 2002; Hauer & Quill, 2011). Surveys and questionnaires typically involve testing participants' knowledge, attitudes, and/or skills, as well as ranking their preestablished learning need topics (Adler et al., 2015; Attard et al., 2020; Carnahan et al., 2019; Chung & Lai, 2003; Lawler et al., 2021). This approach can help researchers understand the prioritization of specific learning needs by conducting frequency analyses and encouraging participants to select and rank predetermined needs (Grant, 2002). Some potential drawbacks of using quantitative methods include a large dependency on the quality of survey questions to elicit truthful responses and the lack of choice provided to participants to discuss learning needs candidly (Grant, 2002; Hauer & Quill, 2011). Quantitative approaches have also been criticized for their scope of needs identification, which focus on immediate needs rather than long-term, foundational needs, and are dependent on the quality of the questions (Al-Ismail et al., 2023; Ferreira & Abbad, 2013).

Qualitative methods are effective mechanisms for identifying learning needs and exploring the contextual factors that shape those needs at individual or group levels. Interviews and focus groups encourage candid responses and provide deep insight into an

individual's or group's perspectives on perceived learning needs, attitudes and beliefs that influence behaviours, potentially offering a more proactive approach in comparison to quantitative methods (Daley & Wilson, 1999; Ferreira & Abbad, 2013; Hauer & Quill, 2011). Previous dementia specific learning need assessments have used focus groups (Allen et al., 2005; Daley & Wilson, 1999; Gillham et al., 2018; Hsiao et al., 2016; Xu et al., 2018) and reported that they allow participants to create meaning collaboratively, engage in discussions and challenge each other's perspectives to uncover knowledge (Daley & Wilson, 1999). However, qualitative methods that identify and report subjective data have been criticized for being "less accurate" when compared to objective, quantitative methods (Grant, 2002). Other methods for collecting learning need data include gap analyses, standardized patients, video assessments, peer reviews, risk assessments, observations, and chart audits (Grant, 2002; Lockyer, 1998). Evidence suggests that for learning need assessments to be comprehensive and limit the potential to miss needs, the methods should be based on the local environment and culture, professional reflection and judgement, discussions, and incorporate various levels of data (Al-Ismail et al., 2023; Grant, 2002).

All studies included in the scoping review used self-assessment methods, which relied on the participant's ability to recognize their own deficiencies in dementia care. For instance, some studies asked participants to rank or select from a list of education topics based on their interest or needs (Morgan et al., 2016; Timmons et al., 2021). Self-assessment of learning needs is an essential component of professional development, especially within medical and health-related disciplines, such as medicine (Al-Ismail et al., 2023; Davis et al., 2008). However, previous evidence has demonstrated that self-reporting and assessment of needs may not be accurate or align with objective measures (Davis et al., 2008). The use of self-assessments has therefore been cautioned as the sole method to data collection in learning need assessments, as the responses may not be reflective of actual learning needs (those that are difficult to admit, perceive in oneself, or are not of interest to learners) and may be subject to biases such as reporting bias and self-enhancement bias (Al-Ismail et al., 2023; Raley, 2020). These biases can be mitigated by pairing self-assessments with more objective methods such as interacting

with standardized patients, standardized survey tools, or peer reviews involving individuals who are knowledgeable about each other and/or participate in similar activities (Campbell, 1990; Raley, 2020). Despite these potential challenges, self-perceived learning needs are vital when assessments are being used to design continuing education programs, as motivation is key for the successful uptake of knowledge (Kosyluk & Rigg, 2021). Changes to behaviours, attitudes, knowledge, and skills are more likely to arise when learners identify the education contents as being personally meaningful and relevant (Kosyluk & Rigg, 2021).

In this review, dementia-specific learning need assessments focused mostly on specific aspects of dementia care, rather than gathering learning need information on dementia care as a broad topic. PSWs are reported to have the least amount of education and the fewest opportunities to participate in continuous education programs compared to other healthcare workers (Chen et al., 2018; Kosteniuk et al., 2016; Timmons et al., 2021) and consider themselves "beginners" in terms of knowledge and skills in dementia care (Adler et al., 2015). Due to the lack of continuing education and training support provided to PSWs in dementia care, their learning needs should be explored in a broad manner to understand the depths and range of their potential for professional growth. Studies have suggested that PSWs are eager to participate in dementia specific continuing education, in part to improve their competency and confidence in completing dementia care activities (Morgan et al., 2016). In a 2016 cross-sectional Canadian study, discrepancies in competency among home care nurses and nursing aides were reported in almost all dementia-related care tasks, such as communication skills with family about dementia and managing responsive behaviours (Morgan et al., 2016). Although all aspects of care are equally as important, the lack of skill and knowledge provided to PSWs through formal education and continuing education opportunities suggests that a more comprehensive inquiry should be made regarding dementia care as a broad topic.

2.4.1 Strengths and Limitations

The strengths of this review include the use of six online databases to comprehensively search for all articles eligible for inclusion. Moreover, the results of the review add to the

literature on PSWs and explicitly present gaps in research surrounding their learning needs related to dementia care. This provides a unique overview of how PSWs are presented within dementia-specific learning need assessments and contributes to the literature on the lack of representation of PSWs within research. The content in this review provides readers with an overview on the current state of PSW involvement in dementia-specific research and how their perspectives and voices are included in data collection and analysis.

A limitation of this scoping review is that it only included studies written or translated in English. Articles written in French or other languages were not included and therefore potential studies could have been missed. Further, this review did not include grey literature, which could have resulted in the exclusion of learning need assessments done by care organizations or governing bodies.

2.4.2 Implications for Future Research

More research needs to be conducted on PSW-specific learning needs in dementia care. Future research in this area should include qualitative or mixed methods to identify learning needs and empower PSWs in the process. Compared to quantitative methods, qualitative and mixed method approaches can provide opportunities to conceptualize the holistic environments which can aid in the understanding of interpersonal and environmental influences on learning needs, such as educational preparedness and workplace support. Further research should also explore learning needs within dementia care as a broad topic, to illuminate the depth, variation, and complexity of needs that exist for PSWs.

2.5 Conclusion

PSWs spend the most time on average caring for people living with dementia in organized care settings and therefore play a critical professional role in ensuring the delivery of optimal dementia care. However, optimal care can only be delivered if the value of PSW's is formally acknowledged, and they are provided the necessary support to thrive in their dementia care roles in meaningful ways. Their perspectives, opinions,

and learning needs should therefore be investigated in a manner that acknowledges the value of PSWs alongside their skills, knowledge, and experience caring for people living with dementia.

This review has demonstrated a gap in knowledge concerning the dementia specific learning needs of PSWs. Ensuring PSWs have ample opportunities to improve their knowledge, skills, and attitudes towards dementia care is a priority to improve PSW satisfaction, reduce work-related stressors and optimize the quality of care provided to people living with dementia. Hughes and colleagues (2008) suggest that simply the presence of training opportunities for healthcare workers has a significant influence on their self-perceived confidence in caring for people living with dementia. Empirical evidence has also suggested that effective training programs targeted to healthcare workers' needs can have positive effects on both healthcare workers and their clients (Li et al., 2021). Providing these opportunities can improve worker satisfaction with workplace roles, potentially reducing turnover rates and improving client care satisfaction by ensuring consistent and highly skilled staff (Li et al., 2021). Uncovering and understanding PSWs' views of dementia-specific learning needs is a crucially necessary step to identifying underlying problems in delivering essential care within formal care organizations. Improved understanding of the gaps in PSWs' knowledge, attitudes, and/or skills related to dementia care are urgently needed to better inform and focus discipline specific and concentrated PSW training in the future.

Chapter 3

3 Method

This study used interpretive description (Thorne, 2016) to identify and examine the dementia-specific learning needs of PSWs participating in a dementia-focused education program. A secondary analysis of focus group data collected between September 2019 to March 2020 was performed.

3.1 Paradigm

This study used a constructivist paradigm based on the notion that learning needs are unique to individuals based on personal experiences. Constructivism views reality as mentally constructed with multiple forms that are local and specific in nature and dependent on the context of the individual or groups of individuals (Guba & Lincoln, 1994). Knowledge is therefore based on subjective experiences and interpretations, as well as socially constructed through dialogue and social interactions (Carpenter & Suto, 2008; Guba & Lincoln, 1994). Thus, rather than certain learning needs being more or less "true" than others, all can be viewed as valid and understood by being derived by personal experiences working in long-term care and/or caring for people living with dementia.

3.2 Methodology: Interpretive Description

This study uses an equity lens to identify the dementia-specific learning needs of PSWs in long-term care homes and critically examine the context in which they exist. In educational contexts, equity can be understood and addressed by identifying barriers and discriminatory practices to ensure students of diverse backgrounds have the opportunity to advance and reach their full potential (Government of Ontario, 2020; Green, 2020). The primary aim of the study is to inform future continuing dementia education programs on the unique needs of PSWs in long-term care and advocate for the inclusion of PSW voices in future research and continuing education programs.

Interpretive description seeks to pragmatically answer broad questions iteratively and inductively concerning subjective clinical experiences from a holistic, interpretive, and relational perspective (Burdine et al., 2021). Interpretation is used to understand subjective experiences within larger social contexts (Thorne, 2016). Pragmatism focuses on developing knowledge and producing findings with practical outcomes that are highly applicable to real-world contexts (DeForge & Shaw, 2012). Thus, inquiry is driven by the desire to address practical problems or understand experiences to generate knowledge that possess clear implications for the local context (e.g., clinical practice, community wellness, etc.), as opposed to metaphysical (i.e., ontology, epistemology, axiology, and methodology) considerations (DeForge & Shaw, 2012; Thorne, 2016).

The philosophical foundation of interpretive description is within interpretive naturalistic orientations (Thorne et al., 2004), which focuses on studying phenomena in their natural context to understand the complexities and nuances of real-world experiences. Key philosophical principles of naturalistic inquiry include: (1) reality is multiple, complex, contextually constructed, and subjective and must be studied holistically, (2) the researcher and participants interact to influence each other, and (3) a single theory is unable to encompass the multiple realities likely to be encountered during interpretive description (Thorne, 2016). Ontologically, interpretive description acknowledges the constructed and contextual nature of human experiences when interpreting and understanding subjective data (Burdine et al., 2021; Thorne, 2016). Experiences and environmental context are fundamental in shaping what is understood as real or true in the world, and contribute to the composition of individual identity and consciousness (DeForge & Shaw, 2012). Epistemologically, environmental contexts and experiences provide meaning for the development of knowledge, and individuals learn from and adapt with their local contexts (DeForge & Shaw, 2012). Knowledge is constructed through dialectic interactivity, typically between the investigator and participants, and is based on subjective experiences and interpretations which is impossible to partition from objectivism (Thorne et al., 2004). Yet, interpretative description supports the integration of theoretical insights in interpreting data, namely, the incorporation of various perspectives to inform the analysis with the intention to enhance relevance and

applicability of the findings (Thorne, 2016). Therefore, the methodology encourages and promotes the generation of knowledge that possesses clear implications for the local context rather than simply theorizing (Jensen et al., 2018; Thestrup Hansen et al., 2021; Thome, 1998; Thorne, 2016). The explicit intention to inform and support practice through contextualized findings is a unique feature of interpretive description (Thorne, 2016) and well suited to support the research aims of this study.

This methodology is particularly well-suited to analyze focus group data since subjective perspectives are interpreted within themes and common characteristics across participants, while concurrently accounting for individual variation to promote the inclusion and equality of all individual realities (Burdine et al., 2021). The guidance towards contextualization and practical implications rather than abstraction or theorization is additionally beneficial for reducing the risk of interpretation bias, which is particularly relevant during secondary analyses (Thorne, 1997). Furthermore, interpretive description supports PSWs as "situated knowers" who can meaningfully contribute to identifying career- and context-specific learning needs. Thus, interpretive description is well suited to understand the dementia-specific continuous learning needs of PSWs in long-term care homes, by centralizing their subjective experiences working with persons living with dementia and presenting results with clear implications for their continuing education.

3.3 Research Methods

3.3.1 Ethics

Ethical approval was granted by the University of Western Ontario Research Ethics Board for the primary study. Ethical approval was re-approved for the secondary qualitative analysis by the University of Western Ontario Research Ethics Board on August 16th, 2023 (see Appendix A). Furthermore, ethical concerns regarding the use of secondary qualitative analysis were addressed. These concerns are primarily focused on issues of informed consent, confidentiality, and anonymity (Ruggiano & Perry, 2019; Thome, 1998). In the original study, all participants consented to the use of their data in additional studies. The data was not anonymized for analysis, as observing focus groups through video recordings was crucial for the analysis of group conversations to consider the tone and behaviours of participants towards each other and the social environments (Krueger, 1998). However, confidentiality was maintained through limiting data access to the first author (GN).

3.3.2 Sampling and Recruitment

This study uses and reexamines qualitative data collected from a Be EPIC research project conducted in 2019 and 2020. The principal investigator of the primary study recruited participants through email and phone calls to care organizations describing the intention of the study, eligibility, and contact information. Care organization managers then shared this information with their PSW staff. A list of interested participants were communicated back to the principal investigator (MYS). Participants were screened to determine eligibility. Details regarding the Be EPIC intervention and how data was generated are described in the following sections.

3.3.3 Be EPIC

Be EPIC is an innovative, evidence-informed and theoretically grounded person-centered communication intervention designed specifically for PSWs who care for persons living with dementia. The program uses simulated persons living with dementia via professional actors to allow participants an opportunity to practice the knowledge and skills introduced in the program. Participants observed each other interacting with simulated persons living with dementia which promoted learning through observation (Savundranayagam et al., 2021). After each simulation, peer feedback and debriefing sessions were held to promote learning through self-reflection, peer problem-solving, and further opportunities to provide feedback (Savundranayagam, et al., 2021).

3.3.4 Data Collection

In the primary study, qualitative data collection was conducted through one-hour focus groups with PSWs (N = 39) from four long-term care homes across London, Ontario between September 2019 to March 2020. Data collection was terminated in March 2020

due to the state of emergency ordered by the Ontario Premier on March 17th (Long et al., 2023). This was the first quarantine and lockdown in Ontario, Canada (Long et al., 2023). As a result, it was not feasible to complete data collection for some groups. The data were originally collected for a study investigating the impact of Be EPIC and exploring the working conditions in long-term care homes that influence person-centered dementia care from 'mid-career' PSW perspectives (Savundranayagam, et al., 2021).

Focus groups were held before, during, and after the PSWs completed a dementiaspecific person-centered communication training intervention called Be EPIC; which focuses on assessing the [E]nvironment, using [P]erson-centered communication, developing client-centered relationships ([I] matter too), and integrating the [C]lient's experiences, abilities, and personality into care practices (Savundranayagam, et al., 2021). See Table 5 for a summary of focus group sessions according to group and training session. During the focus groups, semi-structured interview questions prompted participants to share and discuss their learning goals and reflect on what they learned from the program and the applicability of learned material to their clinical experiences. Discussions between PSWs also provided in-depth information about their perspectives regarding contextual work experiences, challenges in dementia care, and discrepancies between current competencies in dementia care and desired knowledge and skills and dementia care outcomes, identifying dementia-specific learning needs. Focus groups were audio and video recorded. Recordings were transcribed verbatim and stored in a secure file with the principal investigator (MYS).

Focus	Pre-Training			During Training		Post-Training		Total
Group	Baseline 1	Baseline 2	Module 1	Module 2	Module 3	Module 4	Three Month Follow- up	
B1	1	1	1	1	1	1		6
B2	1	1						2

Table 5: Focus group sessions

B3	+	1	1	1	1	1		5
A4			1	1	1	1	1	5
A5			1	1	1	1		4
Total	2	3	4	4	4	4	1	22

+Audio recording of group B3's baseline 1 focus group was not captured in the video recording. Therefore, transcripts could not be produced.

3.3.5 Data Analysis and Interpretation

I conducted a secondary qualitative data analysis of focus group data collected between September 2019 - March 2020. Secondary analysis provides an opportunity for the investigator to use previously collected data to explore new questions and expand on previous findings (Thorne, 1998; Ruggiano & Perry, 2019). Focus groups capitalize on group processes to generate social knowledge such as shared beliefs and attitudes motivating certain behaviours or thought processes (Thorne, 2016). The analysis of focus group data is distinctly different from individual interview data analysis, as participants tend to influence each other, changing opinions, attitudes, and building conversation based on previous comments or perspectives (Krueger, 1998). Analysis of focus group data also relies on being attentive to the tone of participants towards each other, the social atmosphere created, and group and individual behaviours (Krueger, 1998).

The secondary data analysis was conducted through an iterative, inductive approach, following the guidelines for Interpretive Description (Thorne, 2016) (explained below). The iterative process in Interpretive Description (Thorne, 2016) encourages the researcher to engage in data collection and analysis concurrently, with the analytic processes informing subsequent data collection and construction (Thorne, 2016). Due to the secondary analysis nature of this study, iterative practices were involved in the analysis and reasoning processes to organize and conceptualize the data (Slaughter et al., 2007). Data was analyzed chronologically and organized in meaningful ways to address the research question and have clear implications for PSW continuing education.

Thorne (2016) describes three broad stages of data analysis: familiarization and initial reflexivity, initial broad coding, and interrogation and refinement of codes. During the familiarization and reflexivity stages, each focus group transcript and video was reviewed in its entirety and reflexive writing was practiced throughout to discern initial thoughts, opinions, and pre-conceptions. I reviewed the data in the same order it was collected in (i.e., baseline data first, followed by module one, then module two, and so on) and recorded patterns and data that stood out to me for potential codes in a written document (audit trail). Reflexivity during this stage included asking myself why particular pieces of data or themes captured my attention over others. The initial patterns, codes and reflexive notes were then shared with MYS for further review and to discuss future considerations of the analytic approach, such as observing if certain learning needs changed over time.

Once all transcripts and videos had been reviewed, open coding was then performed in NVivo to synthesize meanings within the data by identifying and developing concepts related to dementia-specific learning needs. The same order of transcript and video review was maintained during open coding. The goal was to use coding as a mechanism for grouping together sections of data that could be thematically related to allow for a more detailed interrogation of evolving themes, potential relationships with other codes, and how each code contribute to understanding the dataset (Thorne, 2016). Codes were created when I encountered data that was important to identify for further interrogation. With this approach, codes and themes evolved as I moved through the dataset and my understanding of data became stronger and when patterns began to form. Further, once a new code was identified and created, I would iteratively review previous transcripts to ensure that data was not missed or revise codes to reflect new understandings of the data as it evolved.

Interrogation and refinement of codes involved axial and selective coding techniques (Thorne, 2016). Axial coding develops primary codes into broader, more conceptual categories (Charmaz & Thornberg, 2021). This involves using constant comparison methods to examine relationships between codes and understand their contribution to the

conceptualization of the dataset and address the research question (Thorne, 2016). This involved several interrogations with the initial coding to synthesize meanings, investigate relationships, and refine themes or codes (Charmaz & Thornberg, 2021). For instance, I interrogated a code titled 'lack of confidence', which was originally created to categorize all data encompassing when PSWs reported a lack of confidence in a particular dementia-specific activity. Through interrogation, I reviewed each excerpt assigned to this code and found that data related largely to insecurities in addressing responsive behaviours, and therefore made this a subcode to the 'addressing responsive behaviours' code. Selective coding was the final stage of data analysis, which encompassed interpretive reasoning with the research team to refine conceptualizations of the themes in relation to the broader social contexts and implications for PSW education. For example, excerpts coded within the 'learning methods' theme were investigated to determine if they were specific to Be EPIC or applicable to other training contexts.

3.4 Rigor

In keeping with methodological coherence, credibility, and rigour, the main investigator (GN) practiced reflexivity throughout the analysis process, performed audit trails, and validated findings with the principal investigator of the primary research study (MYS) (Ruggiano & Perry, 2019; Thorne et al., 1997). Reflexivity is essential in interpretive description as the researcher is assumed to influence the interpretation of the data, which must be acknowledged (Thorne, 2016; Thorne, 1997). Audit trailing is of equal importance as the quality of the interpretation resides in the ability to present how and why certain interpretive directions were taken and how they arrived at certain findings (Thorne, 2016; Thorne, 1997). Furthermore, ethical and representational challenges of qualitative secondary analysis were addressed through transparency regarding the primary study's data collection methods, decisions, and context, and collaborating with the principal investigator of the primary study (MYS) (Ruggiano & Perr, 2019; Thorne, 1997).

3.5 Positionality of the Researcher

This research may help contribute valuable knowledge about where PSWs need support in terms of continuing education and training opportunities in dementia care. As the main investigator, I show up to this work as both an 'insider' and an 'outsider', as I worked as a frontline support worker at the outset of this study but am now taking the role of a student and researcher. I worked for a year and three months as a support worker in a community home for older adults living with developmental and physical disabilities. My work was located in a small, rural area in Ontario during the COVID-19 pandemic (May 2021 – August 2022). My role consisted of all PSW responsibilities including aiding with basic and instrumental activities of daily living, mental and emotional support, palliative and hospice care, and household maintenance (e.g., laundry, cleaning, etc.). In addition, I was also responsible for managing and assisting people with personal and government finances, administering daily and 'pro re nata' (or 'taken as needed') medications, scheduling personal appointments, and advocating on behalf of individuals during medical appointments, community engagements, organizational policy changes, and legal consultations.

In my view, the inadequate educational and training support provided to employees during my time as a frontline caregiver was significant. Most of the employee training was rushed or signed off on before I could complete it, due to the staffing crisis at the home, my status as a part-time employee (despite often working overtime), and COVID-19 training priorities. My co-workers shared with me that they accumulated and attributed all of their knowledge to work experience in the homes rather than formal or continuing education opportunities, as they felt their frequency and relevancy to their roles were insufficient. As a result, we often felt lost when navigating individual behaviours derived from medical conditions, such as responsive behaviours or mood swings from individuals living with dementia. I experienced this firsthand when I inquired to my manager about American Sign Language (ASL) classes. In a house of four older adults requiring support, two were hard of hearing or deafness and relied exclusively on ASL to communicate. I continued to request and advocate for ASL classes

by the organization to better care for these individuals, but my request was always dismissed.

This research project has been inspired by working frontline during the pandemic, hearing my co-workers' perspectives on educational preparedness, and my own difficulties advocating for continuing education in this role. Thus, this thesis is situated within these contextual factors, that have shaped who I am as a researcher and the perspectives I bring to this analysis.

Chapter 4

4 Results

This study aimed to identify the dementia-specific learning needs of working PSWs in long-term care homes and examine how learning needs evolve throughout a dementia-specific training program (Be EPIC). The following section presents the demographic data of the participants, followed by a detailed description of three major themes that were discerned from the data: (1) demand for dementia-specific education, (2) learning methods, and (3) mediators. The evolution of learning needs and methods is described within both themes respectively.

4.1 Participants and Demographic Data

Thirty-nine 'mid-career' PSWs participated in the training program and focus group sessions. The average age was 48 years old and participants possessed an average of 16 years of work experience as a PSW. Of these participants, thirty-five identified as female (90%) and four as male (10%). Twenty-four participants self-identified as White (63%), Canadian citizens (61.5%), who spoke English as their first language (82%). The remaining participants self-identified as Black (13%) or Hispanic (13%), Asian (5%), and the remaining as Middle Eastern (6%) or Indigenous (6%).

The length of time participants worked in long-term care ranged from 2-41 years, with the average being 15 years. Thirty-six participants (92%) held permanent positions as PSWs in long-term care homes at the time of data collection. Participants worked an average of 38 hours, most commonly during a morning/day shift (i.e., n = 25, 64%). However, the number of hours worked/week varied from 8 to 75 hours. During these shifts, thirty-four participants (87%) provided care to more than ten residents. Twenty-eight participants (72%) reported spending an average of 15 minutes or less with each residents per shift.

Twenty-four participants attained a college degree (61.5%), eight attained a high school diploma or an equivalency (20%), five attained a university bachelor's degree (13%), and one attained less than a high school diploma (3%). Most participants (i.e., n = 30, 77%) received formal PSW training and thirty-five (90%) received dementia-specific continuing education including GPA, U-First, DementiAbility, in-service training, and others. See Table 6 for further details on the demographic data.

Variables	Ν	%				
Sex						
Female	35	90				
Male	4	10				
Gender*						
Female	34	87				
Male	4	10				
Age						
Mean (Range)	48 (29-65)	-				
Ethnicity*						
White	24	61.5				
Black	5	13				
Asian	2	5				
Hispanic	5	13				
Middle Eastern	1	3				
Aboriginal	1	3				
Immigration Status*						
Canadian citizen by birth	24	61.5				
Canadian citizen by naturalization	8	20.5				
Landed immigrant	5	13				
Years as PSW						
Mean (Range)	16 (2-41)	-				

Table 6: Participant demographics

Years in Long-Term Care		
Mean (Range)	16 (2-41)	-
Position*		
Permanent	36	92
Temporary	2	5
Shifts Worked*		
Morning/Day	25	64
Afternoon	4	10
Evening	6	15
Overnight	1	3
Hours/Week working in Long-Term Care		
Mean (Range)	38 (8-75)	-
Number of Residents Cared for/Shift*		
1-5	1	3
6-10	2	5
>10	34	87
Daily Length of Interaction with Residents*		
≤ 15 minutes	28	72
30-60 minutes	7	18
>60 minutes	3	8
Education*		
Less than High School	1	3
High School or Equivalent	8	20.5
College, CEGEP, other non-university	24	61.5
certificates		
University Bachelor's Degree	5	12.8
Formal PSW Training*		
Yes	30	77
No	7	18

Dementia Specific Training Experiences

GPA	35	87
U-First	15	38
DementiAbility	3	8
In-service Training	19	49
Other	3	8

*Numbers reflect participants who completed the demographic survey questions. Therefore, discrepancies in numbers exist due to participants leaving certain questions blank.

4.2 Demand for Dementia-Specific Education

The learning needs identified by PSWs were informed by their experiences in formal and continuing education. Learning needs were most commonly attributed to limited preparation during formal PSW education and a lack of continuous training opportunities provided to PSWs. As a result of these experiences, the following four learning needs were identified: (1) understanding dementia, (2) addressing responsive behaviours, (3) person-centered communication and attitudes, and (4) delirium. In this section, the educational context in which learning needs emerged is described first, followed by the learning needs themselves.

4.2.1 Formal and Continuing Education

Dementia-specific education was lacking throughout the PSWs' careers. Participants commented that their formal PSW education did not equip them with enough knowledge or skill to provide dementia care within the context of long-term care. In terms of formal education, two PSWs insinuated content on dementia was completely absent from their curriculum during program attendance by advocating for these courses to be added in:

"You should put [dementia care] in the actual PSW classes" (PSW5). "In the actual program itself like you know, you've got health and wellness, and this, and this, and this. There should be like a dementia class" (PSW2). "Like a course like you know

how they have other courses spread out. Maybe in like the PSW part, they can add that in there somewhere" (PSW5).

These judgements were further extended to current education curricula and practices based on their experiences working with placement students and recent graduates. Consequently, PSWs recognized a sense of inadequacy in confidence to provide optimal dementia care among both themselves and new PSWs entering the long-term care sector: "Really the newer people definitely...they're lacking confidence, as well with dealing with a lot of these situations, and I don't know what they're getting taught in school anymore, but I don't think it's dementia-specific..." (PSW5).

PSWs emphasized that in addition to adding more dementia-specific content, the delivery methods also need to be improved. Multiple focus groups identified the need for more experiential learning methods to be included in formal PSW education to improve confidence in dementia care: "It's only the new people that don't have that much experience...And they're just like...they are afraid to step in...They are not having hands-on [training]" (PSW32).

Certified PSWs working in long-term care relied on continuing education opportunities to address gaps in dementia-specific competencies. However, PSWs expressed dissatisfaction with the limited frequency of these opportunities. The lack of opportunities for PSWs was particularly troubling as they voiced concerns regarding equitable educational support compared to other industries as well as remaining current with care research:

"I took the PSW course 18 years ago. So, what changed from 18 years to present? There's been a lot of research, and obviously, a lot of stuff changed. So, like I was saying, in any industry, there's always educational training and upgrading, and I don't know if the focus for PSWs really has that strength on it." (PSW30).

PSWs emphasized that the opportunities they did receive fell short of addressing their learning needs, both in terms of delivery methods and content. These opportunities were most commonly online, asynchronous modules mandated to either address problems and/or meet regulation standards. While topics such as oxygen administration, infection control, and Workplace Hazardous Materials Information System (WHMIS) were covered, PSWs felt they did not receive enough continuing education opportunities tailored specifically to dementia care. As a result, they advocated strongly for more opportunities to be offered on a regular and ongoing basis:

"My problem with the training though, with the oxygen and stuff, it's all done in reaction. So, there is either a ministry order, there is a problem and then they'd go around and [mandate] everybody in the training and then it's done. That stuff should be done on a regular basis." (PSW26).

Another PSW commented:

"In two months down the road, we might completely forget this training, not forget, but we're not going to be implementing it. But if we have continued education, it will always be there. As PSWs in the workforce, we don't have any room for advancement. We are PSWs, we will never advance in this home, right? That is our job: PSW, but we need continuous training. There is no training at all in dementia training, teamwork training, or morale training. All-around training needs to be done continuously here." (PSW29).

A discrepancy in knowledge based on experience levels existed in some homes as a consequence of inadequate continuing education opportunities for certified PSWs in long-term care. One PSW emphasized the importance of equal education opportunities, regardless of experience level, to ensure consistency in dementia care practices. One suggestion was to offer continuing education courses aligned with updated learning outcomes in PSW programs:

"And for the PSWs, like for example, you've been doing it for a long time, versus a person who's been doing it for two years who just finished school two years ago and learned about the responsive behaviours and stuff so then when a PSW like that comes in and then works maybe with somebody who has been doing it for a longer time who seems a little bit shorter in that kind of thing, maybe refresher courses...or maybe they

should be rolled out to everybody everywhere you know...so that everybody is educated and people aren't just feeling like 'oh my God she's coming in and she's going to tell me what to do, she's only been doing it for a year or whatever it is." (PSW24)

4.2.2 Learning Needs

4.2.2.1 Understanding Dementia

PSWs identified a general need for more education to understand dementia more precisely. Most PSWs identified this as a broad learning need without specifying exactly what they wanted to understand about dementia. For instance, one PSW stated their desired learning outcome from the training program was to: "Understand the disease more, to be able to be more hands-on with the clients I'm working with" (PSW29). Others elaborated on this by distinguishing a need to understand the different types of dementia: "And also, I would like to learn if there are certain things that differentiate or are common between different types of dementias" (PSW26).

4.2.2.2 Addressing Responsive Behaviours

Addressing responsive behaviours was the most frequently discussed learning need across all focus groups. Responsive behaviours can be understood as how an individual living with dementia might communicate unmet needs and feelings in response to their personal, social or physical environments and contexts (Clifford & Doody, 2018). Participants described these behaviours as 'violent' and 'aggressive' and emphasized a need for additional education on how to appropriately address them:

"I would like to see more tools that we can implement working with people with heavy dementia. And when they are a little bit on the side of aggression, a gentle approach yes, but we would like to get more training with it." (PSW32)

The number of staff who are trained to appropriately address responsive behaviours was not sufficient for the frequency of responsive behaviours in their long-term care homes: "As a staff member, I would like to feel more confident to approach this situation right. Like I know we have [gestures to PSWs in the group] as [staff with additional behaviour-specific training], but sometimes when we are waiting for them to come, even just to prevent anything bad from happening right, so some tools maybe. Like that's why I signed up for education." (PSW33)

PSWs did not feel confident in appropriately addressing responsive behaviours when caring for people living with dementia. This included preventing behaviours before they happen and while someone is experiencing responsive behaviours. When engaging in the experiential-based learning with simulated persons living with dementia in Be EPIC, PSWs noted that despite having previous experiences in similar situations, they still felt uncertain, uncomfortable, and afraid in their approach. One participant described feeling their 'fight or flight' response activate when they engaged with a simulated person living with dementia displaying responsive behaviours:

"I was ready to be thrown [by the simulated person living with dementia] ...and I'm not saying that I had to jump out of my comfort zone...but like you said, [it was a] fight or flight [response for me]...I want more of a confidence to deal with aggressive residents, like I mean the ones with high-end dementia but [they're] more aggressive...getting the confidence up there...if it does happen again or to stop it before it happens again." (PSW23)

The lack of confidence was also extended to newly graduated PSWs, noting that many appeared unprepared when addressing responsive behaviours. Many of the 'mid-career' PSWs within the focus groups expressed concern regarding the adequacy of current training programs to address responsive behaviours appropriately and confidently. As a consequence, new PSWs were described as feeling overwhelmed and potentially abandoning a person living with dementia. One PSW emphasized the necessity of a comprehensive training program to equip PSWs with the skills, knowledge, and confidence required to address responsive behaviours:

"In order for the industry to somehow survive without any great tragedies taking place they need a two-day annual training and several hours... Because I don't even know if they're being taught that and they don't seem to have...new PSWs don't seem to have any idea of what is in front of them...they just don't know how to handle [responsive behaviours] and many of them would have just walked away from [the simulation] and left [the simulated person living with dementia] to his own demise." (PSW7)

Responsive behaviours can be expressed through verbal/vocal and physical behaviours (Cohen-Mansfield, 2000). Verbal/vocal behaviours can include complaining, yelling or screaming, cursing, repetitive sentences, questions or unwarranted requests for attention or help, and verbal sexual advances, while physical behaviours can include pacing, restlessness, hoarding or hiding objects, scratching, grabbing, kicking, biting, hitting, and physical sexual advances, to name a few (Cohen-Mansfield, 2000). The approach used to address these behaviours will vary due to the differences in behaviour presentation and the potential for harm. However, distinguishing between verbal and physical responsive behaviours was difficult for some PSWs, as they often automatically labelled all responsive behaviours and, most importantly, the person expressing these behaviours, as 'aggressive', or 'violent'. This was particularly evident during a focus group immediately following the education module on addressing responsive behaviours with the simulated persons living with dementia. During the focus group, one PSW asked if the others had received "violence" from the actor, prompting a discussion about how to accurately classify behaviours. The facilitator as well as the PSWs involved in the simulation explained the importance of describing the behaviours accurately as it can influence the care approach used with that individual in the future:

"Because my impression of violence is that someone's got their hands and they're ready to smoke you, so if you come to me and say '[person living with dementia] is being violent.' I am going into that room expecting that he's going to have to be restrained or something. (Facilitator) Well, I said [they] were violent in the sense that I thought he was going to be banging his head, for example. He was agitated, he was frantic, he was scared, he was crying, he was panicked. He seemed panicked and I

think that if I got in his way, I would've gotten hit, but I don't think it was intentional to hurt me, it was just in the state of panic that he was in." (PSW28)

In another example, one of the PSWs involved in a simulation expressed uncertainty concerning how to address the behaviours appropriately. This sparked several PSWs to jump in and answer their question, however, each with different approaches, including calling a "code white" (violent/aggressive behaviour) and leaving the person exhibiting the behaviours. However, the facilitator reminded the group that the simulated person living with dementia was not displaying physically responsive behaviours, and therefore did not constitute a "code white" for violent/aggressive behaviour or being abandoned:

"But for the safety part, what do we do? They are trying to jump out, they are falling, knocking themselves so what do we do? (PSW19) You call a code white (PSW21). I would say take yourself out and bring someone else in." (PSW22). "[The simulated person living with dementia] wasn't being aggressive, like [they weren't] being physically responsive to you, she wasn't a code white, she was just escalating, she just wanted to get out." (Facilitator).

Involving mental health support within care practices for people living with dementia was becoming more common for PSWs, however, adequate education and training in this area was severely lacking:

"There are a lot of mental health problems coming into long-term care, so trying to deal with those behaviours as well...some more tools would be helpful." (PSW12). "We're struggling... definitely mental health is huge right now." (PSW16).

PSWs recognized trauma as having a pervasive impact on the prevalence and patterns of responsive behaviours amongst persons living with dementia in long-term care. Several PSWs shared instances where they uncovered a direct link between specific responsive behaviours and past traumatic events in their client's lives. Discovering and acknowledging the association was an important skill set for PSWs to have to prevent and address behaviours in person-centered ways. However, PSWs were rarely provided information regarding potential traumatic experiences or education on how to address

such situations effectively and efficiently. This left many in vulnerable positions where they needed to develop their own strategies and approaches to uncover such information to better meet the needs of individuals in their care. PSWs noted that this process often required a trial-and-error approach due to a lack of education on trauma-informed care:

"We had a gentleman. This gentleman, he was afraid. Every time we put him in the shower, he was very aggressive. Then we found out that his brother drowned, that's why he was like that. He was scared of drowning. So, if we know that, we try a different way... we just try, like you know, go maybe with a little wet towel. Just go around his body and like you know in the shower chair without putting the hose on him...put [him] in the tub is better, or sometime not because it depends on history, yeah." (PSW18).

4.2.2.3 Person-centered Communication and Attitudes

PSWs identified a need for further educational support in effectively approaching and communicating with persons living with dementia in person-centered ways. Person-centered communication is a key aspect of person-centered care, which seeks to foster meaningful relationships between persons living with dementia and their peers (Kitwood, 1997; Savundranayagam & Moore-Nielsen, 2015). This approach involves the recognition and inclusion of the life history, preferences, and values of the person living with dementia during social and care interactions (Kitwood, 1997; Savundranayagam & Moore-Nielsen, 2015). Learning needs within this domain include person-centered communication strategies and stigmatizing language, as well as attitudes that influence dementia care approaches. Attitudes included those held by PSWs toward their approach to providing person-centered care and their roles in providing care for people living with dementia.

A general need for further education on person-centered communication was identified by PSWs to optimize the quality of care provided to persons living with dementia in long-term care. Acknowledging and understanding clients as unique individuals through person-centered communication strategies was important. One PSW identified this as a personal learning need during a pre-training focus group session:

"More time and more background knowledge of who they are as a person before the dementia. Like I find that we don't really know their wife's name, their kids' names, their grandkids' names, what they did, where they lived, where they're from. Just to like, have more interaction with them [...] so just would like to know more about them." (PSW30).

Communicating with persons living with dementia in person-centered ways was difficult for most PSWs among all five focus groups. PSWs shared their gap in knowledge and skills in effectively communicating with persons living with dementia often provoked distress during care interactions for both them and their clients. PSWs experienced frustrations and moral distress when navigating repetitive conversations with persons living with dementia. This was particularly distressing when a person would repeatedly ask about topics that had the potential to prompt feelings of grief, such as inquiring about relatives who have passed away; especially if the PSW did not feel knowledgeable on how to respond appropriately.

"One of the questions that [persons living with dementia] ask is, for example, 'where is my husband? Why isn't he coming?' And the husband is gone for a long time, and they forget about that, [...] is there any way that we can do something or we tell them something so that they don't have to go through the grief every day or every hour or every two hours of losing their loved ones? That bothers me as a worker that is working day in and out." (PSW6).

Another PSW shared that not knowing how to navigate repetitive conversations was a particularly frustrating experience when it prevented them from providing care on time:

"So, you feel guilty, and you have less patience for the residents because they are holding you up. This person makes you repeat everything five times, and they don't understand what you're saying, you have to walk away because it is so frustrating." (PSW11). Improving the efficiency of care interactions through person-centered communication was another motivator for further education in this area:

"You know, maybe if we're learning how to communicate with people better, then we'll be able to get it done quicker. Sometimes arguing with somebody and forcing them ends up taking a lot longer and if we get better skills to communicate, then we'll be able to be more efficient at our other jobs." (PSW36).

Complicating this knowledge gap is the addition of communication impairments among persons living with dementia. Without adequate knowledge or skills, PSWs found it difficult to comprehend or use alternative communication strategies in person-centered ways. This led some PSWs to perceive that communication was not possible once an impairment was present:

"We often make the choices for [people living with dementia] which does take away from them but are they able to always answer...like on our floor, not very often." (PSW2).

Another PSW described how they may speak at or over persons living with dementia instead of with them, causing frustrations and potentially responsive behaviours:

"Many people go and talking, talking, talking, doing, doing, doing, and that may upset, may irritate. Many residents get aggressive because of us not... (PSW18) Yeah, they can't follow that conversation, like over their head. (PSW21) They don't cooperate with you when you [do that]...because they can sense sometimes, you know, your voice, [that] it's not about them." (PSW18).

Certain communication impairments resulted in the loss of the ability of people living with dementia to speak English. This was identified as a common problem amongst PSWs in one focus group in particular. This difficulty caused frustration for both persons living with dementia and PSWs. "But then again, you get to know the residents who like to hear you talk, even though they aren't answering you. Some, it angers them because maybe it's frustrating, they can't answer you, no idea. 'You don't even speak my language, right?' That's a big difficulty, not speaking the language." (PSW2).

The use of stigmatizing language toward persons living with dementia was used amongst nearly all PSWs from each long-term care home. The language used to describe persons living with dementia included labels such as 'aggressive', 'dementia or Alzheimer's resident', 'feeders', and 'demented people'. PSWs from one long-term care home referred to people who did not live with dementia as 'cognitive people' in addition to 'dementia residents'. Behaviours amongst persons living with dementia were described as 'unpredictable', 'demanding', 'resistive to care', and 'dangerous'. Furthermore, PSWs described themselves as 'dealing with' persons living with dementia rather than 'caring for' or 'working with' them. For instance, one PSW expressed their learning goal while using stigmatizing language:

"I want more of a confidence to deal with aggressive residents, like I mean the ones that have high-end dementia but more aggressive. I have dealt with a lot of aggressive residents while I have been here, I have been a victim to a lot of aggressive residents here." (PSW23).

The use of this language indicated a knowledge gap related to identifying and using stigmatizing language in all focus groups. However, PSWs only recognized this knowledge gap after being exposed to the topic during the training program:

"I went to school 12 years ago and when I went to school we were taught well they are an aggressive resident, we weren't taught so much about different behaviours and responsive behaviours whereas now, things are 'responsive behaviours' but I wasn't taught that way in school so my whole career leading up to this point is that is an aggressive behaviour, so [the training program]'s changing the whole mindset on how do you deal and implement supports for these residents." (PSW29). PSWs became aware of the consequences of using stigmatizing language. For example, labelling someone as "aggressive" can impact the mindset and approach used to provide care for a person living with dementia:

"Maybe a few weeks ago, I would think, oh I know this is an aggressive resident when they are not aggressive, right? So, if you can change your mindset on the aggression part, like, you said, and realize that it's the dementia or the delirium and how we're interacting with that resident differently from now on compared to how we were say a week ago or two weeks ago." (PSW29).

Person-centered attitudes and approaches for providing care to people living with dementia were identified as a learning need as well. Particularly, approaches and attitudes promoting patience amongst PSWs and independence amongst persons living with dementia were the two most commonly identified learning needs in this topic. Approaching people living with dementia with attitudes centered around promoting independence was identified as a learning need immediately before PSWs were introduced to the training program: "I think I'd like to get more tools to continue to grow to support independence for my residents" (PSW30). This was also extended to approaching people living with dementia with attitudes similar to restorative care and how people living with dementia are perceived by some PSWs as only 'getting worse':

"I kind of think that maybe some of the mindset is that when residents come in here, they're not going to get any better, they are only going to get worse. But that's not how it is [...] I would like to change that mindset." (PSW24). "Yeah, the mindset for sure. Hopefully, we get those tools to carry it forward with the rest of the team." (PSW30).

The need for further education on approaches to promote independence and the attitudes' influencing behaviour was also exemplified through descriptions provided by PSWs on their approaches at the time of the training. For example, one PSW sought advice on how to approach a person living with dementia who does not want them to brush their teeth for them: "I have the blanket on top of him, he is in the bed and I am pushing his arms, so
he doesn't get his arm off because he is going to throw the toothbrush away" (PSW20). This comment indicates that the PSW lacks the communication skills and attitudes to provide care in a manner that aligns with person-centeredness through the promotion of dignity, independence, and respect for the person living with dementia, rather than using physically restrictive strategies. Furthermore, PSWs described often making decisions on behalf of persons living with dementia due to inadequate communication skills and time limits imposed by the conditions of long-term care homes:

"We often make the choices for them [...] It's kind of hard even to give them a choice on what to wear. That decision seems a little difficult for them or we don't give them enough time to make it." (PSW4).

Several PSWs referred to providing care for persons living with dementia as a "task". One PSW described how this mindset influences their verbal and non-verbal communication with persons living with dementia and their approach to providing person-centered care:

"That's one of the things [persons living with dementia] will say is that we probably don't talk enough, and we don't smile enough, but it's because we're on task, right...I don't want to refer to them as a job, but they are a job, a task at that moment, it just so happens that the task involved a human being." (PSW7).

4.2.2.4 Delirium

Delirium was identified as a learning need after PSWs were introduced to the topic in the training program. Familiarity with the topic was varied as some recognized the term while others did not. However, upon reflection, PSWs recognized that delirium education was needed before it was recognized as a learning need, as a handful of PSWs admitted they previously believed delirium and dementia were the same: "A lot of the time we label [delirium] automatically as 'oh they have dementia' but there's a big difference between having dementia and having delirium" (PSW29).

After learning about delirium, PSWs identified the topic as urgently needed in the PSW curriculum and continuing education programs. This included learning about the different types of delirium, causes, symptoms, and approaches to address them in PSW practice:

"How come this isn't in the [PSW] curriculum [...] this isn't in any of the curriculum for any of the courses like college or universities really. I took a little bit of it but the hypo and hyperactive [delirium], they weren't [included]...honestly, I didn't even know the hypo and hyper existed, all I heard was 'there is delirium, there is dementia, here's an umbrella [term], these are the symptoms of what can cause this, this, and this. Your exam is next week.' Where with [Be EPIC], you branched right out, you said it can be caused by this, this, also could be caused by this, these are the signs...then you did it in a scenario with the signs showing it, I didn't learn any of that [in PSW curriculum]." (PSW23).

4.2.3 Evolution of Learning Needs

Figure 2 outlines the evolution of learning needs as PSWs were introduced to concepts and educational content during the training program. Before PSWs were introduced to the program, self-perceived learning needs included understanding dementia, addressing responsive behaviours, and person-centered communication strategies and approaches in dementia care. The content introduced to PSWs during Be EPIC expanded their previous learning needs to include a wider range of topics and subtopics. For instance, identifying and using stigmatizing language, distinguishing between verbal/vocal and physical responsive behaviours, and delirium were identified as learning needs after PSWs learned about them.



Figure 2: Identification of learning needs before, during, and after participating in Be EPIC

4.3 Learning Methods

How PSWs wanted to learn was also an important factor in the identification of learning needs. Before being exposed to the training program, PSWs emphasized a need for training and education methods to include opportunities to learn from their co-workers and receive constructive feedback and evaluation within a supportive learning environment. Once engaged in the training program, PSWs also highlighted a need for training/education to be conducted through experiential methods. PSWs commented on the lack of experiential methods in formal and continuing education opportunities, as discussed under heading 4.3.1. See Figure 3 for a timeline on the identification of learning methods before, during, and after participating in Be EPIC.



Figure 3: Identification of learning methods before, during, and after participating in Be EPIC

PSWs expressed a need for continuing education to reflect the collaborative nature of PSW work within long-term care homes by incorporating opportunities to learn from their peers through conversational and observational methods. The preference for peer learning was present before participants engaged in the training program, however, the importance of this element in future education programs was reinforced by the effectiveness of the debriefing opportunities in the training program. During the training program, participants were provided with opportunities to observe each other participate with a simulated person living with dementia. Afterwards, the participants would reflect on their experiences and provide feedback to each other. Particularly, the uniqueness of individual experiences, and therefore diversity across a team/group of PSWs, was highlighted as an important resource for learning:

"When people come in from other units as well, like someone who's part-time or there for one time, they can bring knowledge with them as well. Sometimes they'll say, 'Oh, we have a person on our unit, and we try this.' And so sometimes that's a really good tool for us to use is the knowledge that everybody brings to the table." (PSW36).

Learning while observing offered PSWs the opportunity to learn by watching their coworkers. This method added to the learning being done through conversations by modelling various care approaches with the simulated persons living with dementia: "Yeah, I think it was a great exercise because you can see how another person is acting and how I act too." (PSW20). Other PSWs commented:

"I like that there was three of us [in the simulation] and two could observe so then I could know what is going on after. (PSW21) So, you could see how someone else does a thing too not just thinking [...]" (PSW22).

Further, PSWs identified that having their peers share their observations about each other following the simulations generated a greater sense of self-awareness. This was particularly helpful for identifying unconscious behaviours and relating the educational material to their self-concepts and experiences. The significance of receiving feedback emerged during a baseline focus group and was subsequently amplified as PSWs actively participated in providing and receiving feedback from their peers, simulated persons living with dementia, and training facilitators throughout Be EPIC. Peer feedback was valued by participants because it provided an opportunity to share knowledge, experiences, and skills, offer reassurance, and build confidence and self-awareness during care interactions:

"I think it would be nice to see if there was training like this in the workforce [...] for example, giving us feedback as PSWs and what [we] are doing wrong [something that could be improved] and right. Because I think there are a lot of staff who might do something they are not aware of, whereas doing this will make them aware that 'Hey I have done that every day in my whole ten-year career.' Right where [I am] now, someone has made me aware of it, where I can stop and [tell myself] 'don't do that'." (PSW29).

The feedback provided by the simulated persons living with dementia was viewed as crucial for PSWs' learning process and professional growth because it was from the unique perspective of an individual receiving dementia care: "And you can change things too after and they'll tell you like 'Hey you tugged on my shoulder a little too heavy,' or

something. You can work on that. It will always stick in the back of your head" (PSW25). Another PSW shared:

"[Receiving feedback from the actors] was huge. That was invaluable because they can give us a different perspective than co-workers [...] They can tell us how they felt. They can tell us whether or not they felt safe, whether they felt mistreated, or you know, respected. They can tell us those things better because they are the ones who feel it." (PSW26).

This feedback was especially important for PSWs, as their clients living with dementia often conveyed difficulties communicating feedback verbally.

"On the job, we are not ever going to get that feedback from someone who actually has dementia. So, that's the thing, they have got into their acting roles so well, so to get that feedback was good. Because on the job, we'll never [know]." (PSW29).

PSWs also recognized that for feedback to truly foster professional growth and learning, it needed to be delivered constructively and positively: "And [the facilitators] are always encouraging to learn and like make mistakes and learn from your mistakes and like what could've been done better and 'you did this great'" (PSW27).

Across PSW accounts, the positive and constructive nature of feedback was seen to contribute to building a supportive peer learning environment. Participants described feeling safe within the learning environment when it fostered collaboration, trust, and mutual respect. Two key areas stuck out to the participants in this area: (1) providing and receiving peer feedback as a group and (2) supportive teamwork dynamics.

The ability to provide and receive constructive feedback as a group was described as creating a psychologically safe learning environment for all PSWs. As a result, trust could be built and maintained for future training sessions or opportunities for peer feedback:

"I think the debriefs, even in the smaller groups, allow us to rethink things, and give tips. I think one of the things I've noticed though is that it has built a lot of trust amongst us, being able to trust that I can say something and I'm not going to offend anybody and trust that it's coming from... (PSW26). "An honest place." (PSW24). "A safe place." (PSW25).

One PSW mentioned the presence of the supportive learning environment facilitated a greater ability to engage and focus on learning within the simulations:

"And when I'm doing the scenarios, it doesn't bother me because if I screw up, I'm going to learn and if I do it right, I'm going to learn and not be offended...the feedback I've gotten has been constructive feedback [...] everybody has been okay with the feedback from each other which is not what I was expecting." (PSW24).

The assurance of support from co-workers during interactions with simulated persons living with dementia was another element of the learning environment that fostered a feeling of safety and trust. This was particularly important as some PSWs perceived this level of teamwork was missing in real-world care interactions: "It was nice to know that when you needed the tap out, the tap out was like welcoming not just like 'oh you can't handle it?" (PSW27) "There was a positive energy in this room rather than on the floor" (PSW28).

4.3.1 Experiential Learning

Learning through experience and practice was a highly effective learning method for PSWs, a consensus observed across all focus groups. Simulations during the training program involved practicing and applying new concepts with the simulated persons living with dementia. PSWs recognized both the practicality of these simulations and the handson approach as crucial for successfully applying knowledge, skills, and attitudes to realworld care interactions. The practicality of the simulations facilitated learning retention by accurately modelling the PSW role within long-term care. The hands-on approach involved both physically and verbally interacting with the simulated persons living with dementia, allowing PSWs to apply knowledge obtained from the training program and

gain further knowledge through experience within the simulation itself: "I like the handson because that's what we do. It puts us in our comfort zone, it puts us in our world and then our instincts and our training kind of kicks in and you go through it" (PSW26). Another PSW expressed:

"There was a practical component added in as opposed to just like theory, so we were able to apply what you were showing to us. So, it gave us an opportunity to see if we were able to or if we had to expand and grow." (PSW30).

The hands-on approach to simulation-based training with simulated persons living with dementia was also beneficial for learning as it reduced the possibility of distractions outside of the learning environment:

"And I find with [online training], I will sit and watch a video for two hours and I am trying to do it and all of a sudden, I am like I don't remember. I've got kids screaming in the background, I have my husband watching TV, yelling at the TV, and at the end of it I am like 'I have no idea what I just watched'. Then I have to watch that whole two-hour video again. My eight-hour training took me four days." (PSW29).

Furthermore, when compared to previous learning methods, such as simulation-based learning with mannequins in formal education and asynchronous online methods in continuing education, the utilization of simulated persons living with dementia was perceived as the most effective method for learning. PSWs described this method to be particularly realistic to their daily interactions with persons living with dementia:

"I went to [name of school], we actually had like the simulation with the...mannequins...but that is nothing compared to [simulated persons living with dementia]. Like I feel like the simulation [in school] should involve something like this, you know what I mean? Because we actually have the simulation rooms, and it is set up as like an actual hospital room. Like there is actual oxygen, you check their blood pressure, you do everything to these mannequins. But when you have the actual person that can get up and actually do, they should implement that." (PSW25). Another PSW commented:

"Every home has their mandatory education stuff and a program like this being built into it where it's not just online, but where you're actually able to get in and do more in-service stuff...But I think the benefit to your residents, the benefit to your workers and the value you've put in your work when you do [this type of training], as opposed to going home, watching a bunch of videos, taking a couple of tests, have your computer crash all the time, get frustrated." (PSW20).

4.4 Mediators

Openness to dementia education and the presence of a good teamwork culture were identified as key mediators of learning being successfully transferred into long-term care practices. PSWs commented on the fact that for education to have any real meaningful impact on a PSW and the people they care for, they must be open to dementia education. This openness was described by PSWs as being associated with the individual's passion for providing optimal dementia care:

"You can give them free education all you want, but it's not even that if you're just giving education away and somebody just goes and does it, but if your heart isn't [in] it like this job is not for everybody." (PSW7).

Furthermore, throughout the focus groups, a good teamwork culture was identified as an essential meditator to successfully transferring learning into clinical contexts and behaviours. PSWs described their roles as working largely in pairs or teams to provide care to people living with dementia. A good teamwork culture in long-term care homes was described as involving the presence of mutual respect, trust, and openness to learn from each other, and to facilitate an exchange of knowledge. For instance, PSWs who received training needed to inform and support their peers who did not have the same opportunity to facilitate the exchange of new knowledge and promote changes in clinical behaviours. This was described as an essential component of applying new knowledge, skills, and attitudes to real-world care interactions:

"If we don't have something that bridges us using the skills and sharing them with other [staff]...to use it on a daily basis, it is going to be like anything else we have ever been taught: it's going to slide under the table. Not because it's not a good program, not because we don't want to [apply new knowledge], but because there aren't opportunities for us to share it. The more we end up having to teach this, as opposed to just modelling it...Like, talk about certain things and different approaches and whatnot where [staff] are willing to listen. And I think as long as we are able to do that over the period of time, it'll stick. But that has to be a buy-in by co-workers." (PSW26).

However, this was described as an issue within some teamwork dynamics when an openness to learn from peers was not present. PSWs mentioned this problem with more experienced PSWs discrediting the knowledge of those who are relatively new to the role:

"We do have some newer, younger PSWs here that some of the more senior PSWs are going to discredit just because they're young, you're new, you don't know, I've been doing this for a lifetime. And we need to switch that mentality and have something collaborative from management and staff to be able to bring this [...] if we are expected to just learn this information, go on the floor, and model it, it's going to die and then we've wasted our time." (PSW26).

Others shared concerns that their co-workers might be close-minded and dogmatic about learning something new if their current knowledge and experiences have worked for them in the past:

"It depends on the person again that you're speaking with. Again, with [name of PSW], I 100% feel like I can say, 'Hey, da da da da, this is what I find is working with this resident,' and they'll take that information. But some other PSWs: 'Oh no no no we just do this. Oh no we just do this'. It just depends." (PSW24).

In addition to being open to learning from peers, team members needed to be open to giving and receiving constructive feedback on work performance. This was a learning strategy used in the Be EPIC and was recognized by PSWs as an essential skill for

transferring knowledge from training into practice within the long-term care setting. However, many noted that this openness was missing in many long-term care homes, along with the knowledge and skills to provide feedback constructively:

"I feel like some staff members take [feedback] really personally. If you say, 'Hey why don't you try doing this?' they're like 'Okay well you're going to tell me how to do my job' kind of thing. There's a lot of that. Or 'Don't worry I worked here for thirty years, I know what I'm doing' there's a lot of that so..." (PSW38).

Chapter 5

5 Discussion

The findings from this study provide insight into the dementia-specific learning needs of PSWs working in long-term care homes, the desired methods for delivering such education, and the significance of ongoing education opportunities. This is the first study to identify and examine dementia-specific learning needs from the exclusive perspectives and experiences of PSWs working in long-term care homes. Furthermore, this is one of few dementia-specific learning need assessments to acknowledge PSWs as healthcare professionals with a rich composite of experience to guide interests and needs in dementia care education.

5.1 Demand for Dementia-Specific Education

The findings of this study contribute to and expand upon the existing literature on dementia-specific learning needs of unregulated healthcare workers. PSWs identified several gaps in competencies concerning dementia care including understanding dementia, addressing responsive behaviours, person-centered communication and attitudes, and delirium. Similar topics have been previously identified in other dementiaspecific learning need assessments (Attard et al., 2020; Bolt et al., 2020; Chang et al., 2009; Foster et al., 2019; Hughes et al., 2008; Kolanowski et al., 2015; Marx et al., 2014; Morgan, Kosteniuk, O'Connell, Bello-Haas, et al., 2016; Sabbe et al., 2023; Timmons et al., 2021), despite only a handful reporting those specific to unregulated healthcare workers. In the Canadian literature, rural home care PSWs also identified understanding dementia, assessing cognitive and functional abilities, communicating with persons living with dementia and family members, and addressing responsive behaviours as significant areas needing effective educational support (Morgan, Kosteniuk, O'Connell, Bello-Haas, et al., 2016). Outside of the Canadian context, addressing responsive behaviours, personcentered communication and attitudes, delirium, and understanding dementia have been consistently identified areas of knowledge and skill requiring additional support by unregulated healthcare workers (Attard et al., 2020; Bolt et al., 2020; Chang et al., 2009;

Foster et al., 2019; Hughes et al., 2008; Kolanowski et al., 2015; Marx et al., 2014; Sabbe et al., 2023; Timmons et al., 2021; Toteh Osakwe et al., 2022).

However, the present study is the first to identify trauma-informed care and dementiaspecific stigmatizing language. The novelty of these findings can be due to differences in data collection and analysis, and dementia-specific objectives. Quantitative data collection approaches, such as surveys and questionnaires, are the most commonly used methods for identifying learning needs, especially within the dementia-specific literature (Al-Ismail et al., 2023; Ferreira & Abbad, 2013; Grant, 2002; Hauer & Quill, 2011; Norris et al., 2024). However, these methods involve providing a predetermined list of competencies and/or topics for participants to choose from, potentially limiting their ability to express needs. Studies that employed a qualitative approach have narrowly focused on an aspect of dementia care rather than broadly exploring the topic, such as end-of-life care (Foster et al., 2019), self-management (Verkaik et al., 2017), and care plans (Toteh Osakwe et al., 2022). This narrow focus may limit the identification of dementia-specific learning needs that are beyond the objective's scope, yet within the realm of dementia care (Norris et al., 2024). Moreover, this study is the first to examine the identification of learning needs throughout a dementia-specific program, which led to the identification of the following learning needs: dementia-specific stigmatizing language and delirium (See Figure 2).

The wide range of learning needs identified by PSWs within this study align with existing research associating competency and confidence in dementia care with formal qualifications, dementia-specific training, and knowledge levels (Chen et al., 2013; Chung & Lai, 2003; Hughes et al., 2008; Morgan et al., 2016; Nguyen et al., 2022; Wang et al., 2020; Zimmerman et al., 2005). PSWs receive the least comprehensive formal education and fewest continuing education opportunities, compared to regulated healthcare professionals, which contributes to the perception of their profession being of 'low-skill' with minimal competencies and confidence in dementia care provision (Grant, 2016; Savundranayagam et al., 2021). Limitations in education exist despite PSWs representing the largest workforce in Canadian long-term care homes and providing over

80% of daily direct care to persons living with dementia (Chamberlain et al., 2019; Chen et al., 2018; Kane et al., 2023; Morgan, Kosteniuk, O'Connell, Bello-Haas, et al., 2016; Statistics Canada, 2004; Timmons et al., 2021). The range and quantity of learning needs identified in this study reflect a discrepancy in the competencies outlined in formal education standards and the reality of dementia care within the context of long-term care homes (Kary et al., 2020; Kelly, 2017; Kelly & Bourgeault, 2015; Scales, 2022). PSWs in this study identified the lack of dementia-specific education was evident within themselves as well as newly graduated PSWs, suggesting minimal improvements in dementia-specific curriculum within formal PSW education (Kane et al., 2023; Kelly, 2017; Ontario Centres for Learning, Research & Innovation in Long-Term Care, 2021; Scales, 2022).

Inadequate formal and continuing education significantly contribute to adverse outcomes on the well-being of PSWs and persons living with dementia, such as increasing the risk of burnout and dissatisfaction in the PSW role, challenges in retention, and issues providing person-centered care (Rivett et al., 2019). In the current study, PSWs working in long-term care homes did not feel adequately prepared with the competencies and confidence to provide dementia care through their formal and continuing education programs. As a result, PSWs reported feeling scared, uncertain, and uncomfortable during care interactions and during the training program's simulations, despite possessing an average of 16 years of working experience in long-term care. The findings of this study are consistent with previous literature reporting that PSWs experience feelings of fear, unpreparedness, and distress during care interactions with persons living with dementia due to insufficient education (Holst & Skär, 2017; Rivett et al., 2019). As a consequence, the risk of burnout and feeling dissatisfied in the PSW role increases, contributing to high turnover rates and issues in recruitment to long-term care sectors (Ministry of Long-Term Care, 2020; Savundranayagam & Lee, 2017). Staff who are stressed, dissatisfied, burnt out, and underprepared may experience a reduction in the capacity to empathize with persons living with dementia and are at an increased risk of providing depersonalized and task-oriented care (Rivett et al., 2019; Savundranayagam et al., 2021; Scott et al., 2011). This can lead to viewing persons living with dementia as tasks or objects rather than

unique individuals deserving of respectful, dignifying, and person-centered care (Holst & Skär, 2017; T. Kitwood, 1997; Scott et al., 2011), and is often reflected in depersonalized and technical care approaches (Baines & Armstrong, 2018; Leiter & Maslach, 1988; Scott et al., 2011). In the current study, PSWs' described challenges in providing person-centered care due to inadequate dementia-specific teachings in formal education and opportunities to engage in ongoing continuing education. This was discerned from PSWs' acknowledgement of dementia-specific learning needs as well as their descriptions of task-oriented attitudes and approaches towards care, situations where they limited autonomous choices for persons living with dementia and the use of stigmatizing language. In light of these challenges, it is evident that PSWs' well-being and the quality of care provided to persons living with dementia is intricately intertwined with the quality and effectiveness of PSW education to adequately prepare them for their roles within the long-term care sector.

Formal and continuing education for healthcare professionals has been internationally recognized as crucial for delivering high-quality care by competent, satisfied, and healthy healthcare workers (Public Health Agency of Canada, 2021; Kary et al., 2020; Savundranayagam et al., 2021). Studies consistently demonstrate that higher confidence and competency levels are linked to reductions in work-related stress, higher role satisfaction and retention rates (Li et al., 2021; Scales, 2022; Zimmerman et al., 2005). As a result, frontline staff are more confident in providing optimal care, increasing positive care outcomes and person-centered care (Kane et al., 2023; Li et al., 2021; Morgan, Kosteniuk, O'Connell, Bello-Haas, et al., 2016; Zimmerman et al., 2005). The most effective strategy in fostering feelings of preparedness and improving competency and confidence is through education that is tailored to the needs and contexts of its intended audience (Institute of Medicine, 2010; Laxdal, 1982; Rasmussen et al., 2023).

Adult learning theory specifies that the content of education programs must be immediately applicable to the learners' contexts or necessary to effectively cope with real-life situations, for learners to be motivated and ready to learn (Knowles, 1990). To do this, the learning needs to be directed by the learners themselves. Adults are self-

directing individuals responsible for their own decisions and have a deep psychological need to be recognized, acknowledged, and treated accordingly by others (Knowles, 1990). The most critical element of understanding competencies and developing an effective education model is the individual learners' perception of the discrepancies between their current abilities and the level at which they want (or need) to achieve (Knowles, 1990). This self-concept of being self-directed and independent in adults is intimately connected with unique and personal experiences accumulated over an individual's lifetime. Experiences, according to Knowles (1990), are the greatest resource for learning as they encourage a need to learn and can be drawn upon to help foster applicability and relevancy (Knowles, 1990). The assumption implies that if experiences are ignored or devalued, adult learners will associate this with a threat to their selfconcept and ability to be self-directing, increasing the risk of becoming closed-off, apathetic, and uncommitted to learning (Knowles, 1990; Merriam & Bierema, 2013). This emphasizes the importance of recognizing PSWs as experienced and knowledgeable professionals and providing opportunities to contribute to the development and improvement of their education. However, PSWs have historically been overlooked in their capacity to contribute in these ways (Norris et al., 2024). To enhance the effectiveness of dementia-specific education, PSWs must be acknowledged as individuals capable of identifying learning needs based on their perceived competencies and experiences caring for people living with dementia, and programs must be tailored accordingly (Kary et al., 2020; Rasmussen et al., 2023).

5.1.1 Understanding Dementia

Understanding dementia was identified by PSWs in the current study as a broad learning need encompassing an overall need for foundational dementia education. Understanding dementia has been reported as a learning need by unregulated healthcare workers across care settings and countries (Attard et al., 2020; Morgan et al., 2016; Timmons et al., 2021). Often encompassed within this need are understanding and recognizing signs and symptoms of dementia, progression, and the different types of dementias (Attard et al., 2020; Morgan et al., 2016; Timmons et al., 2021). This learning need is reflective of a gap in the quality and quantity of dementia-specific education afforded to PSWs.

Participants in the current study reported between two to forty-one years of experience as a PSW, indicating that their formal education adhered to the 2014 or earlier Ontario PSW Training Standards, which placed minimal focus on competencies related to dementia care and predominantly focused on task-based skills (Grant, 2016; Sethna, 2013). The standard outlines 14 vocational outcomes in which formal education institutions must incorporate into their curriculum, and students must reliably demonstrate to graduate with PSW certification (Ministry of Training, Colleges and Universities, 2022). One outcome pertains to competencies related to caring for individuals living with cognitive and/or intellectual impairments, mental health challenges, and responsive behaviours. Therefore, the findings underscore the importance of ensuring formal PSW standards accurately reflect the necessary competencies for delivering optimal dementia care in long-term care homes (Baines & Armstrong, 2018; Kosteniuk et al., 2016; Lightman, 2022).

PSWs must rely on continuing education programs when formal education is insufficient in establishing a strong foundation for dementia-specific competencies. PSWs in this research study advocated for equitable and ongoing opportunities for dementia-specific continuing education programs. Their accounts revealed that these opportunities are infrequent and often inadequate to maintain competencies and integrate new ones into practice throughout their professional career. Unregulated healthcare professionals have the fewest opportunities for dementia-specific continuing education compared to other healthcare professionals (Morgan, Kosteniuk, O'Connell, Bello-Haas, et al., 2016; Timmons et al., 2021), with some reporting no opportunities for dementia-specific education at all (Chen et al., 2018). PSWs in this study reported previous experiences in several dementia-specific continuing education programs, including Gentle Persuasive Approaches (GPA) (87%), U-First (38%), and DementiAbility (8%). As of 2022, GPA is endorsed in the 2022 Ontario PSW Training Standard for addressing responsive behaviours through best practices and person-centered approaches (Ministry of Training, Colleges and Universities, 2022). GPA's curriculum focuses on providing healthcare workers with the knowledge and skills to understand and address responsive behaviours including self-protection, gentle redirecting, communication strategies, and supportive approaches for persons at risk of delirium (Advanced Gerontological Education, n.d.).

Similarly, U-First and DementiAbility aim to provide education to healthcare professionals and care partners on person-centered perspectives and approaches to understanding dementia, the importance of physical environments, and addressing responsive behaviours (Alzheimer Society, n.d.; DementiAbility, n.d.). However, despite experiences in these programs, the current findings recognized that PSWs' learning needs overlapped with program objectives such as addressing responsive behaviours, personcentered communication, and delirium. One explanation for this could be that these programs target a large range of healthcare professionals and care partners, rather than being tailored specifically to PSWs. Currently, there's a lack of dementia-specific continuing education programs tailored to the needs of PSWs. Education programs based on the learning needs of the intended audience are the most successful in changing behaviours, practices, and competencies among healthcare professionals (Davis et al., 1995; Davis et al., 2008; Fox & Bennett, 1998; Grant, 2002; Institute of Medicine, 2010). The importance of education being oriented to the needs of the learner is further emphasized in adult learning theory, as Knowles (1990) claims that adults will "learn new knowledge, understandings, skills, and attitudes most effectively when they are presented in the context of application to real-life situations" (p.59).

The absence of continuing education opportunities became evident with the participants' progress in Be EPIC, which promoted new learning needs and methods. These included topics on dementia-specific stigmatizing language, differentiating between responsive behaviours, and delirium. Further, PSWs identified experiential learning methods as significantly effective once they were exposed to approaches such as simulated persons with dementia (See Figure 3). The addition of learning needs and methods with the training program indicates that PSWs became more aware of their deficiencies in the knowledge and skills that were introduced and the methods which yielded the most successful learning results. PSWs may not have had opportunities to learn about these topics or learn from experiential methods in ways that are meaningful to their roles and contexts. For instance, dementia-specific person-centered language is a relatively new area of education (Scott et al., 2011). PSWs in the current study described their lack of awareness and opportunities to learn about stigmatizing language before their

participation in Be EPIC. This has been cited in other studies (Clifford & Doody, 2018; Scott et al., 2011) which suggest that the use of stigmatizing language is a result of inadequate knowledge about person-centered care approaches.

Adult learning theory acknowledges that learners will not always be immediately aware of their learning needs (Knowles, 1990). However, an individual's awareness can be raised through exposure to new information and understanding the value learning something will bring to their lives (Knowles, 1990; Merriam & Bierema, 2013). Learners need to understand the value of learning something in the context of their own lives (Knowles, 1990; Merriam & Bierema, 2013), such as increased self-confidence in a specific skill, job satisfaction, or care outcomes. Raising an individual's awareness of the need to know something can be done through exposure to simulated experiences (e.g., role-playing with simulated persons with dementia) or relating new information to personal experiences (Knowles, 1990). This raised awareness can help individuals understand where their capabilities and deficiencies lie within a particular topic or ability, to more realistically gauge their learning needs (Knowles, 1990). Therefore, ongoing continuing education is essential to increase an individual's awareness of the need to learn something relevant to their lives (Knowles, 1990), highlighting the importance of increasing dementia-specific continuing education opportunities for PSWs (Rasmussen et al., 2023; Savundranayagam et al., 2021). Given that half (50%) of the PSW workforce in Ontario is between the ages of 35-54, continuing education is essential for addressing gaps left by formal education, re-skilling and maintaining competencies, changing attitudes and/or behaviours, and remaining knowledgeable in best practices (Kary et al., 2020; Ministry of Long-Term Care, 2020; Mlambo et al., 2021; Rasmussen et al., 2023).

5.1.2 Person-Centered Communication and Addressing Responsive Behaviours

In this study, PSWs revealed a need for further education on person-centered communication and attitudes. Without adequate knowledge or skills, PSWs in this study expressed difficulties communicating with persons living with dementia using alternative communication strategies, namely with persons who reverted to their mother tongue and

experienced challenges in formulating speech. Communication is an essential component of effective caregiving, especially in long-term care settings where PSWs interact with residents living with dementia and communication impairments (Savundranayagam & Lee, 2017). However, dementia can impact the ability of individuals to communicate effectively through expression and comprehension of language (American Psychiatric Association, 2022; Mundadan et al., 2023; Savundranayagam & Moore-Nielsen, 2015; Scott et al., 2011). Despite communication challenges, persons living with dementia remain relational beings, capable of engaging with their environments and the people within them and are not only willing to connect with others in meaningful ways but are deserving of dignified and respectful relationships (Clifford & Doody, 2018; Swinton et al., 2023). Maintaining these relationships is not only beneficial for the well-being of people living with dementia but is also key for maintaining and improving satisfaction among PSWs (Cohen et al., 2022; Savundranayagam & Lee, 2017). Furthermore, research suggests these relationships help to sustain a sense of identity and agency among persons living with dementia and facilitate successful care outcomes (Cohen et al., 2022; Gaviola et al., 2024; Medvene & Lann-Wolcott, 2010). This is particularly true for persons living with dementia who revert to their mother tongue, as providing care in an individual's own language is essential to prevent disengagement in meaningful activities and isolation (Gaviola et al., 2024). Thus, strong verbal and nonverbal communication skills are crucial for PSWs to possess to help deliver person-centered care and meet the needs of persons living with dementia (Foster et al., 2019).

Moreover, PSWs in the current study often assumed that persons with communication impairments were incapable of connecting and engaging with care staff, which has been a previously reported perception among PSWs (Savundranayagam et al., 2016). PSWs described instances where they limited personal autonomy and used inappropriate restraints during care interactions (e.g., holding a person with dementia down with their own body to brush their teeth) due to attitudes regarding the person with dementia's ability to communicate. A caregiver's perception of a person living with dementia shapes their behaviour during care (Cohen et al., 2022). Thus, perceptions of persons living with dementia being unable to engage with staff due to communication impairments increases

the risk of an individual receiving care that is task-oriented and focused on primarily meeting physical needs rather than considering the individuals' preferences, abilities, personality, or autonomy (Kitwood, 1997; Savundranayagam & Lee, 2017). Cohen and colleagues (2022) reported person-centered attitudes towards residents living with dementia were associated with more positive interactions, as nursing assistants were able to empathetically understand and interpret the meaning behind their behaviours and consider the uniqueness of each resident. Therefore, person-centered communication and attitudes in long-term care homes are essential skills that must be nurtured among PSWs to help develop and sustain relationships and promote person-centered care.

In the current study, PSWs described a significant lack of confidence in addressing responsive behaviours, both during care interactions with people living with dementia and the training programs' simulations. Through a person-centered lens, responsive behaviours are understood as alternative approaches to communication among persons living with dementia to convey emotional, physical, spiritual, or occupational needs, such as verbal expressions or physical behaviours (Cohen-Mansfield, 2000; Rivett et al., 2019). Responsive behaviours can be precipitated by feelings of confusion, fear, or frustration due an inability to recognize their caregivers, themselves, or their environment, being misunderstood, dependency on others, and waiting for care (Holst & Skär, 2021). Addressing responsive behaviours has been consistently identified as an area of dementia care requiring significant support, especially among unregulated healthcare workers such as PSWs (Hughes et al., 2008; Marx et al., 2014; Morgan et al., 2016; Timmons et al., 2021). Responsive behaviours have been associated with moral distress and burnout from fear, hopelessness, and negative impacts on professional self-esteem, as well as physical injuries, contributing to issues in retention and concerns regarding the quality of care provided to persons living with dementia (Holst & Skär, 2021; Rivett et al., 2019; Scott et al., 2010). PSWs in long-term care homes encounter responsive behaviours more frequently compared to other healthcare workers due to their prominent roles in providing daily personal care (Holst & Skär, 2021; Scott et al., 2010). Despite this, PSWs have the lowest levels of confidence and self-perceived competence in addressing responsive behaviours (Morgan et al., 2016). Without the knowledge, skills,

or attitudes to understand responsive behaviours as a form of communication and to identify precipitating factors, staff are unable to consider alternative or creative approaches to meeting their needs, most often resulting in non-person-centered strategies (Scott et al., 2011). This was observed among some of the PSWs in the current study, who described using improper physical restraints, abandoning the person displaying responsive behaviours, and using stigmatizing language.

The lack of sufficient education for addressing responsive behaviours through personcentered methods was also evident in the attitudes and approaches PSWs described towards persons living with dementia. How staff understand and view responsive behaviours can dictate how they will respond (Clifford & Doody, 2018). In this study, PSWs often automatically labelled persons living with dementia displaying responsive behaviour as 'violent', 'aggressive', 'dangerous', 'difficult', 'demanding', and 'unpredictable'. In previous studies, the use of these terms has been found to reflect beliefs that behaviours are deliberate and premeditated acts to cause intentional harm, rather than a form of communication regarding unmet needs (Holst & Skär, 2017; Scott et al., 2011). The use of stigmatizing labels can damage relationships between PSWs and persons living with dementia and increase the potential for caregivers to adopt a taskoriented care approach, reduce the capacity for empathy and increase the prevalence of dehumanizing behaviour (Rivett et al., 2019; Scott et al., 2010). When responsive behaviours are understood according to person-centered care, staff can respond by using strategies to uncover the meaning in the behaviour (i.e., the unmet need) or search for precipitating factors (Kitwood, 1997). Fortunately, some of the PSWs in the current study viewed responsive behaviours from a person-centered paradigm by describing instances when they found meaning within behaviours, rather than attributing them to dementia symptoms, such as previous traumatic events in an individual's life as well as communication impairments. Nonpharmacological interventions for addressing responsive behaviours are encouraged to maintain the individual's dignity and personhood and are the most effective methods to prevent/reduce responsive behaviours (Azermai et al., 2014; Clifford & Doody, 2018; Duxbury et al., 2013). Therefore, PSWs

must be provided with education regarding person-centered communication, attitudes, and responsive behaviour, to promote person-centered care.

5.1.3 Delirium

This research showed that delirium was recognized as a learning need once PSWs were exposed to the topic and its relevancy to their roles in caring for persons living with dementia in long-term care (see Figure 2). Sabbe and colleagues (2023) reported similar results among certified nursing assistants in Belgium, who provided delirium care based on intuition and pervious working experiences, highlighting an urgent need for formal delirium education. The current study is the first to identify delirium as a dementiaspecific learning need from the perspectives of PSWs working in Canada. Persons living with dementia are at an increased risk of experiencing delirium compared to persons without dementia (Holt et al., 2013). Delirium is an acute-onset syndrome triggered by acute illness, injury, or intoxication, emphasizing the necessity of its recognition as it can predict poor functionality and mortality, as well as cause significant distress (Kristensen et al., 2018; Oh et al., 2017). Behavioural symptoms of dementia and delirium have some overlap such as rapid onset and fluctuations of symptoms, impairments in attention, memory, disorientation, and disturbances to perception, thinking/speech, emotional regulation, and sleep (Dening, 2020). This overlap can create difficulties for care providers in identifying delirium among persons living with dementia, especially if they are provided limited education and training on both delirium and dementia. This was found in the current study, as participants were not aware of the need for delirium education until they were exposed to the topic in the training program. Failure to recognize delirium among persons living with dementia is an established issue within the nursing literature (FitzGerald et al., 2019; Lee & Roh, 2023). For instance, hypoactive delirium (characterized by low concentration, apathy, inactivity, and drowsiness) is reported to be misdiagnosed or unidentified at a rate of 33-66% (Dening, 2020). Moreover, recognizing delirium from dementia can reduce the potential for unnecessary antipsychotic administration (Dening, 2020; FitzGerald et al., 2018). Since PSWs spend the most time providing direct care to people living with dementia on a daily basis, they possess the most significant familiarity with the personalities, behaviours, and baseline

functioning of their clients compared to any other long-term care healthcare worker (Ministry of Long-Term Care, 2020; Sabbe et al., 2023). Therefore, they are optimally positioned to recognize delirium symptoms based on changes to behaviours, personalities, and baseline functioning. Proper education and training therefore must be provided to equip PSWs with the knowledge and confidence to identify and address delirium among persons living with dementia.

5.2 Learning Methods

In addition to the content of formal and continuing education not meeting the dementiaspecific learning needs of PSWs in long-term care, delivery methods can also be improved to ensure knowledge and skills are retained and, most importantly, applied to care practices. PSWs in this study advocated for education approaches to incorporate experiential methods that are conscientious of the context of application to real-life situations, peer learning and feedback, and a safe peer learning environment. A systematic review performed by Kane and colleagues (2023) found that interactive, experiential, and peer-based learning methods facilitated successful learning outcomes among dementia homecare workers. Similarly, other studies have revealed that combining traditional teaching techniques (e.g., classroom-style teaching and lecturing) with experiential methods (e.g., peer learning during care interactions) was most successful in enhancing confidence, competence, and application to dementia care practices (Rasmussen et al., 2023; Rivett et al., 2019).

According to adult learning theory, the uptake and application of knowledge, skills, and attitudes are the most effective when presented in the context of where they will be applied (Knowles, 1990), underscoring the importance of experiential methods being directly applicable and immediately relevant to the learners. In the current study, participants engaged in experiential methods with simulated persons living with dementia, which were described as a unique and useful opportunity to draw on individual experiences providing care to persons living with dementia while simultaneously applying their learning. Incorporating techniques that align with and tap into the learners' previous experiences and existing knowledge are key suggestions for improving the

efficiency and effectiveness of continuing education based on principles of adult learning theory (Knowles, 1990). Examples of additional experiential techniques include group discussions, role-playing, problem-solving activities, case methods, and peer-learning activities (Knowles, 1990).

Moreover, PSWs expressed a desire to learn from their peers through conversational and observational methods within a safe peer learning environment. These findings align with previous research indicating that peer learning can increase opportunities for knowledge exchange and emotional support, which have been associated with fostering personcentered attitudes, competencies, and knowledge (Kane et al., 2023; Su et al., 2021). Knowles (1990) emphasized the importance of a collaborative learning environment that supports interpersonal relationships and normalizes interactive participation opportunities. Incorporating and promoting peer learning in education programs tailored to PSWs is particularly important given the prevalence of their causal and part-time employment in long-term care homes (Baines & Armstrong, 2018; Boamah et al., 2023; Rossiter & Godderis, 2020). Inconsistent staffing is a barrier to providing person-centered care as it limits the opportunities to form positive relationships and understand the person living with dementia (Baines & Armstrong, 2018). Peer learning can facilitate knowledge exchange to help address knowledge gaps about persons living with dementia and ensure all PSWs, regardless of their employment status, have access to critical information and skills needed for providing person-centered care (Baines & Armstrong, 2018).

While the delivery methods identified in this study have been emphasized previously as essential components of continuing medical education for practicing physicians (Institute of Medicine, 2010), their importance has not had the same recognition for unregulated healthcare workers. For instance, many PSWs in this study identified that experiential methods were largely missing from both formal and continuing education opportunities. This oversight not only undermines the effectiveness of these methods for knowledge retention and application but also reflects a lack of recognition regarding the importance of adequately preparing PSWs to provide dementia care. Current continuing education is offered to PSWs through passive online modules, mainly concerned with meeting

compliance rather than focusing on improving care interactions (Bains & Armstrong, 2018; Savundranayagam et al., 2021). Learning methods that are meaningful and effective for PSW learning retention need to be integrated into formal and continuing education opportunities with the same importance as education opportunities afforded to other healthcare professionals, to maximize their effectiveness in knowledge application and uptake.

5.3 Mediators

For dementia-specific competencies to be integrated into care practices sustainably, learners need to be open to learning from educational programs and their peers (Rasmussen et al., 2023; Savundranayagam et al., 2020). In the current study, PSWs emphasized the importance of fostering a culture of teamwork within homes to promote openness to learning and sustain changes in behaviour and attitudes. Several other studies have acknowledged the crucial role of learners' openness in the success of dementiaspecific education programs (Auxier et al., 2020; Hung et al., 2022; Kary et al., 2020; Rasmussen et al., 2023). For example, Rasmussen and colleagues (2023) found that the participants' openness to learn and engage in education programs was equally as important as the program's design and delivery methods in facilitating successful learning outcomes. An openness to learning was demonstrated by a willingness to participate in education programs, receive new knowledge, and/or change behaviours (Rasmussen et al., 2023). Learning from coworkers was described by PSWs in this current study as an essential method of sustaining knowledge uptake and integration into dementia care practices within the long-term care setting. However, this was often described as an issue among more experienced PSWs being close-minded and dogmatic about learning due to the virtue of their titles (i.e., tenured vs. new) or the perceived effectiveness of their previous care approach. Experience can contribute to a close-mindedness for learning among PSWs with the development of habits, biases, and presuppositions (Knowles, 1990). If a PSW's prior knowledge and approaches have been effective in the past, or their biases and presuppositions have been reinforced, they can develop a closemindedness toward new ideas, perspectives, and ways of thinking or doing (Knowles,

1990). Since PSWs largely lack quality dementia-specific education and opportunities, their practices may not reflect best practices or person-centered care, as discerned in the current study's learning needs. Remaining close-minded about learning, whether through formal programs or peer interactions, could result in care practices that are unable to understand and address the needs of individuals living with dementia. Savundranayagam and Lee (2017) observed similar issues among PSWs who endorsed communication strategies that were not person-centered, such as ignoring the resident or completing care on their own. The use of these strategies was attributed to a lack of education regarding the abilities of persons living with dementia to meaningfully engage in care interactions as well as how to engage them as PSWs (Savundranayagam & Lee, 2017). To foster an openness to learning, organizations and continuing education providers must create environments that foster a continuous desire for learning and assist PSWs in understanding the need for additional education (Institute of Medicine, 2010; Merriam & Bierema, 2013).

5.4 Strengths and Limitations

The strengths of this study include the exclusive focus on the dementia-specific learning needs of PSWs from the voices of PSWs themselves. The representation of PSWs within the study is reflective of demographics in the Ontario workforce, regarding participants' age, sex and gender, education and employment experiences. The results of this study add to the literature advocating for the inclusion of PSW perspectives in future research and improved dementia-specific education. The use of focus group data capitalized on group processes to generate social knowledge such as shared beliefs and attitudes motivating certain behaviours or thought processes (Thorne, 2016). This allowed for the exploration of both unique perspectives from individuals as well as patterns across and within groups, based on social interactions and common experiences among PSWs relating to dementia-specific learning needs, methods, mediators, and education. Although there is a risk of focus groups obscuring non-dominant perspectives among participants (Thorne, 2016), a skilled facilitator was used to mitigate this risk by encouraging participants to address

each other in conversation and ensured equal opportunities for participants to contribute to focus groups.

The current study conducted a secondary qualitative interpretive approach to identify the dementia-specific learning needs of PSWs and examine how they evolve with education. Therefore, limitations exist regarding the interpretation of the data and inherent limitations to secondary analysis. While the discussions around the learning needs, methods, and mediators arose naturally within the focus groups, all but one of the semistructured focus group questions were not specifically tailored to the exploration of these findings (see Appendix B). Therefore, the questions developed for the primary research study were often prioritized over probing questions to further explore needs relative to the secondary analysis. It is important to acknowledge that qualitative data can often be shaped by social, cultural, and political contexts at the time of data collection (Ruggiano & Perry, 2019). These contexts can also change over time, resulting in a potential for being masked to contexts that may have been important to the interpretation and understanding of data, increasing the risk of misinterpreted results (Beck, 2018; Ruggiano & Perry, 2019). To mitigate the potential for misinterpretation of the secondary data analysis, I worked collaboratively with the primary investigator (MYS) of the original study and accessed video recordings of all focus groups. The primary investigator provided essential insight regarding the social and political contexts at the time of data collection, which aided my understanding of the subjective data. Moreover, observing the behaviours and dynamics of focus groups through video recordings enhanced my understanding of the context of conversations within the group setting.

5.5 Implications and Future Directions

The working and living conditions of long-term care homes have worsened since the time of data collection due to the COVID-19 pandemic, creating a significant urgency for addressing care provision issues through upskilling the PSW workforce. The COVID-19 pandemic was exceptionally difficult for both PSWs and persons living with dementia in long-term care homes (Gray et al., 2022; Hapsari et al., 2022; Kane et al., 2023). Long-term care homes were the most critically impacted locations for COVID-19 outbreaks

across Canada, representing over 80% of COVID-19-related deaths in 2020 (Lightman, 2022). The profound impact the pandemic had on long-term care homes was attributed to pre-existing deficiencies such as underfunding, overcrowding, inadequate infrastructure, staffing shortages, lack of oversight and accountability for upholding best practices, and recognition of staff (Boamah et al., 2023; Estabrooks et al., 2022; Rossiter & Godderis, 2020; Savundranayagam et al., 2021). As a result of the COVID-19 pandemic, the prevalence of burnout, anxiety, and depression increased significantly among both longterm care residents and staff, causing record-high rates of staff-related challenges, absences, and turnover (Boamah et al., 2023; Clarke, 2021). Vacancy rates for PSWs and nurses reached alarming levels of 8-12%, resulting in approximately 8,000 open positions in Ontario alone (Office of the Auditor General of Ontario, 2023). The compounded effects of poor working conditions and the COVID-19 pandemic compromised the quality of care provided to residents, 63% of whom were individuals living with dementia (Office of the Auditor General of Ontario, 2023). During the first wave of the pandemic in 2020, reports by the Canadian Armed Forces uncovered undignifying, dehumanizing, and unsafe care including the inability to meet the most basic needs (e.g., bathing and using the toilets), restricting resident independence, inadequate nutrition and hydration, inappropriate restraint usage (e.g., removing walking aides), and using degrading and inappropriate language toward residents (Mialkowski, 2020). In the wake of the pandemic's devastation, supporting the well-being of both PSWs and people living with dementia is more critical than ever. Providing PSWs with education that is meaningful to their experiences and roles, addresses their learning needs, and is offered on an ongoing basis can help improve the quality of care provided to people living with dementia (Kary et al., 2020; Rasmussen et al., 2023). Additionally, such education can enhance the wellbeing of PSWs by improving their competencies and self-efficacy, which can positively impact job satisfaction and retention in PSW roles (Boamah et al., 2023; Chamberlain et al., 2019; Hung et al., 2022; Kane et al., 2023; Zimmerman et al., 2005).

Many of the learning needs identified in this study, such as addressing responsive behaviours, person-centered communication, trauma-informed care, and communication impairments, should be included in the PSW Training Standard. The PSW Training Standard was updated in 2022 to include more person-centered language and dementiaspecific knowledge such as understanding dementia, delirium, and mental health care (Ministry of Colleges and Universities, 2022). For instance, vocational learning outcome 14, performance B, now includes three additional performance objectives relevant to caring for persons living with dementia: 1) observing and identifying changes in behaviour and advocating for revisions to the care plan with an interprofessional team; 2) describe how symptoms and/or presentation can impact clients' behaviour; and 3) identifying irreversible and reversible conditions (Ministry of Training, Colleges and Universities, 2022, p. 47). Indeed, this update reflects positive and timely changes to formal education to better equip PSW students with additional knowledge in personcentered dementia care, compared to the 2014 standard. Yet, it is unclear whether these changes will supply incoming PSWs with sufficient confidence and competencies to meet the needs of persons living with dementia in long-term care settings. Therefore, future research should investigate PSWs' perception of preparedness to provide dementia care and learning needs following education guided by the updated standards. Furthermore, this standard does little to improve the dementia-specific competencies of working PSWs, who represent the majority of the long-term care workforce (Ministry of Long-Term Care, 2020; Sethna, 2013). Therefore, the learning needs identified in the current study remain relevant for PSWs working in long-term care settings and must be addressed to improve dementia care.

This study has the potential to contribute to improving the quality of dementia-specific training programs for PSWs. The findings of this study outline the content (learning needs), delivery methods (learning methods), and mediators required by PSWs to address competency and confidence gaps in dementia care provision. The findings demonstrated that PSWs' dementia-specific learning needs extend well beyond the content and delivery methods offered to them through formal and continuing education programs. Future dementia-specific education programs targeting PSWs should integrate the identified learning needs (understanding dementia, addressing responsive behaviours, person-centered communication and attitudes, and delivery) and methods into the design and delivery, as well as consider the influential mediators.

This study also emphasizes the importance of including PSWs' experiences, perspectives, and needs in the development and improvement of educational programs. Due to the intimate nature of PSW work in long-term care homes, these professionals often spend the most time with persons living with dementia. Compared to other healthcare workers, this uniquely positions them with the opportunities to gather a significant amount of insight regarding daily dementia care capabilities and deficiencies in long-term care homes. Yet, the voices of PSWs and other unregulated healthcare workers have been largely unacknowledged, undervalued, and ignored (Lightman, 2022; Rossiter & Godderis, 2020). Going forward, the perspectives, knowledge, and needs of PSWs must be prioritized to continue to understand approaches to better supporting and promoting optimal dementia care practices in care settings.

5.6 Conclusion

As the incidence of dementia increases worldwide, the necessity to effectively equip PSWs with the competencies and confidence to provide optimal, person-centered dementia care becomes increasingly evident. The provision of dementia-specific training can go beyond improving the quality of care provided to people living with dementia, as it can also improve PSWs' well-being and satisfaction within their role. However, this study observed that current formal and continuing education opportunities do not effectively meet the needs of PSWs in long-term care concerning content and delivery methods that are relevant to their roles and promote sustainable learning and application. To ensure the delivery of optimal dementia care, education and training need to be tailored to the unique needs of PSWs working within long-term care homes.

The findings of this study demonstrate a gap in dementia-specific knowledge, skills, and attitudes among PSWs, due to a lack of education opportunities that adequately meet their needs. Education not only needs to teach topics relevant to the context and needs of PSWs in long-term care but also needs to be delivered with supportive and practical learning methods. Topics such as addressing responsive behaviours, using person-centered communication and attitudes, identifying and addressing delirium, and understanding dementia are highlighted as areas requiring greater emphasis in PSW

education. These topics should be delivered within supportive peer learning environments that incorporate experiential methods and opportunities for peer learning and feedback. Continuing education needs to be provided on an ongoing basis for knowledge, behaviour, and attitude changes to be sustained in busy and demanding settings, such as long-term care homes. The addition of learning needs and methods with the progression of Be EPIC suggests that opportunities for dementia-specific continuing education are rarely provided. These findings underscore the importance of providing ongoing, consistent education opportunities informed by the valuable perspectives, experiences, and voices of PSWs.

By highlighting the gaps in formal and continuing dementia-specific education for PSWs, the study findings may help to better inform the development and implementation of dementia-specific education for PSWs in the future. Furthermore, the findings highlight an ongoing and urgent need to acknowledge the value of PSWs' perspectives, opinions, and expertise related to caring for people living with dementia in long-term care settings.

References

- Adler, G., Lawrence, B. M., Ounpraseuth, S. T., & Asghar-Ali, A. A. (2015). A Survey on Dementia Training Needs Among Staff at Community-Based Outpatient Clinics. *Educational Gerontology*, 41(12), 903–915. https://doi.org/10.1080/03601277.2015.1071549
- Advanced Gerontological Education. (n.d.). *About GPA*. Advanced Gerontological Education. Retrieved May 11, 2024, from https://ageinc.ca/about-gpa-2/
- Afzal, A., Stolee, P., Heckman, G. A., Boscart, V. M., & Sanyal, C. (2018). The role of unregulated care providers in Canada—A scoping review. *International Journal* of Older People Nursing, 13(3), e12190. https://doi.org/10.1111/opn.12190
- Al-Ismail, M. S., Naseralallah, L. M., Hussain, T. A., Stewart, D., Alkhiyami, D., Abu Rasheed, H. M., Daud, A., Pallivalapila, A., & Nazar, Z. (2023). Learning needs assessments in continuing professional development: A scoping review. *Medical Teacher*, 45(2), 203–211. https://doi.org/10.1080/0142159X.2022.2126756
- Allen, M., Ferrier, S., Sargeant, J., Loney, E., Bethune, G., & Murphy, G. (2005).
 Alzheimer's Disease and Other Dementias: An Organizational Approach to Identifying and Addressing Practices and Learning Needs of Family Physicians. *Educational Gerontology*, 31(7), 521–539.
 https://doi.org/10.1080/03601270590962497
- Alzheimer Society. (n.d.). *About U-First!* U-First. Retrieved May 12, 2024, from https://u-first.ca/about-u-first/
- Alzheimer Society of Canada. (2014). PC P.E.A.R.L.S. 7 key elements of person-centred care of people with dementia in long-term care homes. Alzheimer Society of Canada. https://alzheimer.ca/sites/default/files/documents/PC-PEARLS_Full_Alzheimer-Society-Canada.pdf
- Alzheimer Society of Canada. (2022). *Navigating the path forward for dementia in Canada* (1; Landmark Study, pp. 1–58). Alzheimer Society of Canada. https://alzheimer.ca/sites/default/files/documents/Landmark-Study-Report-1-Path_Alzheimer-Society-Canada_0.pdf
- Alzheimer Society of Ontario. (n.d.). *Courses for health care providers*. ALZeducate. Retrieved June 6, 2024, from https://alzeducate.ca/course/index.php?categoryid=7
- Alzheimer's Disease International. (n.d.). *Dementia statistics*. Retrieved March 6, 2023, from https://www.alzint.org/about/dementia-facts-figures/dementia-statistics/
- American Psychiatric Association. (2022). Neurocognitive Disorders. In *Diagnostic and Statistical Manual of Mental Disorders* (5th ed.). American Psychiatric Association Publishing.

https://doi.org/10.1176/appi.books.9780890425787.x17_Neurocognitive_Disorde rs

- Annear, M., Nagasawa, F., Terawaki, K., Nagarekawa, F., Gao, X., & Otani, J. (2017). Educational needs of Japan's dementia care workforce: Results of a national online survey. *International Journal of Ageing and Later Life*, 11, 1–26. https://doi.org/10.3384/ijal.1652-8670.17340
- Attard, R., Sammut, R., & Scerri, A. (2020). Exploring the knowledge, attitudes and perceived learning needs of formal carers of people with dementia. *Nursing Older People*, 32(3). https://doi.org/10.7748/nop.2020.e1225
- Auxier, J. N., Roberts, S., Lauren Laing, Lee Finch, Sara Tung, & Lillian Hung. (2020). An appreciative inquiry into older adults' pain experience in long term care facilities: A pain education initiative: International Practice Development Journal. *International Practice Development Journal*, 10(1), 1–17. https://doi.org/10.19043/ipdj.101.006
- Azermai, M., Vander Stichele, R. R. H., Van Bortel, L. M., & Elseviers, M. M. (2014).
 Barriers to antipsychotic discontinuation in nursing homes: An exploratory study.
 Aging & Mental Health, 18(3), 346–353.
 https://doi.org/10.1080/13607863.2013.832732
- Baines, D., & Armstrong, P. (2018, February 7). Promising Practices in Long Term Care: Can Work Organisation Treat Both Residents and Providers with Dignity and Respect? https://www.semanticscholar.org/paper/Promising-Practices-in-Long-Term-Care%3A-Can-Work-and-Baines-Armstrong/99c1a3f1eb663d0c5ddfd7cb2280dd746af81ad9
- Beck, C. T. (2018). Secondary Qualitative Data Analysis in the Health and Social Sciences. Routledge. https://doi.org/10.4324/9781315098753
- Bing-Jonsson, P. C., Bjork, I. T., Hofoss, D., Kirkevoid, M., & Foss, C. (2013). Instruments Measuring Nursing Staff Competence in Community Health Care. *Home Health Care Management & Practice*, 25(6), 282–294. https://doi.org/10.1177/1084822313494784
- Block, S., & Galabuzi, G.-E. (2011). Canada's Colour Coded Labour Market: The gap for racialized workers. *Canadian Centre for Policy Alternatives*. https://policyalternatives.ca/publications/reports/canadas-colour-coded-labourmarket
- Boamah, S., Weldrick, R., Havaei, F., Irshad, A., & Amy Hutchinson. (2023). Experiences of healthcare workers in long-term care during COVID-19: A scoping review. *Journal of Applied Gerontology*, 42(5), 1118–1136. https://doi.org/10.1177/07334648221146252

- Bolt, S. R., Meijers, J. M. M., Steen, J. T., Schols, J. M. G. A., & Zwakhalen, S. M. G. (2020). Nursing Staff Needs in Providing Palliative Care for Persons With Dementia at Home or in Nursing Homes: A Survey. *Journal of Nursing Scholarship*, 52(2), 164–173. https://doi.org/10.1111/jnu.12542
- Breen, R., Savundranayagam, M. Y., Orange, J. B., & Kothari, A. (2022). Quality home care for persons living with dementia: Personal support workers' perspectives in Ontario, Canada. *Health & Social Care in the Community*, 30(5). https://doi.org/10.1111/hsc.13692
- Brooks, D., Gibson, B., & DeMatteo, D. (2008). Perspectives of personal support workers and ventilator-users on training needs. *Patient Education and Counseling*, 71(2), 244–250. https://doi.org/10.1016/j.pec.2008.01.018
- Burdine, J. T., Thorne, S., & Sandhu, G. (2021). Interpretive description: A flexible qualitative methodology for medical education research. *Medical Education*, 55(3), 336–343. https://doi.org/10.1111/medu.14380
- Campbell, J. (1990). Self-Esteem and Clarity of the Self-Concept. *Journal of Personality* and Social Psychology, 59(3), 538–549. https://doi.org/10.1037/0022-3514.59.3.538
- Carnahan, R. M., Daly, J. M., Minion, S., Gryzlak, B., Weckmann, M. T., Levy, B. T., & Bay, C. P. (2019). A needs assessment of family physicians to inform development of educational resources on antipsychotic use in dementia. *Journal* of Primary Care & Community Health, 10, 2150132719840113. https://doi.org/10.1177/2150132719840113
- Carpenter, C., & Suto, M. (2008). Chapter 2: Why choose qualitative research in rehabilitation? In C. Carpenter & M. Suto, *Qualitative research for occupational and physical therapists: A practical guide* (pp. 21–39). Blackwell Publishing.
- Chamberlain, S. A., Hoben, M., Squires, J. E., Cummings, G. G., Norton, P., & Estabrooks, C. A. (2019). Who Is (Still) Looking After Mom and Dad? Few Improvements in Care Aides' Quality-of-Work Life. *Canadian Journal on Aging / La Revue Canadienne Du Vieillissement*, 38(1), 35–50. https://doi.org/10.1017/S0714980818000338
- Chang, E., Daly, J., Johnson, A., Harrison, K., Easterbrook, S., Bidewell, J., Stewart, H., Noel, M., & Hancock, K. (2009). Challenges for professional care of advanced dementia. *International Journal of Nursing Practice*, 15(1), 41–47. https://doi.org/10.1111/j.1440-172X.2008.01723.x
- Charmaz, K., & Thornberg, R. (2021). The pursuit of quality in grounded theory. *Qualitative Research in Psychology*, *18*(3), 305–327. https://doi.org/10.1080/14780887.2020.1780357

- Chen, I., Lin, K., Hu, S. H., Chuang, Y., Long, C. O., Chang, C., & Liu, M. F. (2018). Palliative care for advanced dementia: Knowledge and attitudes of long-term care staff. *Journal of Clinical Nursing*, 27(3–4), 848–858. https://doi.org/10.1111/jocn.14132
- Chen, I.-H., Lin, K.-Y., Hu, S. H., Chuang, Y.-H., Long, C. O., Chang, C.-C., & Liu, M. F. (2018). Palliative care for advanced dementia: Knowledge and attitudes of long-term care staff. *Journal of Clinical Nursing*, 27(3–4), 848–858. https://doi.org/10.1111/jocn.14132
- Chung, J. C. C., & Lai, C. K. Y. (2003). Addressing the informational needs of healthcare professionals in dementia care: An explorative study in a Chinese society. *Aging* and Mental Health, 7(4), 287–293. https://doi.org/10.1080/1360786031000120688
- Clarke, J. (2021). Impacts of the COVID-19 pandemic in nursing and residential care facilities in Canada.
- Clifford, C., & Doody, O. (2018a). Exploring nursing staff views of responsive behaviours of people with dementia in long-stay facilities. *Journal of Psychiatric and Mental Health Nursing*, 25(1), 26–36. https://doi.org/10.1111/jpm.12436
- Cohen, L., Sher-Censor, E., Oppenheim, D., Dassa, A., Ayalon, L., & Palgi, Y. (2022). Nursing aides' mentalization, expressed emotion, and observed interaction with residents with dementia: A quasi-experimental study. *Geriatric Nursing*, 45, 100– 107. https://doi.org/10.1016/j.gerinurse.2022.03.002
- Cunningham, N., Cowie, J., Watchman, K., & Methven, K. (2020). Understanding the training and education needs of homecare workers supporting people with dementia and cancer: A systematic review of reviews. *Dementia*, 19(8), 2780– 2803. https://doi.org/10.1177/1471301219859781
- Daley, B. J., & Wilson, S. A. (1999). Needs assessment in long-term care facilities: Linking research and continuing education. *Journal of Continuing Education in the Health Professions*, 19(2), 111–121. https://doi.org/10.1002/chp.1340190207
- Davis, D. A., Thomson, M. A., Oxman, A. D., & Haynes, R. B. (1995). Changing physician performance: a systematic review of the effect of continuing medical education strategies. *JAMA*, 274(9), 700–705. https://doi.org/10.1001/jama.1995.03530090032018
- Davis, N., Davis, D., & Bloch, R. (2008). Continuing medical education: AMEE Education Guide No 35. *Medical Teacher*, *30*(7), 652–666. https://doi.org/10.1080/01421590802108323
- DeForge, R., & Shaw, J. (2012). Back- and fore-grounding ontology: Exploring the linkages between critical realism, pragmatism, and methodologies in health &
rehabilitation sciences. *Nursing Inquiry*, *19*(1), 83–95. https://doi.org/10.1111/j.1440-1800.2011.00550.x

- DementiAbility. (n.d.). *The DementiAbility methods*. Dementiability. Retrieved May 12, 2024, from https://dementiability.com/who-we-are/
- Dening, K. H. (2020). Differentiating between dementia, delirium and depression in older people. Nursing Standard (2014+), 35(1), 43–50. https://doi.org/10.7748/ns.2019.e11361
- Duxbury, J., Pulsford, D., Hadi, M., & Sykes, S. (2013). Staff and relatives' perspectives on the aggressive behaviour of older people with dementia in residential care: A qualitative study. *Journal of Psychiatric and Mental Health Nursing*, 20(9), 792– 800. https://doi.org/10.1111/jpm.12018
- Eggenberger, E., Heimerl, K., & Bennett, M. I. (2013). Communication skills training in dementia care: A systematic review of effectiveness, training content, and didactic methods in different care settings. *International Psychogeriatrics*, 25(3), 345–358. https://doi.org/10.1017/S1041610212001664
- Estabrooks, C., Song, Y., Anderson, R., Beeber, A., Berta, W., Chamberlain, S.,
 Cummings, G., Duan, Y., Hayduk, L., Hoben, M., Iaconi, A., Lanham, H., Perez,
 J., Wang, J., & Norton, P. (2022). The Influence of Context on Implementation
 and Improvement: Protocol for a Mixed Methods, Secondary Analyses Study. *JMIR Research Protocols*, *11*(9). https://doi.org/10.2196/40611
- Ferreira, R. R., & Abbad, G. (2013). Training needs assessment: Where we are and where we should go. BAR - Brazilian Administration Review, 10(1), 77–100.
- FitzGerald, J. M., Perera, G., Chang-Tave, A., Price, A., Rajkumar, A. P., Bhattarai, M., O'Brien, J. T., Ballard, C., Aarsland, D., Stewart, R., & Mueller, C. (2019). The Incidence of Recorded Delirium Episodes Before and After Dementia Diagnosis: Differences Between Dementia With Lewy Bodies and Alzheimer's Disease. *Journal of the American Medical Directors Association*, 20(5), 604–609. https://doi.org/10.1016/j.jamda.2018.09.021
- Foster, S., Balmer, D., Gott, M., Frey, R., Robinson, J., & Boyd, M. (2019). Patientcentred care training needs of health care assistants who provide care for people with dementia. *Health & Social Care in the Community*, 27(4), 917–925. https://doi.org/10.1111/hsc.12709
- Fox, R. D., & Bennett, N. L. (1998). Continuing medical education: Learning and change: implications for continuing medical education. *BMJ*, 316(7129), 466. https://doi.org/10.1136/bmj.316.7129.466
- Fukada, M. (2018). Nursing Competency: Definition, Structure and Development. *Yonago Acta Medica*, 61(1), 1–7.

- Gandesha, A., Souza, R., Chaplin, R., & Hood, C. (2012). Adequacy of training in dementia care for acute hospital staff. *Nursing Older People*, 24(4), 26–31.
- Gaviola, M. A., Omura, M., Inder, K. J., & Johnson, A. (2024). Caring for people with dementia from culturally and linguistically diverse backgrounds in nursing homes: A scoping review. *International Journal of Nursing Studies*, 151, 104674. https://doi.org/10.1016/j.ijnurstu.2023.104674
- Geriatric Certificate Program. (n.d.-a). *DementiAbility Methods*. Geriatric Certificate Program. Retrieved June 6, 2024, from https://www.geriatriccp.ca/courses/3
- Geriatric Certificate Program. (n.d.-b). *Gentle persuasive approaches (GPA) in dementia care*. Geriatric Certificate Program. Retrieved June 6, 2024, from https://www.geriatriccp.ca/courses/4
- Geriatric Certificate Program. (n.d.-c). *U-First! Workshop*. Geriatric Certificate Program. Retrieved June 6, 2024, from https://www.geriatriccp.ca/courses/2
- Gillham, D., De Bellis, A., Xiao, L., Willis, E., Harrington, A., Morey, W., & Jeffers, L. (2018). Using research evidence to inform staff learning needs in cross-cultural communication in aged care homes. *Nurse Education Today*, 63, 18–23. https://doi.org/10.1016/j.nedt.2018.01.007
- Government of Alberta. (2018). *Alberta health care aide competency profile*. Government of Alberta. https://open.alberta.ca/dataset/e1fbd562-1571-40b8bc17-71687dab444c/resource/9e34d246-c389-4753-bda4f53a77e02f80/download/ah-hca-core-competency-profile-2018.pdf
- Government of British Columbia. (2014). *B.C. health care assistants core competency profile*. Government of British Columbia. https://www.health.gov.bc.ca/library/publications/year/2014/HCA-Core-Competency-Profile_March2014.pdf
- Government of Canada. (2021). Nurse aides, orderlies and patient service associates. National Occupational Classification. https://noc.esdc.gc.ca/Structure/NocProfile?objectid=%2BlRFFq8NKNmTC5InH PJHzHGHFRL1nXdowfu204iMsNk%3D
- Government of Ontario. (2020). *Greater equity means greater student success*. http://www.ontario.ca/page/greater-equity-means-greater-student-success
- Government of Ontario. (2023). *Fixing long-term care act, 2021, S.O. 2021, c. 39, Sched. 1.* Government of Ontario. https://www.ontario.ca/laws/view
- Grant, J. (2002). Learning needs assessment: Assessing the need. *BMJ*, 324(7330), 156–159. https://doi.org/10.1136/bmj.324.7330.156

- Grant, K. A. (2016). The Design and Content of Personal Support Worker Training Programs [Thesis Dissertation, University of Toronto]. https://tspace.library.utoronto.ca/bitstream/1807/74838/3/Grant_Keira_A_201611 _MSc_thesis.pdf
- Gray, K. L., Birtles, H., Reichelt, K., & James, I. A. (2022). The experiences of care home staff during the COVID-19 pandemic: A systematic review. *Aging and Mental Health*, 26(10), 2080–2089. https://doi.org/10.1080/13607863.2021.2013433
- Green, C. (2020). Equity and diversity in nursing education. *Teaching and Learning in Nursing*, 15(4), 280–283. https://doi.org/10.1016/j.teln.2020.07.004
- Guba, E. G., & Lincoln, Y. S. (1994). Competing paradigms in qualitative research. In N.
 K. Denzin & Y. S. Lincoln, *Handbook of qualitative research* (1st ed., pp. 105–117). Sage.
- Gurnik, M., & Hollis-Sawyer, L. (2003). Empowering assisted living front-line care staffs to better care for alzheimer's and dementia residents. *Ageing International*, 28(1), 82–97. https://doi.org/10.1007/s12126-003-1017-5
- Hapsari, A. P., Ho, J. W., Meaney, C., Avery, L., Hassen, N., Jetha, A., Lay, A. M., Rotondi, M., Zuberi, D., & Pinto, A. (2022). The working conditions for personal support workers in the Greater Toronto Area during the COVID-19 pandemic: A mixed-methods study. *Canadian Journal of Public Health*, 113(6), 817–833. https://doi.org/10.17269/s41997-022-00643-7
- Hauer, J., & Quill, T. (2011). Educational Needs Assessment, Development of Learning Objectives, and Choosing a Teaching Approach. *Journal of Palliative Medicine*, 14(4), 503–508. https://doi.org/10.1089/jpm.2010.0232
- Hewko, S. J., Cooper, S. L., Huynh, H., Spiwek, T. L., Carleton, H. L., Reid, S., & Cummings, G. G. (2015). Invisible no more: A scoping review of the health care aide workforce literature. *BMC Nursing*, 14(1), 38. https://doi.org/10.1186/s12912-015-0090-x
- Hignett, S., Edmunds Otter, M., & Keen, C. (2016). Safety risks associated with physical interactions between patients and caregivers during treatment and care delivery in Home Care settings: A systematic review. *International Journal of Nursing Studies*, 59, 1–14. https://doi.org/10.1016/j.ijnurstu.2016.02.011
- Holst, A., & Skär, L. (2017). Formal caregivers' experiences of aggressive behaviour in older people living with dementia in nursing homes: A systematic review. *International Journal of Older People Nursing*, 12(4), e12158. https://doi.org/10.1111/opn.12158

- Holt, R., Young, J., & Heseltine, D. (2013). Effectiveness of a multi-component intervention to reduce delirium incidence in elderly care wards. *Age and Ageing*, 42(6), 721–727. https://doi.org/10.1093/ageing/aft120
- Hsiao, H.-Y., Liu, Z., Xu, L., Huang, Y., & Chi, I. (2016). Knowledge, Attitudes, and Clinical Practices for Patients With Dementia Among Mental Health Providers in China: City and Town Differences. *Gerontology & Geriatrics Education*, 37(4), 342–358. https://doi.org/10.1080/02701960.2014.990152
- Hughes, J., Bagley, H., Reilly, S., Burns, A., & Challis, D. (2008). Care staff working with people with dementia; Training, knowledge and confidence. *Dementia*, 7(2), 227–239. https://doi.org/10.1177/1471301208091159
- Hung, L., Yang, S. C., Guo, E., Sakamoto, M., Mann, J., Dunn, S., & Horne, N. (2022). Staff experience of a Canadian long-term care home during a COVID-19 outbreak: A qualitative study. *BMC Nursing*, 21(1), 45. https://doi.org/10.1186/s12912-022-00823-3
- Institute of Medicine. (2010). Continuing Professional Development: Building and Sustaining a Quality Workforce. In *Redesigning Continuing Education in the Health Professions*. National Academies Press (US). https://www.ncbi.nlm.nih.gov/books/NBK219809/
- Jensen, A. L., Handberg, C., Thomsen, T. G., Kjerholt, M., Soelver, L., & Hølge-Hazelton, B. (2018). Not for the World of Theorizing But for the Need of Knowledge in the Nursing Discipline: The Significance of Establishing a National Interpretive Description Network. *International Journal of Qualitative Methods*, 17(1), 1609406918776066. https://doi.org/10.1177/1609406918776066
- Kane, L., Leighton, C., Limbrick, H., Kilinc, S., Ling, J., & Eberhardt, J. (2023). You clapped, you cheered, but did anybody hear? A mixed-methods systematic review of dementia homecare workers' training and psychosocial needs. *Home Health Care Services Quarterly*, 42(4), 282–310. https://doi.org/10.1080/01621424.2023.2246415
- Kang, Y., Moyle, W., Cooke, M., & O'Dwyer, S. T. (2017). An educational programme to improve acute care nurses' knowledge, attitudes and family caregiver involvement in care of people with cognitive impairment. *Scandinavian Journal* of Caring Sciences, 31(3), 631–640. https://doi.org/10.1111/scs.12377
- Karlin, B. E., Young, D., & Dash, K. (2017). Empowering the dementia care workforce to manage behavioral symptoms of dementia: Development and training outcomes from the VOICE Dementia Care Program. *Gerontology and Geriatrics Education*, 38(4), 375–391. https://doi.org/10.1080/02701960.2016.1209419

- Kary, M., Lindsay, B., Stewart, J., & Hudson, M. (2020). Dementia-Friendly Care Homes: Best Practices in Dementia Care. *Kishore Seetharaman and Habib Chaudhury Department of Gerontology, Simon Fraser University*, 3–54.
- Kelly, C. (2017). Exploring experiences of Personal Support Worker education in Ontario, Canada. *Health & Social Care in the Community*, 25(4), 1430–1438. https://doi.org/10.1111/hsc.12443
- Kelly, C., & Bourgeault, I. L. (2015). Developing a Common Education Standard for Personal Support Workers in Ontario. *Health Reform Observer – Observatoire Des Réformes de Santé*, 3(1), Article 1. https://doi.org/10.13162/hro-ors.v3i1.169
- Kitwood, T. (1997). The experience of dementia. *Aging & Mental Health*, *1*(1), 13–22. https://doi.org/10.1080/13607869757344
- Kitwood, T. M. (1997). *Dementia reconsidered: The person comes first*. Open University Press.
- Knopp-Sihota, J., MacGregor, T., Nuspl, M., Reeves, J., & Kennedy, M. (2023).
 Healthcare aide-focused interventions to improve pain management in long-term care homes: A systematic review. *Archives of Gerontology and Geriatrics*, 104, 104808. https://doi.org/10.1016/j.archger.2022.104808
- Knowles, M. S. (1990). *The adult learner: A neglected species* (4th ed.). Gulf Publishing Company.
- Kolanowski, A., Van Haitsma, K., Penrod, J., Hill, N., & Yevchak, A. (2015). Wish we would have known that!" Communication breakdown impedes person-centered care. *Gerontologist*, 55, S50–S60. https://doi.org/10.1093/geront/gnv014
- Kosteniuk, J. G., Morgan, D. G., O'Connell, M. E., Dal Bello-Haas, V., & Stewart, N. J. (2016). Focus on dementia care: Continuing education preferences, challenges, and catalysts among rural home care providers. *Educational Gerontology*, 42(9), 608–620. https://doi.org/10.1080/03601277.2016.1205404
- Kosyluk, K. K., & Rigg, K. A. (2021). Developing and evaluating community-based health interventions: The role of data. In J. M. Choi & J. W. Murphy (Eds.), *Community-Based Service Delivery* (pp. 40–55). Routledge.
- Kristensen, F., Thuesen, J., & Håkonsen, S. J. (2018). Systematic assessment of the rehabilitation needs in early stage dementia for home-dwelling people: A scoping review protocol. JBI Database of Systematic Reviews & Implementation Reports, 16(10), 1947–1958. https://doi.org/10.11124/JBISRIR-2017-003469
- Krueger, R. (1998). Chapter 3: Analysis principles of particular importance to focus group research. In *Analyzing & Reporting Focus Group Results* (1–6, pp. 19–30). SAGE Publications, Inc. https://doi.org/10.4135/9781483328157

- Kusendova-Bashta, N., & Smith, L. (2023). Bill 121, Improving Dementia Care in Ontario Act, 2023. Legislative Assembly of Ontario. https://www.ola.org/en/legislative-business/bills/parliament-43/session-1/bill-121
- Lawler, K., Kitsos, A., Bindoff, A. D., Callisaya, M. L., Eccleston, C. E. A., & Doherty, K. V. (2021). Room for improvement: An online survey of allied health professionals' dementia knowledge. *Australasian Journal on Ageing*, 40(2), 195– 201. https://doi.org/10.1111/ajag.12886
- Laxdal, O. E. (1982). Needs assessment in continuing medical education: A practical guide. *Journal of Medical Education*, 57(11), 827–834.
- Leach, D. C. (2002). Competence Is a Habit. *JAMA*, 287(2), 243–244. https://doi.org/10.1001/jama.287.2.243
- Lee, G., & Roh, Y. S. (2023). Knowledge, barriers, and training needs of nurses working in delirium care. *Nursing in Critical Care*, *28*(5), 637–644. https://doi.org/10.1111/nicc.12724
- Leiter, M. P., & Maslach, C. (1988). The impact of interpersonal environment on burnout and organizational commitment. *Journal of Organizational Behavior*, 9(4), 297– 308. https://doi.org/10.1002/job.4030090402
- Leverton, M., Burton, A., Beresford-Dent, J., Rapaport, P., Manthorpe, J., Mansour, H., Guerra Ceballos, S., Downs, M., Samus, Q., Dow, B., Lord, K., & Cooper, C. (2021). "You can't just put somebody in a situation with no armour". An ethnographic exploration of the training and support needs of homecare workers caring for people living with dementia. *Dementia* (14713012), 20(8), 2982–3005. https://doi.org/10.1177/14713012211023676
- Li, X., Dorstyn, D., Mpofu, E., O`Neill, L., Li, Q., Zhang, C., & Ingman, S. (2021). Nursing assistants and resident satisfaction in long-term care: A systematic review. *Geriatric Nursing*, 42(6), 1323–1331. https://doi.org/10.1016/j.gerinurse.2021.08.006
- Lightman, N. (2022). Caring during the COVID-19 crisis: Intersectional exclusion of immigrant women health care aides in Canadian long-term care. *Health & Social Care in the Community*, *30*(4). https://doi.org/10.1111/hsc.13541
- Lockyer, J. (1998). Needs assessment: Lessons learned. *Journal of Continuing Education in the Health Professions*, *18*(3), 190–192. https://doi.org/10.1002/chp.1340180310
- Long, J. A., Malekzadeh, M., Klar, B., & Martin, G. (2023). Do regionally targeted lockdowns alter movement to non-lockdown regions? Evidence from Ontario, Canada. *Health & Place*, 79, 102668. https://doi.org/10.1016/j.healthplace.2021.102668

- Marx, K. A., Stanley, I. H., Van Haitsma, K., Moody, J., Alonzi, D., Hansen, B. R., & Gitlin, L. N. (2014). Knowing Versus Doing: Education and Training Needs of Staff in a Chronic Care Hospital Unit for Individuals With Dementia: [1]. *Journal* of Gerontological Nursing, 40(12), 26–34. https://doi.org/10.3928/00989134-20140905-01
- Mastel-Smith, B., Kimzey, M., & He, Z. (2019). Dementia Care Education for Nursing Students. *Journal of Nursing Education*, 58(3), 136–144. https://doi.org/10.3928/01484834-20190221-03
- Mayrhofer, A., Goodman, C., & Smeeton, N. (2016). The role of Dementia Champion in dementia care: Its aspirations, development and training needs (innovative practice). *Dementia*, 15(5), 1306–1312. https://doi.org/10.1177/1471301216631448
- McAiney, C. A., & Service, A. (2005). *Evaluation of the U-First! Learning initiative* (Evaluation Report 1; p. 22). https://brainxchange.ca/Public/Files/Ontario-Strategy/Init-1-Evaluation-of-the-U-First-Program.aspx
- Medvene, L. J., & Lann-Wolcott, H. (2010). An exploratory study of nurse aides' communication behaviours: Giving 'positive regard' as a strategy. *International Journal of Older People Nursing*, 5(1), 41–50. https://doi.org/10.1111/j.1748-3743.2009.00183.x
- Merriam, S. B., & Bierema, L. L. (2013). Adult Learning: Linking Theory and Practice. John Wiley & Sons, Incorporated. http://ebookcentral.proquest.com/lib/west/detail.action?docID=1376941
- Mialkowski, C. J. J. (2020). OP-LASER-JTFC Observations in long term care facilities in Ontario (OP-LASER-JTFC Observations in Long Term Care Facilities in Ontario 3350-Op Laser 20-02 [COS]; p. 15). Canadian Armed Forces. https://s3.documentcloud.org/documents/6928480/OP-LASER-JTFC-Observations-in-LTCF-in-On.pdf
- Ministry of Long-Term Care. (2020). *Long-Term Care Staffing Study*. Government of Ontario. https://www.ontario.ca/page/long-term-care-staffing-study
- Ministry of Training, Colleges and Universities. (2014). *Personal Support Worker Training Standard*. Government of Ontario.
- Ministry of Training, Colleges and Universities. (2022). *Personal Support Worker Training Standard*. https://files.ontario.ca/mcu-personal-support-worker-jan2022-41469-en-2022-03-31.pdf
- Mlambo, M., Silén, C., & McGrath, C. (2021). Lifelong learning and nurses' continuing professional development, a metasynthesis of the literature. *BMC Nursing*, 20(1), 62. https://doi.org/10.1186/s12912-021-00579-2

- Morgan, D. G., Kosteniuk, J. G., O'Connell, M. E., Bello-Haas, V. D., Stewart, N. J., & Karunanayake, C. (2016). Dementia-related work activities of home care nurses and aides: Frequency, perceived competence, and continuing education priorities. *Educational Gerontology*, 42(2), 120–135. https://doi.org/10.1080/03601277.2015.1083390
- Mukhalalati, B. A., & Taylor, A. (2019). Adult Learning Theories in Context: A Quick Guide for Healthcare Professional Educators. *Journal of Medical Education & Curricular Development*, 6, N.PAG-N.PAG. https://doi.org/10.1177/2382120519840332
- Mundadan, R. G., Savundranayagam, M. Y., Orange, J. B., & Murray, L. (2023).
 Language-Based Strategies that Support Person-Centered Communication in Formal Home Care Interactions with Persons Living with Dementia. *Journal of Applied Gerontology*, 42(4), 639–650.
 https://doi.org/10.1177/07334648221142852
- Nguyen, H., Eccleston, C. E., Doherty, K. V., Jang, S., & McInerney, F. (2022). Communication in dementia care: Experiences and needs of carers. *Dementia*, 21(4), 1381–1398. https://doi.org/10.1177/14713012221080003
- Norman, G. R., Shannon, S. I., & Marrin, M. L. (2004). The need for needs assessment in continuing medical education. *BMJ: British Medical Journal*, 328(7446), 999– 1001.
- Norris, G., Savundranayagam, M. Y., Vafaei, A., & Teachman, G. (2024). The representation of personal support workers in dementia-specific learning need assessments: A scoping review. *Educational Gerontology*, 0(0), 1–19. https://doi.org/10.1080/03601277.2024.2356925
- Nova Scotia Department of Health and Wellness. (2019). *Continuing care assistant* (*CCA*), *scope of practice & competency framework*. Nova Scotia Department of Health and Wellness. https://nsccaprod.wpenginepowered.com/wp-content/uploads/2021/10/CCA-SoP-and-Competency-Framework-June-2021.pdf
- Office of the Auditor General of Ontario. (2023). *Long-term care homes: Delivery of resident-centred care*. Ministry of Long-Term Care. https://www.auditor.on.ca/en/content/annualreports/arreports/en23/AR_LTCresid ential_en23.pdf
- Oh, E. S., Fong, T. G., Hshieh, T. T., & Inouye, S. K. (2017). Delirium in Older Persons: Advances in Diagnosis and Treatment. *JAMA*, *318*(12), 1161–1174. https://doi.org/10.1001/jama.2017.12067
- Ontario Centres for Learning, Research & Innovation in Long-Term Care. (2021). *PSW* perspectives on the staffing challenge in long-term care (p. 32). https://clriltc.ca/files/2021/02/PSW_Perspectives_FinalReport_Feb25_Accessible.pdf

- Ostaszkiewicz, J., Dunning, T., & Dickson-Swift, V. (2020). Translating dignity principles into practice for continence care for older people in care homes: A study protocol. *Journal of Advanced Nursing*, *76*(11), 3147–3154. https://doi.org/10.1111/jan.14481
- Peters, M. D. J., Godfrey, C., McInerney, P., Munn, Z., Tricco, A. C., & Khalil, H. (2020). Chapter 11: Scoping review. In *JBI Manual for Evidence Synthesis*. JBI. https://jbi-globalwiki.refined.site/download/attachments/355863557/JBI_Reviewers_Manual_2020 June.pdf?download=true
- Public Health Agency of Canada. (2021, April 20). A Dementia Strategy for Canada: Together We Aspire [Education and awareness]. https://www.canada.ca/en/publichealth/services/publications/diseases-conditions/dementia-strategy.html
- Raley, K. S. (2020). Discovering Blind Spots: Analyzing Teacher and Principal Perceptions of Principal Leadership Behaviors [Ed.D., University of Kansas]. https://www.proquest.com/docview/2429740224/abstract/5CBCC5521AF84265P Q/1
- Rasmussen, B. M., Andersen, P. T., Waldorff, F. B., & Berg-Beckhoff, G. (2023). Effectiveness of dementia education for professional care staff and factors influencing staff-related outcomes: An overview of systematic reviews. *International Journal of Nursing Studies*, 142. https://doi.org/10.1016/j.ijnurstu.2023.104469
- Rivett, E., Hammond, L., & West, J. (2019). What influences self-perceived competence and confidence in dementia care home staff? A systematic review. *Psychogeriatrics*, 19(5), 440–456. https://doi.org/10.1111/psyg.12422
- Rossiter, K., & Godderis, R. (2020). Essentially invisible: Risk and personal support workers in the time of COVID-19. *Sociology of Health & Illness*, 42(8). https://doi.org/10.1111/1467-9566.13203
- Ruggiano, N., & Perry, T. E. (2019). Conducting secondary analysis of qualitative data: Should we, can we, and how? *Qualitative Social Work*, *18*(1), 81–97. https://doi.org/10.1177/1473325017700701
- Sabbe, K., Aerts, N., van der Mast, R., & Rompaey, B. V. (2023). Certified Nursing Assistants' Perspectives on Delirium Care. *Journal of Gerontological Nursing*, 49(2), 43–51. https://doi.org/10.3928/00989134-20230106-07
- Savundranayagam, M. Y., Basque, S. R., & Johnson, K. (2020). Feasibility of a Dementia-Focused Person-Centered Communication Intervention for Home Care Workers. *Clinical Gerontologist*, 43(2), 181–192. https://doi.org/10.1080/07317115.2019.1694116

- Savundranayagam, M. Y., Docherty, -Skippen Susan Maureen, & Basque, S. R. (2021). Qualitative Insights into the Working Conditions of Personal Support Workers in Long-Term Care in the Context of a Person-Centered Communication Training Intervention. *Research in Gerontological Nursing*, 14(5), 245–253. https://doi.org/10.3928/19404921-20210708-01
- Savundranayagam, M. Y., & Lee, C. (2017). Roles of Communication Problems and Communication Strategies on Resident-Related Role Demand and Role Satisfaction. American Journal of Alzheimer's Disease & Other Dementias®, 32(2), 116–122. https://doi.org/10.1177/1533317517689876
- Savundranayagam, M. Y., & Moore-Nielsen, K. (2015). Language-based communication strategies that support person-centered communication with persons with dementia. *International Psychogeriatrics*, 27(10), 1707–1718. https://doi.org/10.1017/S1041610215000903
- Savundranayagam, M. Y., Sibalija, J., & Scotchmer, E. (2016). Resident Reactions to Person-Centered Communication by Long-Term Care Staff. American Journal of Alzheimer's Disease & Other Dementias[®], 31(6), 530–537. https://doi.org/10.1177/1533317515622291
- Scales, K. (2022). Transforming Direct Care Jobs, Reimagining Long-Term Services and Supports. *Journal of the American Medical Directors Association*, 23(2), 207– 213. https://doi.org/10.1016/j.jamda.2021.12.005
- Scott, A., Ryan, A., James, I., & Mitchell, E. A. (2011). Perceptions and implications of violence from care home residents with dementia: A review and commentary. *International Journal of Older People Nursing*, 6(2), 110–122. https://doi.org/10.1111/j.1748-3743.2010.00226.x
- Sethna, C. (2013). Understanding Personal Support Workers and their Role in Ontario [M.H.Sc., University of Ontario Institute of Technology (Canada)]. https://www.proquest.com/docview/1459763539/abstract/473C41F077CE4EB1P Q/1
- Slaughter, S., Dean, Y., Knight, H., Krieg, B., Mor, P., Nour, V., Polegato, E., Seneviratne, C., Shenfield, D. G., & Sherwood, E. (2007). The Inevitable Pull of the River's Current: Interpretations Derived From a Single Text Using Multiple Research Traditions. *Qualitative Health Research*, 17(4), 548–561. https://doi.org/10.1177/1049732306298812
- Statistics Canada. (2004). *Employment trends in nursing*. https://www150.statcan.gc.ca/n1/pub/75-001-x/11104/7611-eng.pdf
- Statistics Canada. (2021, September 16). *The Daily—A profile of nursing and residential care facilities, 2019.* https://www150.statcan.gc.ca/n1/dailyquotidien/210916/dq210916c-eng.htm

- Su, H.-F., Koo, M., Lee, W.-L., Sung, H.-C., Lee, R.-P., & Liu, W.-I. (2021). A dementia care training using mobile e-learning with mentoring support for home care workers: A controlled study. *BMC Geriatrics*, 21(1). https://doi.org/10.1186/s12877-021-02075-3
- Swinton, J., Vilapakkam Nagarajan, S., Atee, M., Morris, T., & Poulos, C. J. (2023). Being present and meaningful engagement for aged care residents living with dementia: A mixed-methods evaluation of Australian care worker's experiences. *The Gerontologist*, gnad131. https://doi.org/10.1093/geront/gnad131
- Szymczynska, P., & Innes, A. (2011). Evaluation of a dementia training workshop for health and social care staff in rural Scotland. *Rural & Remote Health*, *11*(2), 1–3.
- Thestrup Hansen, S., Kjerholt, M., Friis Christensen, S., Brodersen, J., & Hølge-Hazelton, B. (2021). Nurses' Experiences When Introducing Patient-Reported Outcome Measures in an Outpatient Clinic: An Interpretive Description Study. *Cancer Nursing*, 44(2), E108. https://doi.org/10.1097/NCC.00000000000808
- Thome, S. (1998). Ethical and Representational Issues in Qualitative Secondary Analysis. *Qualitative Health Research*, 8(4), 547–555. https://doi.org/10.1177/104973239800800408
- Thorne, S. (2016). *Interpretive Description: Qualitative Research for Applied Practice*. Taylor & Francis Group. http://ebookcentral.proquest.com/lib/west/detail.action?docID=4511832
- Thorne, S., Kirkham, S. R., & MacDonald-Emes, J. (1997). Interpretive description: A noncategorical qualitative alternative for developing nursing knowledge. *Research in Nursing & Health*, 20(2), 169–177. https://doi.org/10.1002/(SICI)1098-240X(199704)20:2<169::AID-NUR9>3.0.CO;2-I
- Thorne, S., Kirkham, S. R., & O'Flynn-Magee, K. (2004). The Analytic Challenge in Interpretive Description. *International Journal of Qualitative Methods*, *3*(1), 1–11. https://doi.org/10.1177/160940690400300101
- Timmons, S., O'Loughlin, C., Buckley, C., Cornally, N., Hartigan, I., Lehane, E., Finn, C., & Coffey, A. (2021). Dementia palliative care: A multi-site survey of long term care STAFF'S education needs and readiness to change. *Nurse Education in Practice*, 52. https://doi.org/10.1016/j.nepr.2021.103006
- Toteh Osakwe, Z., Madu, E. N., Céspedes, A., Atairu, M., Osborne, J. C., & Stefancic, A. (2022). Home health aide perceived information needs for dementia-specific care plans. *Geriatric Nursing*, 43, 1–6. https://doi.org/10.1016/j.gerinurse.2021.10.017
- Tricco, A. C., Lillie, E., Zarin, W., O'Brien, K. K., Colquhoun, H., Levac, D., Moher, D., Peters, M. D. J., Horsley, T., Weeks, L., Hempel, S., Akl, E. A., Chang, C.,

McGowan, J., Stewart, L., Hartling, L., Aldcroft, A., Wilson, M. G., Garritty, C., ... Straus, S. E. (2018). PRISMA Extension for Scoping Reviews (PRISMA-ScR): Checklist and Explanation. *Annals of Internal Medicine*, *169*(7), 467–473. https://doi.org/10.7326/M18-0850

- Verkaik, R., van Antwerpen-Hoogenraad, P., de Veer, A., Francke, A., & Huis In Het Veld, J. (2017). Self-management-support in dementia care: A mixed methods study among nursing staff. *Dementia (London, England)*, 16(8), 1032–1044. https://doi.org/10.1177/1471301216632416
- Wang, Y., Liu, Y., Tian, J., Jing, M., & Zhang, K. (2020). Analysis on nursing competence and training needs of dementia caregivers in long-term care institutions. *International Journal of Nursing Sciences*, 7(2), 198–205. https://doi.org/10.1016/j.ijnss.2020.03.003
- Whall, A. L., & Kolanowski, A. M. (2004). Special Section—Behavioral symptoms of dementia: Their measurement and intervention. Editorial: The need-driven dementia-compromised behavior model—a framework for understanding the behavioral symptoms of dementia. Aging & Mental Health, 8(2), 106–108. https://doi.org/10.1080/13607860410001649590
- World Health Organization. (2017). Global action plan on the public health response to dementia 2017–2025. World Health Organization. https://iris.who.int/handle/10665/259615
- World Health Organization. (2023). *Dementia*. World Health Organization. https://www.who.int/news-room/fact-sheets/detail/dementia
- Xu, L., Hsiao, H.-Y., Denq, W., & Chi, I. (2018). Training needs for dementia care in China from the perspectives of mental health providers: Who, what, and how. *International Psychogeriatrics*, 30(7), 929–940. https://doi.org/10.1017/S1041610217002654
- Zagrodney, K. A. P., King, E. C., Simon, D., & McKay, K. A. N. and S. M. (2023). Economic Evidence for Home and Community Care Investment: The Case for Ontario Personal Support Workers' Wage Parity. *Healthcare Policy*, 19(1). https://www.longwoods.com/content/27161/healthcare-policy/economicevidence-for-home-and-community-care-investment-the-case-for-ontariopersonal-support-work
- Zagrodney, K., & Saks, M. (2017). Personal Support Workers in Canada: The New Precariat? *Healthcare Policy*, *13*(2). https://doi.org/doi:10.12927/hcpol.2017.25324
- Zeytinoglu, I. U., Denton, M., Davies, S., & Plenderleith, J. M. (2009). Casualized employment and turnover intention: Home care workers in Ontario, Canada. *Health Policy*, 91(3), 258–268. https://doi.org/10.1016/j.healthpol.2008.12.004

Zimmerman, S., Williams, C. S., Reed, P. S., Boustani, M., Preisser, J. S., Heck, E., & Sloane, P. D. (2005). Attitudes, Stress, and Satisfaction of Staff Who Care for Residents With Dementia. *The Gerontologist*, 45(suppl_1), 96–105. https://doi.org/10.1093/geront/45.suppl_1.96

Appendices

Appendix A: Ethics Approval



Date: 16 August 2023

Project ID: 114354 Review Reference: 2023-114354-82760 Study Title: Be EPIC: Dementia Training for Mid-Career Workers Application Type: Continuing Ethics Review (CER) Form Review Type: Delegated Date Approval Issued: 16/Aug/2023 15:02 REB Approval Expiry Date: 30/Aug/2024

Dear

The Western University Research Ethics Board has reviewed the application. This study, including all currently approved documents, has been re-approved until the expiry date noted above.

REB members involved in the research project do not participate in the review, discussion or decision.

Western University REB operates in compliance with, and is constituted in accordance with, the requirements of the Tri-Council Policy Statement: Ethical Conduct for Research Involving Humans (TCPS 2); the International Conference on Harmonisation Good Clinical Practice Consolidated Guideline (ICH GCP); Part C, Division 5 of the Food and Drug Regulations; Part 4 of the Natural Health Products Regulations; Part 3 of the Medical Devices Regulations and the provisions of the Ontario Personal Health Information Protection Act (PHIPA 2004) and its applicable regulations. The REB is registered with the U.S. Department of Health & Human Services under the IRB registration number IRB 00000940.

Please do not hesitate to contact us if you have any questions.

Electronically signed by:

Reason: I am approving this document

Note: This correspondence includes an electronic signature (validation and approval via an online system that is compliant with all regulations).

Appendix B: Semi-Structured Focus Group Questions

Baseline 1 Session:

- 1. What would you like to get out of this training program?
 - a. OR: What is one thing you would like to work on with your residents who have dementia?
- 2. Does being 'mid-career' change your experience working with people living with dementia, compared to when you first started in the PSW role?
- 3. What makes a good PSW? (impromptu question)
- 4. How do you communicate with the rest of your team? What are your relationships like in your workplace?

Baseline 2 Session:

- 1. Have any of your goals for this training changed or have new ones arose since engaging in the first training session?
- 2. Has anything changed in the last week in your long-term care home (e.g., changes in shifts, client caseloads, relationships with family members, relationships within the workplace)?
- 3. How does your interactions with the actors compare to your interactions with your clients living with dementia?
- 4. Do you find yourself spending too much time with your clients? Or do you not have enough time with them on a daily basis? (impromptu question)
- 5. Did you feel you had enough time to interact with the actor?
 - a. Probe: Is more than five minutes needed with the actor?
- 6. How often do you feel you have triggered responsive behaviours from a client due to rushing in the workplace?
- 7. What brings you joy in your job as a PSW?
- 8. Do you have any questions?

Module 1 Session:

- A. What would you like to get out of this training program?
 - a. Probe: Is there something you would like to learn from this program? Is there something you would like to work on with your residents with dementia?
- B. Can you share your experiences of working with people who have dementia?
- C. Does being 'mid-career' change your experience working with people who have dementia?
- D. How do you communicate information about the client with dementia with your co-workers?
- 1. Describe your relationship with your clients?
 - a. Probe: What are some positive interactions you've had? What helps create positive interactions with clients?
 - b. Probe: What are some not so positive interactions you've had?

- c. What do you think establishes trust with your clients? (impromptu question)
- 2. What supports are in place to help you come back to work if you've had a bad day?
 - a. How do you cope with an interaction with a resident or co-workers that does not go as you had hoped?
- 3. Do you find that your interactions with clients is better when you have more information about them, such as their history? Does it make a difference in your care approach?
- 4. How do you share information about a client with your co-workers?

Modules 2 and 3:

- 1. How did today's training session go for you?
- 2. What did you like the most about today's training session and why?
- 3. What did you learn from today's training session?
 - a. Probe: did you learn something new today?)
- 4. Did the training session challenge how you would normally interact with a person living with dementia? (impromptu question)
- 5. Will what you learned from today's training session influence how you are going to provide care in the future?
- 6. What did you think about giving and receiving feedback? How was the feedback?
- 7. Did you like giving feedback to your co-workers today?
- 8. How did it feel to be able to remove yourself from the simulation and ask your coworkers for help?
- 9. How do you feel about the communication within your groups?
- 10. Do you feel as if you can speak more freely in the debrief groups compared to the first focus group session?
 - a. Probe: In the smaller groups following a simulation, do you feel comfortable sharing feedback with your peers?
- 11. Do you enjoy the debriefing as a group?
- 12. What did you find the most unique about today's session?
- 13. Do you have any constructive feedback for today's session?
- 14. Did today's session change any of the beliefs you previously had about persons with dementia?
- 15. Do you have any questions regarding what you learned during today's session?

Module 4 Session:

- 1. What did you learn from Be EPIC? What did you take away from Be EPIC?
- 2. Will anything you learned from Be EPIC influence your approaches and interactions with clients living with dementia?
- 3. How did the simulations affect your learning?
- 4. Did you feel safe during the simulations?
- 5. Did feedback from the actors affect your learning?

- 6. How did you feel about receiving feedback from your co-workers after completing a simulation?
- 7. In your opinion, what is more important: doing the training with the actors or doing the training with the actors and the feedback?
- 8. What is the most valuable aspect of Be EPIC? (Probe: What is one strength or advantage of Be EPIC as a training program?)
- 9. What did you learn about yourself?
- 10. Do you have any constructive feedback for today's session?
 - a. Probe: Does Be EPIC have enough opportunities to practice what you learned?
 - b. Probe: Do you have any constructive feedback for the facilitator? Do you have any constructive feedback for how the focus groups were held?
 - c. Probe: Do you have any constructive feedback regarding the content you learned?
- 11. How does Be EPIC support 'mid-career' PSWs?
- 12. How is Be EPIC different when compared to other dementia-specific training programs?
- 13. Would you recommend Be EPIC?

Three Month Follow-up

- 1. What did you learn from the Be EPIC program that you've applied in the last three months?
- 2. How did the simulations affect your learning? How did having the actors portray people living with dementia affect your learning?
- 3. How did receiving feedback from the actors affect your learning?
- 4. What have you learned about yourself since completing Be EPIC?
- 5. How can Be EPIC be improved? How can the Be EPIC facilitators be improved?
- 9. Has anything changed in the last three months in your long-term care home? (Probe: changes in shifts, client caseloads, relationships with family members, relationships within the workplace)
- 6. What support have you received (i.e., from co-workers, client families, managers) to enable you to use person-centered communication?
- 7. How has your communication with colleagues changed since completing Be EPIC?

Curriculum Vitae

Name:	Grace Norris
Post-secondary Education and Degrees:	The University of Western Ontario London, Ontario, Canada 2022-2024 M.Sc.
	Carleton University Ottawa, Ontario, Canada 2017-2021 B.HSc. (Hons)
Honours and Awards:	Master's Student Poster Presentation – 1 st Place Annual Health & Rehabilitation Sciences Graduate Research Conference Western University 2023
	Dean's Honour List Scholar Carleton University 2021 Admission Scholarship Carleton University
	2017 Health Science Scholarship Carleton University 2017
Related Work Experience	Graduate Research Assistant Sam Katz Community Health and Aging Research Unit 2022-2024
	Knowledge Mobilization Lead Sam Katz Community Health and Aging Research Unit 2022-2024

Be EPIC-VR Facilitator Sam Katz Community Health and Aging Research Unit 2022-2024

Graduate Teaching Assistant University of Western Ontario 2022-2023

Undergraduate Research Placement Student Boxing4Health, Ottawa, Ontario 2020-2021

Clinical Research Assistant Children's Hospital of Eastern Ontario (CHEO), Ottawa, Ontario 2019

Publications:

Norris, G., Savundranayagam, MY., Teachman, G., & Vafaei, A. (2024). The representation of personal support workers in dementia-specific learning need assessments: A scoping review. *Educational Gerontology*, 1–19. https://doi.org/10.1080/03601277.2024.2356925

Norris, G., Savundranayagam, MY., Teachman, G., & Vafaei, A. The continuous learning needs of personal support workers who care for people living with dementia in long-term care. [Manuscript in progress].

Savundranayagam, MY., Chen, A., **Norris, G**., Schumann, A., Campos, J., & Orange, J. Be EPIC-VR: Translating an In-Person Person-Centered Communication Training Program into Virtual Reality. [Manuscript in progress].

Savundranayagam, MY., Schumann, A., Sagrak, A., **Norris, G.**, Chen, A., Campos, J., & Orange, JB. Organizational Readiness for a Virtual Reality Training Program called Be EPIC-VR in Home Care and Long-term Care Settings. [Manuscript in progress].

Presentations:

Norris, G., Savundranayagam, MY., Teachman, G., & Vafaei, A. (accepted June 11, 2024). The continuous learning needs of personal support workers who care for people

living with dementia in long-term care. To be presented at the annual meeting of the Gerontological Society of America.

Savundranayagam, MY., **Norris, G**., Schumann, A., Campos, J., Sarma, A., & Orange, JB. (accepted June11, 2024). Acceptability and Preliminary Efficacy of Be EPIC-VR training on Frontline Healthcare Workers. To be presented at the annual meeting of the Gerontological Society of America.

Savundranayagam, MY., **Norris, G**., Malheiro, G., Schumann, A., Campos, J., & Orange, JB. (accepted June 11, 2024). Implementing Be EPIC-VR in Healthcare: Aligning Virtual Reality Training with Organizational Readiness. To be presented at the annual meeting of the Gerontological Society of America.

Norris, G., Savundranayagam, MY., Teachman, G., & Vafaei, A. (2024, March). The continuous learning needs of personal support workers who care for people living with dementia in long-term care. Presented at the annual Western Research Forum Annual Graduate Conference, University of Western Ontario, London, ON.

Norris, G., Savundranayagam, M.Y., Teachman, G., & Vafaei, A. (2024, February 2). The continuous learning needs of personal support workers who care for people living with dementia in long-term care. Presented at the annual Health & Rehabilitation Sciences Graduate Research Conference, University of Western Ontario, London, ON.

Savundranayagam, MY, Schumann, A., **Norris, G**., Chen, A., Campos, J., & Orange, JB. (2023, October 28). Organizational Readiness for a Virtual Reality Training Program called Be EPIC-VR in Home Care and Long-Term Care Settings. Presented at the annual meeting of the Canadian Association on Gerontology, Toronto, ON.

Savundranayagam, MY., Chen, A., **Norris, G**., Schumann, A., Campos, J., & Orange, JB. (2023, October 28). Be EPIC-VR: Translating an In-Person Person-Centered Communication Training Program into Virtual Reality. Presented at the annual meeting of the Canadian Association on Gerontology, Toronto, ON.

Norris, G., Savundranayagam, M.Y., Teachman, G., & Vafaei, A. (2023, October 27). The representation of personal support workers in dementia-specific learning need assessments: a scoping review. Presented at the annual meeting of the Canadian Association on Gerontology, Toronto, ON.

Schumann, A., Savundranayagam, MY., Chen, A., **Norris, G**., Campos, J., & Orange, JB. (2023, October 18). Implementation Science: Implementing a virtual reality program

in long-term care and home care settings. Presented in Health and Rehabilitation Sciences Health and Aging Graduate Seminar, Western University.

Savundranayagam, MY., **Norris, G**., Malheiro, G., Chen, A., & Schumann, A. (2023, September). Using implementation science to optimize the research impact of Be EPIC-VR. BrainsCAN Research Impact Day, London, ON.

Savundranayagam, MY., Schumann, A., Sagrak, A., **Norris, G.**, Chen, A., Campos, J., & Orange, JB. (2023, November 10). Organizational Readiness for a Virtual Reality Training Program called Be EPIC-VR in Home Care and Long-term Care Settings. Presented at the annual meeting of the Gerontological Society of Amerixa, Tampa, FL.

Savundranayagam, MY, Chen, A., **Norris, G**., Schumann, A., Jaur, S., Campos, J., & Orange, JB. (2023, November 10). Be EPIC-VR: Translating an In-Person Person-Centered Communication Training Program into Virtual Reality. Presented at the annual meeting of the Gerontological Society of America, Tampa, FL.

Norris, G., & Savundranayagam, M.Y. (2023, February 1). The continuous learning needs of personal support workers working with persons living with dementia in long-term care homes: a rapid review. Presented at the annual Health & Rehabilitation Sciences Graduate Research Conference, University of Western Ontario, London, ON.